HOW DO SCHOOL RESOURCES AND LEARNING ENVIRONMENTS DIFFER ACROSS AUSTRALIAN RURAL, REGIONAL AND METROPOLITAN COMMUNITIES?

Kevin Sullivan

This thesis is presented for the degree of

Doctor of Philosophy

Murdoch University

2018

School of Education

Murdoch University

Perth, Western Australia, Australia

I declare that:

- The thesis is my own account of my research, except where other sources are acknowledged.
- The extent to which the work of others has been used is clearly stated in each chapter and certified by my supervisors.
- The thesis contains as its main content work that has not been previously submitted for a degree at any other university.

Kevin Sullivan

A note on formatting and style

This PhD thesis comprises three research papers, two of which have been published and one which is under review. These formatted documents are incorporated into the thesis along with additional text that has been provided to introduce and link the published work. It is hoped that the final amalgamation allows for the development of a cohesive body of research that can be easily followed.

The thesis has continuous pagination, which can be seen at the top right corner of each page. For published documents, the original journal page numbers are also visible. References and appendices have been amalgamated and placed at the end of the thesis.

The three papers have undergone double blind peer-review. I co-authored these papers with my supervisors, as is common in the social and natural sciences. As the first author, I conducted the analyses and wrote the papers. My supervisors provided guidance and thorough feedback on all aspects of the papers, including editorial assistance.

Acknowledgements

A thesis is more than words, numbers, and characters on a page. The dimensional components of a study may not expose the hopes and sacrifices of an individual, nor adequately express an author's appreciation for generous mentors and love for patient family who have invested time, energy, support and understanding without personal gain, but they are symbolic of a pathway and journey through time which whilst often infertile sustain a richness that will forever yield.

To my supervisors, Associate Professors Laura Perry and Andrew McConney, I am deeply indebted to you for the level of expertise, support and guidance that you have given me. I hope that I can inspire others in the way that you have inspired me. To those that care I say, "Thank you!" And to onlookers who dare, I say, "Dream!"

Dr Perry and Dr McConney have been amazing mentors and guides. I value the time, effort, dedication, and leadership that they have shown me. I could not have asked for more patient or experienced researchers to act as my supervisors. I not only value the working relationship that we have forged but marvel at the achievements that we have made publishing two articles of significant merit with an Australian education journal, and a pending publication with an international journal.

Dedication

This research would not have been possible without the love and support of my family. To my beautiful wife, Sharon and amazing children Andrew and Olivia, "Thanks guys!"

Abstract

Despite recent emphasis on improved government funding and advances in technology that reduce the isolation of rural communities, research continues to highlight that Australian students attending rural schools, on average, achieve poorer academic outcomes than their urban peers. It is plausible that these lower academic outcomes are associated with the characteristics of rural schools. Little is known, however, about the nature and degree to which schools differ between rural and metropolitan communities in Australia. The aim of this study is to compare school characteristics across a range of rural and metropolitan settings, using a large-scale and nationally representative dataset.

The study comprised three investigations that examined how student achievement, school resources and school learning environments vary across urban, regional, rural, and remote communities using data from the *Programme for International Student Assessment* (PISA). PISA is an international assessment created by the Organization for Economic Cooperation and Development (OECD) that assesses the reading, mathematics and science literacies of 15-year-old students. PISA also collects data from school principals and students about a range of student and school related variables that may be related to student literacy in the three subject domains. The three investigations used data from the 2009 cycle of PISA, which comprised approximately 470,000 students from 65 countries and economies, including over 14,000 Australian students attending 353 schools. Descriptive statistics were used to compare student and school principal perspectives about a range of school resources and learning environments.

The initial paper investigated school resource variables across eight rural-urban community categories in Australia. The school resource variables included computers for education, the ratio of computers to students, computers with internet access, and principals' perspectives of the degree to which shortages of teaching personnel and teaching materials and resources hinder student learning. On average, principals of schools in rural communities were more likely than their counterparts in larger communities to perceive that instruction was hindered by shortages of teaching personnel and to a lesser extent by shortages of teaching resources. Principals in larger towns and very large towns (ranging in size from 15,000 to 50,000 residents) reported that shortages of mathematics teachers were a hindrance to a similar degree as school principals in small rural communities.

The second paper examined differences in school learning environments across eight rural-urban community categories in Australia. Learning environments were measured by the following: principals' perceptions of teacher and student behaviour, student attitudes towards school, and student perceptions of their classroom disciplinary climate and relationships with teachers. The findings show that regardless of location, most Australian students believed that schooling is worthwhile and reported positive relationships with their teachers. However, both student and principal perceptions of disciplinary climate and learning environments were more positive in urban communities than in rural communities.

The third paper compared school community differences at an international level, contrasting two economic, culturally, and socially similar nations, Canada and New Zealand, with Australia. Research focused on: average student reading performance, socioeconomic status and parent education, principals' perceptions about their school's resources, and student perceptions of classroom disciplinary climate, teacher-student relations, and teacher instructional strategies. The findings showed that across Canada, New Zealand and Australia reading literacy performance and school learning environments are less positive in rural communities than in urban communities. However, these inequalities between rural and urban school communities are greater in Australia than in the other two countries. Of the three countries, rural school principals in Australia are the most likely to report that shortages of teaching personnel hinder learning. The findings show that school learning environments and school resources vary substantially across Australian school communities. Given the patterning of student performance favouring urban over rural school communities, it may very well be that elements such as rural school shortage of resources and relations between student and teacher negatively impact the academic performance of students. The three studies highlight that much still needs to be learned about: (1) recruiting and retaining teachers in large regional Australian towns; (2) the degree to which shortages of instructional material and equipment are associated with geographic location; and, (3) the reasons underlying students' and principals' views of school learning environments in large regional towns (up to 50,000 residence) are less positive than their counterparts' views in rural and remote communities. The findings also suggest that education policies and structures can play a role in ameliorating or exacerbating rural educational disadvantage.

Key words: School resources, school learning environments, educational opportunities, educational experiences, educational outcomes, geographic location, rural education, PISA

Table of Contents

THESIS DECLARATION	2
FORMATING AND STYLE	3
ACKNOWLEDGEMENTS	4
DEDICATION	4
ABSTRACT	5
TABLE OF CONTENTS	8
LIST OF TABLES	11
LIST OF FIGURES	13
LIST OF PUBLICATIONS	14
1. CHAPTER 1: INTRODUCTION	15
a. Introduction	15
b. Purpose	15
c. Background	
d. Positionality of the researcher	21
e. Problem statement	
f. Research questions	
g. Significance of study	
h. Structure of thesis	
2. CHAPTER 2: LITERATURE REVIEW	
a. Overview	
b. Significance of socioeconomic status	
c. Educational opportunities	
d. Educational experience	
e. Educational outcomes	

f. Rurality in the Australian context	
g. Conclusion	
3. CHAPTER 3: METHODOLOGY	41
a. Methodology	41
b. Research design	41
b. Data source	41
c. Variables	43
d. Analytical strategy	47
e. Clarification of terms	48
4. CHAPTER 4: STUDY 1	50
a. Background	50
b. Published manuscript	52
c. Significance of study and link to the next chapter	69
5. CHAPTER 5: STUDY 2	71
a. Background	71
b. Published manuscript	73
c. Significance of study and link to the next chapter	91
6. CHAPTER 6: STUDY 3	93
a. Background	93
b. Research paper	95
c. Appendix: Questionnaire item for selected indexes	105
d. Significance of study and link to the next chapter	
7. CHAPTER 7: CONCLUSION	
a. Key findings	
b. Significance	

	c. Limitations	. 117
	d. Future research	. 118
	e. Recommendations	. 119
R	EFERENCES	. 122

List of Tables

Chapter 2. Background	
Table 1. Overview of Australian Council for Educational Research and	
Programme of International Student Assessment Classification of School	
Community	45
Paper 1	
Table 1. Distribution of students and schools by school community	58
Table 2. Indigeneity of school communities	58
Table 3. Mean mathematics, reading and science literacy performance scores	
by geographic location	60
Table 4. Ratio of student numbers to computers by school community	62
Table 5. Lack of teaching personnel by school community as reported by	
Principals	63
Table 6. Shortages of teaching resources	64
Paper 2	
Table 1. Classification of school communities and distribution of students	78
Table 2. Principals' perception of teacher and student behaviour related to	
school climate	9-80
Table 3. Indexes of teacher and student behaviour related to school climate	81
Table 4. Student attitudes towards school	82
Table 5. Student perceptions of their relations with teachers	83
Table 6. Student perceptions of classroom disciplinary climate	87
Table 7. Student perceptions of their learning environments	88
Paper 3	
Table 1. Number and proportion of participating schools and students	97

Table 2.	Mean student reading performance, socioeconomic status and parental
	education, by school community type
Table 3.	Score point differences between country mean and school community
	mean in reading literature using the Programme of International Student
	Assessment (2009)
Table 4.	Principals' perceptions about their school's resources and teacher
	instructional shortages101
Table 5.	Student perceptions of classroom disciplinary climate, teacher-student
	relations and teacher instructional strategies102

List of Figures

Paper 1

Figure 1. Mathematics, reading and science academic performance in
Programme of International Student Assessment 2009 by mean
school socioeconomic status and geographic location
Paper 2
Figure 1. Distribution of academic performance in Programme of International
Student Assessment 2009 by school community
Figure 2. School principals' perceptions of teacher and student behaviour by
community type, in comparison to students' socioeconomic status 85
Figure 3. Student perceptions of learning environments and socioeconomic
status
Paper 3
Figure 1. Mean socioeconomic status and reading performance,
disaggregated by school community type in Australia, Canada
and New Zealand100

Publications

- 1. Sullivan, K., Perry, L., & McConney, A. (2013). How do school resources and academic performance differ across Australia's rural, regional, and metropolitan communities? *Australian Educational Researcher*, 40(3), 353-372.
- Sullivan, K., Perry, L. B., & McConney, A. (2014). How do school learning environments differ across Australia's rural, regional and metropolitan communities? *Australian Educational Researcher*, *41*(5), 521-540. DOI: 10.1007/s13384-014-0144-1.
- Sullivan, K., McConney, A., & Perry, L. B. (2018). A Comparison of Rural Educational Disadvantage in Australia, Canada, and New Zealand using OECD's PISA. Sage Open Journal of Educational Research, 8(4), 1-12. DOI: 10.1177/2158244018805791

Chapter 1:

Introduction

This chapter provides background information, describes the study structure, problem statement, purpose and research questions, and concludes with a reflection on the potential significance of the findings.

Purpose

The disparity between rural and urban education in Australia is a strong driver behind my investigation into school resources and learning environments. Results from the Programme of International Student Assessment (PISA), an international assessment designed to assess the reading, mathematics and science literacy of 15-year-old students, have shown that students in rural communities consistently have lower reading, mathematics and science literacy performance than students in urban school community areas in Australia (Cresswell & Underwood, 2004; Thomson & De Bortoli, 2010). These results from PISA are consistent with other measures of educational outcomes; for example, Australian students in rural communities are less likely than their urban peers to complete secondary school or university (James, 2001). They also have, on average, lower scores on Australia's National Assessment Program-Literacy and Numeracy (Lamb, Glover, & Walstab, 2014). The reasons why rural students have on average lower educational outcomes is undoubtedly complex, and it is likely due to a range of factors related to students, communities and schools. It is plausible that rural students' lower outcomes are related to inequalities of resources and learning environments between rural and urban schools. Not much is known, however, about the nature and extent of these inequalities between schools. The purpose of this study, therefore, is to investigate how schools that are geographically distributed across Australia's rural, remote and metropolitan communities vary in terms of school resources and learning environments.

Using a large nationally representative dataset like PISA, which is designed and administered by the Organization for Economic Cooperation and Development (OECD), provides researchers with the opportunity to undertake descriptive mapping of school characteristics: for example, in this study school resource inputs, such as qualified teachers and curricula are viewed as a constructs of education opportunity whilst variables that reflect the school learning environment, like student and teacher interpersonal interactions are deemed education experience. The level of student literacies attained are determined by the PISA 2009 assessment outcomes in reading, mathematics and science. Identifying patterns within this data set can be useful for understanding inequalities in educational opportunities, experiences and outcomes for Australian students in rural communities. Moreover, understanding how schools vary across geographic locations is important as learning resources and learning environments are educational ends in themselves and not just predictors of academic outcomes.

My research is intended to uncover patterns, drawn from PISA (2009) literacy scores in reading, mathematics and science, and school characteristic variables derived from student and principal questionnaires, about the nature and extent of rural educational disadvantage in Australia. Also, international data are sourced from two economically and culturally similar countries, Canada and New Zealand, to provide comparison data. By contrasting the Australian findings with school communities across Canada and New Zealand, patterns may be identified that could help to gauge how unique Australian school communities are and whether insights can be gained from other similar countries.

This study has enabled me to examine how school characteristics vary across urbanrural contexts and settings in Australia. I utilised information from the OECD website to conduct secondary analyses of PISA, downloading PISA 2009 data, which included principal and school questionnaires and the student and school data files. During this process, I discovered that PISA is not a perfect tool for evaluating educational systems and student outcomes. A disadvantage is that like other cross-national tests of student achievement, PISA is not longitudinal (Hopmann, Brinek & Retzl, 2008) and therefore cannot be used to establish causality. Some academics highlight the complications that surround the collection, assessment and reporting of large-scale educational data (Cresswell, Schwantner & Waters, 2015; Rutkowski, Gonzalez, Joncas & Davier, 2010). Rutkowski and Rutkowski (2010) raised issues about PISA's treatment of missing data and questioned the reliability of its student questionnaire, whilst Stewart (2013) regarded as problematic PISA's potential to act as global arbiter to the world's schooling systems. There is also growing concern that PISA is overly valued and politicised (Sellar, Rutkowski & Thompson, 2017). While these critiques are valid, PISA nevertheless has many advantages. It is nationally representative, and the number of participating countries and students is very large, with 65 participating countries and over 470,000 students included in the PISA 2009 sample.

Through my research I have gained knowledge on possible reasons why rural students across Australia, on average, have lower educational outcomes in reading, mathematics and science literacy performance in PISA 2009 assessments. Whilst no single factor can be attributed with causing urban-rural school community assessment trends, I have discovered that more research needs to be undertaken to increase understanding of the significance of school resources' and learning environments' inequity across Australian school communities. The purpose of this study is to investigate how schools that are geographically distributed across Australia's rural, remote and metropolitan communities vary in terms of school resources and learning environments. However, my goal as a researcher is to understand the significance, inter-connectedness and impact of these trends on student experience and learning outcomes and better understand the cause of inequality of education across urban-rural school communities.

Background

Learning is fundamental to human evolution. Nelson Mandela (2003) referred to education in a speech as, "...the most powerful weapon, which you can use to change the world." At the most basic level education provides individuals within a society the opportunity to maximize their ability to think, problem-solve and communicate. It is my belief that education can help people overcome adversity and maximize their opportunity to learn, teach others, secure employment and make a positive contribution to society.

However, due to inequality based on racial, social, economic or gendered prejudice education opportunities are often not distributed evenly. The topic of education inequality within the Australian context has been relevant for many years. Arguably Australia's greatest education reform occurred during Prime Minister Gough Whitlam's tenure of office (1972-1975). A driving force behind 'The Karmel Report and special education in Australia' issued by the Whitlam government was a desire to minimize inequality in education (Andrews, 1973). However, decades later the Australian Human Rights and Equal Opportunities Commission (2000) determined that students from rural school communities fared worse than their urban counterparts. Today, there is an increased focus on national education standards and assessment programmes, such as National Assessment Programme Literacy and Numeracy (NAPLAN) and PISA. These assessments, while sometimes controversial, are useful for uncovering the magnitude of educational inequalities and their trajectories over time: for example, data from PISA has shown that educational inequalities between socially advantaged and disadvantaged students is large and stable. As noted by Sue Thompson from the Australian Council for Educational Research,

The difference between advantaged and disadvantaged students [in Australia] is around three years of schooling. That's not changed in 16 years of [PISA] testing (for reading). That's the critical thing. We're still not attending to those gaps. (http://www.teachforaustralia.org/2016/12/07/australias-

pisa-results-show-educational-disadvantage-gapremains/)

Whilst assessment programmes may not in themselves be a solution to rectify inequality, they provide comparable data at state, national and international levels, uncover or highlight education trends, increase community awareness and ensure people, organisations and governments with responsibility for appropriate provision in education are held accountable.

My vocation as a teacher is to support young learners and help them to recognize the importance and value of education. Empowering students to learn how to learn will hopefully maximize their opportunity and success in life. In following my calling, I have discovered inequities within Australia's education system. I have made it my responsibility to attempt to understand the complexity of education disadvantage and direct my own learning towards ameliorating the disparity faced by educationally underserved groups, such as students from lower socioeconomic (SES) backgrounds, students in rural and remote communities, and Indigenous students within Australia.

This study was conducted as part of an Australian Research Council (ARC) grant awarded to my principal supervisor, Laura Perry, in 2010. At the time that Dr Perry was awarded the ARC grant I was working full-time as a primary school teacher and tutoring a firstyear School of Education unit at Murdoch University. Having taught in the Australian education system for 16-years I was endeavouring to expand my contribution to the profession of teaching by educating future teachers. Whilst I was tutoring an undergraduate educational unit I gained a real interest in the value of research. I discovered that research could extend my scope to make a positive impact on future teachers. So, when the opportunity arose I immediately saw the value and understood the privilege I had been provided to work with Dr Perry and Dr McConney on a cross-national investigation into school resources and school learning environments. Through discussions with Dr Perry I learned the educational significance of conducting descriptive research that could establish a better understanding of the school characteristics that may influence students' educational outcomes. I also became aware that not much was known about the ways in which schools vary across rural and urban communities in Australia. Recognising how schools vary across communities could help us identify why rural students often have lower outcomes than their urban peers, which in turn could highlight possible strategies for remedying their lower outcomes. Importantly, I discovered the significance of recognising that educational experiences in schools are an important end in themselves.

The PISA dataset provides researchers with an extensive range of student and school predictor variables to analyse. The PISA 'school community' variable contains data that provides information relating to school community. I was particularly interested in school community predictor variables as reported by students and principals, as they could provide clues as to how school communities vary. I concentrated my research on the most recent PISA dataset at this time, and committed to using the PISA 2009 data, for the entirety of my research.

A possible correlation exists between school resources, school learning environments and student achievement. Willms (2010) showed that learning environment characteristics are highly correlated with school mean socioeconomic status (SES). Similarly, the main PISA 2009 report published by the OECD showed that 13% of the variation in student achievement is associated with school learning environment, however, much of this effect is linked with student and school SES (OECD, 2010b, p. 107). The PISA report concludes that learning environments do matter for student outcomes and learning environments conducive to academic achievement are more easily realized in schools that enroll students from middle and upper-class backgrounds.

Positionality of the researcher

During my 23-year education career, I have fulfilled many roles and taught across a broad spectrum of learning environments, spanning from early childhood, primary and senior school through to tertiary education. Positions of leadership that I have undertaken at the school level include assistant principal, senior teacher, information communication and technology coordinator, learning area coordinator, and administration team leadership member. Leadership roles that I have held at the tertiary level include being employed as the leader of a faculty learning engagement team and serving as a member on a teaching and learning committee. With many years of involvement across a range of education contexts, I have gathered a unique insight into the role that education opportunities and experiences can play in mediating the relationships between students and their education outcome. As such, I believe there is a possible correlation between school resources, learning environments and academic outcomes.

Like most of my colleagues I am committed to building connected relationships with the school community and improving the education outcomes and welfare of students. I utilise my pastoral care and class management skills to ensure I maintain optimal learning conditions for the students I teach. This enables me to maximise the benefit of education resources and supports my ability to integrate digital literacy skills with the Australian Curriculum.

I recognise the importance that student background, teaching strategies, learning environment and learning resources have on student academic performance. To learn more, I have undertaken research to investigate how schools that are geographically distributed across Australia's rural, remote and metropolitan communities vary in terms of school resources and learning environments. Gathering knowledge in the way school communities across Australia vary by school characteristics and student learning is of vital importance as it will lead me to better understand the role of education experience and opportunities and influence of inequity of learning and education disadvantage in Australia. My goal as a researcher is to understand the significance, inter-connectedness and impact of these trends on student experience and learning outcomes and better understand the cause of inequality of education across urban-rural school communities. However, my main objective as an educator is to connect, support, engage and inspire students to achieve their very best and endeavour to mentor less experienced educators following this same approach

Problem statement

Australian students' literacy performance in PISA reading, mathematics and science assessments, since 2000, has highlighted that education disadvantage exists in rural school communities across Australia. While it is likely that the characteristics of students explain some of the achievement gap between rural and urban students, it is plausible that the characteristics of schools also play a role. Generally, researchers are unable to identify the exact cause of the disparity between the educational outcomes of Australian urban-rural school communities. As with many social phenomena, multiple causes are likely.

Rural students in Australia typically have, on average, lower socioeconomic status than their urban counterparts (Welch, Helme & Lamb, 2007). As suggested by Young (1998), this lower socioeconomic status could explain why rural students typically have lower outcomes than their urban peers. While the relationship between socioeconomic status (SES) and student academic achievement is well documented (Caro, McDonald & Willms, 2009; White, 1982), however, the degree to which SES is associated with student learning is debatable. What is known is that the relationship between family SES and academic achievement is complex, as illustrated in an Australian study by Considine and Zappalà (2002) which examined SES in the context of variables such as gender, family structure, household income, parental educational attainment, type, and geographical location of school. Considine and Zappalà's (2002) findings suggest that the social and economic components SES may have distinct and separate influences on educational outcomes. As suggested earlier, school characteristics may also explain why rural students have on average lower educational outcomes than their urban peers. While rural schools often have difficulties attracting and retaining experienced teachers (Halsey, 2018), the degree to which their resources and learning environments are comparable to other schools is less well known. Understanding how rural and urban schools vary can therefore shed light on the causes of rural educational disadvantage, as well as possible solutions. At the same time, however, positive learning environments are ends in themselves, not just a means for fostering academic outcomes. Uncovering inequalities of educational resources, opportunities and experiences between rural and urban schools is just as important as identifying and explaining inequalities of educational outcomes.

Research questions

This study investigates how school resources and learning environments vary by geographical location of school communities across Australia, and how these variations correlate with students' academic literacy. The characteristics of school communities have also been examined to uncover more knowledge about education opportunities and experiences of students and the variation between rural-urban schools, within Australia and contrasted against Canada and New Zealand. Through analysing a range of school community variables, it is possible to gain an understanding of school resources, learning environments, academic literacy, SES, and parent education levels. It is through the analysis of these constructs that possible correlations between education opportunities, education experiences and education outcomes can be uncovered.

School characteristics refer amongst other things to the general school environment, class milieu, teaching practices, learning resources, relationships between students, teachers, administrators and parent. They are influenced by internal and external school factors, such as socio-economic status, geographic location, school type and school leadership. Academics often interchangeably use school characteristics, culture, community, and climate. According to Nias, the nature of school culture is "...often applied to school with a wilful lack of precision" (1989, p. 143). Kaplan and Owings (2013) believe many school reform efforts do not succeed because a school's culture is not considered. Whether it be school characteristics, culture, community, or school climate, these terms are broad, and according to Hoy, are based upon "...teachers' perceptions of their general work environment; it is influenced by the formal organization, informal organization, personalities of participants, and the leadership of the school" (1990, p. 151). In this study, school characteristics are reflected through the response of student and principal PISA 2009 questionnaire data.

The purpose of this thesis is to focus on PISA 2009 data relating to school community and academic outcomes with the objective of discovering to what extent principal and student perspectives on school resource and school learning environment variables differ across Australian school communities. I endeavoured to find patterns within PISA 2009 questionnaire responses, from students and principals, that highlight the association between school community and academic achievement across Australia's urban-rural school communities. International trends associated with school characteristics provide new perspectives on Australian school communities and enable each Australian school community to be compared with 'like' school communities at an international level. The following research questions have guided my study:

- How do school resources vary by geographic (urban-rural) location in Australia?
- How do school learning environments vary by geographic (urbanrural) location in Australia?
- To what extent do inequalities in educational opportunities (resources), experiences (learning environments) and outcomes (PISA scores) exist for Australian students in rural communities in comparison to other school communities distributed across rural-urban regions?
- To what extent is there rural disadvantage in educational opportunities, experiences and outcomes in Australia?
- How does rural disadvantage in educational opportunities, experiences and outcomes in Australia compare with two like countries, such as Canada and New Zealand?

Significance of study

Government funding, like the New South Wales (NSW) government's 'Rural and remote education blueprint' which is designed to improve learning across rural and regional NSW public schools, is an example of the Australian public's desire to improve equity for Australian rural school community students. However, the problem is complex. A diverse number of elements influence education outcomes, such as student and family SES, community type and school resources. To ensure that the Australian public are not shielded from the deeper issues affecting education outcome in rural Australia it is important to consider the strengths and weaknesses of rural education and understand that rural outcomes will only improve once the problem is understood.

There is extensive international research regarding rural education, but it is hard to find research that disentangles the effects of rurality from other variables, such as Indigeneity and SES or compares how learning environments and learning resource vary between rural and urban schools. However, research in this field is steadily growing with committed researchers such as Cuervo (2016), Goodwin and Kosnik (2013), Pini and Bhopal (2017) and White (2015) exploring Indigenous, social and cultural issues along with other important rural matters in the Australian education context. Previous research from Australia has shown that rural schools have difficulties attracting and retaining experienced teachers, as found by Lock, Reid, Green, Hastings, Cooper and White (2009), Thomson and De Bortoli (2008) and Welch et al. (2007). Strategies to improve student outcomes and learning experiences need to be prioritised to ensure funding and other resources are distributed appropriately and inroads are made to reduce rural-urban education disparity.

It is essential for researchers to gain a better understanding of the interconnectedness of school characteristics and the role they can play on student education experiences and the possible correlations between school resources, learning environments and academic outcomes.

This is particularly important if we are to 'bridge the gap' for students within Australia who consistently have lower educational outcomes than their peers: students from lower SES backgrounds, students in rural and remote communities, and Indigenous students (Connell et al., 2007; De Bortoli & Thomson, 2010; Lokan, Greenwood, & Cresswell, 2001; Thomson & De Bortoli, 2008). While it is well known that rural schools in Australia have difficulty attracting and retaining experienced teachers, the ways in which rural schools vary from schools in other communities, to a large extent, has not been extensively explored.

A significant aspect of my study is that it utilises PISA 2009 data to examine how schools in different geographic communities differ. There is a growing body of work that utilises international assessment programmes like PISA to highlight the performance of Australian students on the international stage, including Cresswell and Underwood (2004), De Bortoli and Thomson (2010), Hattie (2009), Haycock (2001), Lokan, et al. (2001), and Thomson, De Bortoli, Nicholas, Hillman and Buckley (2010). Having access to PISA's large, well-established and respected dataset helps to legitimise this research whilst also providing an internationally accepted way of classifying school communities across urban and rural school geographic regions. The school community classifications, outlined by PISA 2009, provides researchers with the opportunity to contrast 470,000 students across 65 countries using a common framework. Commonality of a significant number of school communities at an international level is the reason PISA was embraced as the foundation for this investigation into rurality in Australia and contrasted across Canada and New Zealand.

My study is designed to provide information about the nature and extent of rural-urban difference in Australian school communities, better understand the correlation between school community characteristics, geographic location and the academic performance of students, and raise questions for future research. International comparisons of principal and student perspectives on school characteristics are used to help gain new knowledge on the nature and possible impact of similarities and differences of school characteristics on student experience and academic outcomes. Seeing how Australian school community characteristics compare to similar countries, such as Canada and New Zealand, is important as it enables us to better understand the significance of school characteristic and other variables, such as SES and parent education, on the student experience and their literacy performance across PISA. It also will provide an indicator to the extent of rural disadvantage in Australia and highlight inequality of educational opportunities and experiences and outcomes for Australian students schooled in rural communities.

The research I have undertaken provides a measure of how schools vary in terms of resources and environments by geographic location and size (school community). My study aims to investigate whether socioeconomic status plays a lone hand in determining academic performance. And identify how school opportunities, experiences and outcomes vary across rural communities, from similar countries such as Canada and New Zealand. The information gathered through my research will provide teachers, principals and rural communities with a clear understanding of the inequity that exists across Australian urban-rural school communities. By highlighting inequity across school community in Australia I aim to encourage policy-makers in Australia to diminish the school characteristic variable disparities that exist between Australia's school communities, particularly in rural communities.

Structure of thesis

My study is structured as 'thesis by publication' comprising three separate analyses, each co-authored with my supervisors and submitted to or published in a peer-reviewed educational research journal. The analyses used data from PISA 2009 to examine how school resources and learning environments vary across rural, regional and metropolitan communities. The first two studies examined school characteristics in Australia, and the third study compared Australia with two economically, culturally and socially similar countries, Canada and New Zealand. The titles of the three studies are as follows:

- 1. How do school resources and academic performance differ across Australia's rural, regional and metropolitan communities?
- 2. How do school learning environments differ across Australia's rural, regional, and metropolitan communities?
- 3. Comparing rural educational disadvantage in Australia with two similar countries: Canada and New Zealand.

These three articles are introduced and explored in detail in subsequent chapters within this thesis. Background information is presented prior to each paper to provide context and 'significance of study' concludes each chapter, helping to provide better understanding of the issues being examined. References for all three papers are included at the end of the thesis rather than at the end of each paper.

The thesis comprises seven chapters. Chapter One introduces the thesis, Chapter Two provides an overarching review of the literature, Chapter 3 provides a general description of the method and approach, Chapter 4 presents paper #1, Chapter 5 presents paper #2, and Chapter 6 presents paper #3. Chapter 7 concludes the thesis and revisits key points, summarises findings, underlines the significance and limitations of the study and makes recommendations for policy, practice and future research.

Chapter 2:

Literature Review

Overview

In this chapter I provide a brief overview of the research literature that underpins my study. I examine literature that looks at the implication of SES and conceptualisations of rural education. Literature from Australia as well as other economically developed countries has been gathered to analyse the differences between rural and urban schooling, as they relate to education opportunities, experience and outcomes. These dimensions of educational equity are characterised in my research as educational opportunities (school resources, such as learning materials, teachers and facilities), educational experiences (learning environment, such as instructional practices and relationships between teachers and students), and education outcomes (i.e., student achievement scores on PISA 2009). The influence of positive school factors, such as abundant school resources and engaging learning environment on student achievement, is supported by a growing number of researchers (Leithwood et al., 2010; Lonsdale, 2003; Teddlie & Reynolds, 2000; Vignoles, Levacic, Walker, Machin & Reynolds, 2000). More focused and in-depth reviews of relevant literature is provided in each of the three papers, which are presented in the ensuing chapters.

Secondary analyses of PISA about rural-urban educational inequalities have been conducted in other countries. Lounkaew (2013) utilized PISA 2009 data to examine the education achievement of urban-rural students from Thailand. Lounkaew suggests that a onesize fits all solution to education reform is not possible and resources alone do not guarantee that education disadvantage can be overcome. There is also a body of evidence that suggests school resources bear little to no significance in educational outcomes. Amini and Nivorozhkin (2015) used PISA 2000-2009 data to investigate rural-urban divide in educational outcomes and conclude that higher SES is associated with higher achievement. They suggest that school resources do not play a significant role in education outcomes. Williams (2005) reported an inconsistent urban-rural difference exists for the countries in his analysis and indicates that whilst SES can be a predictor of student outcomes there is little evidence for a systematic gap. He found that the learning environment was a determinant of student ambition and achievement, even when SES was controlled and suggests the usefulness of exploring the characteristics of schools.

Significance of socioeconomic status

Socioeconomic status (SES) is a predictor of academic performance (OECD, 2010a; Reardon, 2011; Sirin, 2005). In fact, SES has long been established as a major factor, external to school, that is a precursor to academic success (Coleman et al., 1966; Mullis, Martin, Foy & Arora, 2012; Sirin, 2005). Researchers have identified that educational resources (Chiu & Khoo, 2005; OECD, 2005; Vignoles et al., 2000), experiences (Akiba, LeTendre, & Scribner, 2007; Camburn & Han, 2011; Hanushek & Rivkin, 2006) and opportunities, both at home and at school, are influenced by SES (Bourdieu & Passeron, 1990; Coley, 2002; Nash & Harker, 2006; Orr, 2003; Yeung, Linver, & Brooks-Gunn, 2002). However, it is difficult to disentangle variables associated with SES because they are inter-related: f or example, compared to their less advantaged peers, higher expectations are thought to be placed on students who come from higher SES backgrounds (Rumberger & Palardy, 2005), who in turn are often provided more rigorous academic curriculum (Lamb, Hogan, & Johnson, 2001; Oakes, 2000). Thus, it is likely that socioeconomic status is related to academic achievement through both family, home and school related factors.

Factors external to school may impact more significantly on student achievement than school inputs (Chudgar & Luschei, 2009; Konstantopoulos & Borman, 2011; Woessmann, 2016). Ramos, Duque, and Nieto (2012) utilised PISA 2006 and 2009 data to examine Colombian rural school communities. Their research suggests the nucleus of the home rather than the school community is the key to improving student academic outcomes. Tayyaba (2011) utilized a Pakistan national assessment survey in her investigation. She found the availability of academic resources made little difference to student achievement across ruralurban schools. Collectively these studies highlight the need for the interconnectedness of SES, home factors, school resources and school environment variables to be better understood.

The dangers of using large scale international assessments, like PISA, to link academic achievement to SES, is uncovered by Schmidt and Burroughs (2013). They explored the role of introducing 'opportunity to learn' (OTL) indicators to discover the impact of school factors on student achievement at a deeper level. White (1982) considered the correlation between student SES and academic achievement weak. However, statistically students from high SES backgrounds typically enjoy more positive learning environments and better educational resources than low SES counterparts (OECD, 2005; Orr 2003; Thomson, 2002).

In the Australian context, students in rural environments have been found to be less economically advantaged than their urban peers (Lamb, 1994). Low SES in both the school and home are predictors of educational outcomes (Noel & de Broucker, 2001; Rothman & McMillan, 2003). And compared to their more privileged peers, low SES students often experience at home fewer learning opportunities that are aligned with success at school (Nash & Harker, 2006; Yeung et al., 2002). Perhaps this is due to a lack of resources, such as time. Coley (2002) suggests the parents of low SES students are less likely to read to their children. Given this scenario the importance of maintaining high quality school resources and learning environments is imperative. A sub-Saharan study by Zhang (2006) is one of many international studies, like Opoku-Asare and Siaw (2015) and Ramos et al., (2012), which support the notion that school resources are important determinants of rural-urban gaps in student achievement.

Educational opportunities

The Coleman Report is a, "...model of equality of educational opportunity focused on student outcomes" (Jacobs, 2016, p. 319). PISA establishes opportunity to learn variables that can be used to link school effectiveness based on PISA student performance across reading, mathematics and science to equitable resource distribution (OECD, 2010). Shields, Newman and Satz (2017) express:

> Educational opportunities are those opportunities that aim to enable individuals to acquire knowledge and certain skills, and to cultivate certain capacities... we associate the goals that constitute educational opportunities with access to educational institutions such as schools... (3.1 What is Educational Opportunity? section, para. 4).

In this study, the term 'education opportunity' supports the views of (OECD, 2010) and Shields, Newman and Satz (2017) and is measured through analysis of PISA questionnaire school resource variables, including learning materials, education equipment, infrastructure and teachers. In this way my research supports the philosophy of Shields (2015), who states, "...equality of opportunity can offer guidance and assessment of the design of educational institutions" (p. 54).

It is well-known that the quality of school resources plays a role in the development of the learner (Chiu & Khoo, 2005; OECD, 2005; Vignoles et al., 2000). Instructional material and qualified and experienced teachers are associated with student achievement (Chiu & Khoo, 2005; Haycock, 2001; Wenglinsky, 2002). Lower SES schools around the world, on average, are less resourced than higher SES schools (Bowles & Levin, 1968; Centra & Potter, 1980; Chiu & Khoo, 2005). However, whilst it may seem like a simple problem to solve, the solution to over-come resource inequity is complex. Some researchers have questioned whether establishing resource equality across low and high SES school communities is enough to overcome outcome inequities as it is difficult to compensate for limited access to resources outside the school (Minguez & Ballesteros, 2008).

McMorrow (2011) believes school funding reform is vital. At the political level the trend toward improving the educational opportunities for Australian students has been pursued through the introduction of an Australian Curriculum in 2008, a commitment that is supported by the 2008 Melbourne Declaration on Educational Goals for Young Australians Ministerial Council on Education, Employment, Training, and Youth Affairs (Barr et al., 2008), the Gonski Report (Gonski, Boston, Greiner, Scales & Tannock, 2011), an increased focus on national and international testing, and emphasis on school accountability via the My Schools website. However, whilst the government is committed to ensuring the next generation of working Australians is well placed in a global economy, the government's ability to provide a clear understanding of urban-rural opportunity divide is less defined.

Young (1998) led an Australian rural education investigation that determined that whilst school effect, associated with educational opportunities, on student learning is small, there is a great deal of variance left unexplained. For instance, 'shortage of teachers' is believed to impact on learning. Vinson, Esson and Johnston (2002) and Welch et al. (2007) found that some Australian rural schools face teacher shortages. Teacher shortage is associated with the disparity between education opportunities in rural and urban schools (Clotfelter, Ladd, Vigdor & Wheeler, 2007; Darling-Hammond & Ball, 1997; Darling-Hammond, Holtzman, Gatlin & Vasquez Heilig, 2005; Haberman, 2006; Hanushek, 2007; Hattie, 2009; Ofsted, 2000). Issues raised by Akiba et al. (2007) emphasize the importance of ensuring equality of access to qualified teachers across low and high SES school communities. Cresswell and Underwood (2004) and Thomson and De Bortoli (2008) utilised PISA data to show that Australian rural schools have difficulties attracting and retaining experienced teachers. Student learning and achievement are associated with school resources, such as qualified and experienced teachers (Greenwald, Hedges & Laine, 1996; Haycock, 2001; Wenglinsky, 2002) and positive learning environments (Aultman, Williams-Johnson & Schultz, 2009; Frempong, Ma & Mensah, 2012; McHugh, Horner, Colditz & Wallace, 2013; National Research Council, 2003; Newberry, 2010; Spilt, Hughes, Wu & Kwok, 2012). Teachers' knowledge positively predicted student academic gains (Hill, Rowan & Loewenberg Ball, 2005) and a larger teacher effect variance was found in low SES schools than high SES school (Nye, Konstantopoulos & Hedges, 2004).

Education experiences

School learning environment, as used in this research, includes a range of student and teacher relationships, behaviours, attitudes and expectations. It can include associations between teachers and students, school climate, and classroom disciplinary climate. In the context of my investigation it includes classroom and school disciplinary climate, student and staff communication, expectations and absenteeism. The data utilised in this research are based upon principal and student perceptions of their school's learning environment. The data were gathered via principal and student PISA 2009 questionnaires. Principals' and students' perception is based upon their responses to survey questions relating to school, teachers, students and the classroom. Although interconnected these constructs can be broken into three categories: school learning environment, student and teacher behaviours and the classroom learning environment.

A strong interplay between school climate, learning resources and educational policies is suggested to exist in Australian schools (Angus, 1993; Connell, Ashenden, Kessler, & Dowsett, 1982; Human Rights and Equal Opportunity Commission, 2000; Smyth, McInerney, & Hattam, 2003). The characteristics and motivations of rural high school students has been researched by Hardré, Sullivan and Crowson (2009), who found that a student's perceived scholastic confidence most strongly predicted interest and achievement. The relationship between student and teacher, particularly a student's attitude towards teachers as a measure of academic self-concept, was explored by Haslett (1976). She found student interpersonal effectiveness to be the most significant overall predictor of high school students' attitudes towards teachers. Nussbaum (1992) examined literature relating to effective teacher behaviours, highlighting that effective teaching practices and instructional processes made a difference. High quality instructional practices have been shown to engage students and promote student outcomes (Akiba et al., 2007; Hanushek, 2007; Hogrebe & Tate IV, 2010; Winheller, Hattie, & Brown, 2013). However, despite analysis on students, learning, relationships and teaching practices, little is known about the degree to which the learning environments in rural schools differ from other schools in Australia.

Student learning is seen to be promoted by supportive and caring teachers who have high expectations (Auwarter & Aruguete, 2008; Hardré & Reeves, 2003; National Research Council, 2003; Rosenfeld, Richman, & Bowe, 2000), and inspiring teachers (Hardré & Reeves, 2003). Positive, supportive and caring relationships between teachers and students have been shown to improve students' outcomes because they increase students' motivation and engagement with learning (Aultman et al., 2009; McHugh et al., 2013; National Research Council, 2003; Newberry 2010; Spilt et al., 2012). Many researchers, such as Diseth (2007), Hoy, Tarter and Hoy (2006) and Stewart (2008), believe that a positive and cohesive school learning community promotes high academic expectations.

It is clear that school-related factors, such as student and teacher professional relationships, can also provide greater understanding of the inequity associated with the geographic divide in academic outcomes across Australian school communities. At the heart of teacher and student interactions in the classroom is the management skills of the teacher. Classroom disciplinary climate is viewed by many researchers as a strong predictor of student
outcomes (Schleicher, 2009). Randhawa and Michayluk found that Canadian urban classrooms "...meet the needs of the learner in such a way that they perceive their learning experiences sufficiently satisfying" (1975 p. 277). What is largely agreed upon is classrooms that have fewer distractions promote more opportunities for teaching and learning (Frempong et al., 2012; OECD, 2005; Shin, Lee, & Kim, 2009). Moreover, high quality instructional practices can promote student outcomes (Akiba et al., 2007; Hanushek, 2007; Hogrebe & Tate IV, 2010; Winheller et al., 2013).

Education outcomes

Education outcomes include scores on standardized tests, grades on school-based assessments, and school completion rates, among other measures. The Australian government uses national and international testing to gather information on the performance of students. These various tests have shown that Australian students in rural and remote regions do not achieve at the same level as their city peers (Cresswell & Underwood, 2004; Thomson, Cresswell & De Bortoli, 2004). Studies of students from lower socioeconomic backgrounds, which include many Indigenous and rural and remote students, have shown that these students typically achieve lower educational outcomes than their more privileged peers in Australia and elsewhere (Noel & de Broucker, 2001; OECD, 2010a; Sirin, 2005; Teese & Polesel, 2003).

Students from rural school communities, on average, achieve lower results on standardized tests of academic performance (Cresswell & Underwood, 2004) and compared to urban schools, the administration of rural schools have expressed difficulty recruiting and retaining teachers (Ministerial Council on Education, Employment, Training and Youth Affairs, 2003; Vinson, Esson, & Johnston, 2002; Yarrow, Ballantyne, Hansford, Herschell, & Millwater, 1999). In Australia, Young (1998) led an Australian rural education investigation that determined, classroom learning environment has a strong effect on student self-concept, which links to student ambition. She also identified that SES has a positive effect on ambition but not self-concept. However, self-concept was found to affect both ambition and achievement. This research helps educators better understand how to work towards lowering retention rates, which in rural high schools are lower than their city counterparts (Godden, 2007; Marks & Fleming, 1999), and reverse the trend where rural high school students are less likely to graduate and attend university (Alston & Kent, 2006; James, 2001; Marks, Fleming, Long, & McMillan, 2000). Collectively, this research highlights the need for school characteristics to be better understood to help prevent inequity in learning experience for rural school communities.

Rurality in the Australian context

Whilst it is difficult to establish a shared definition for the term 'rural education', either in Australia or amongst international researchers, it is common to examine rurality in terms of population size, population density and distance from a major city. Black (2005, cited in Alston, 2007), highlights the need for a unified view of what it means to be a rural student. The Australian Bureau of Statistics, 1998 (ABS), defines a rural community as a location where people live in clusters of less than 1,000 people. Like the ABS, the Organisation for Economic Cooperation and Development (OECD), which coordinates PISA, utilises population size to distinguish geographic regions.

An insight into education disadvantage in rural school communities in Australia was gained through exploring the Australian Human Rights and Equal Opportunity Commission's Report into Education in Regional and Rural Australia (Human Rights and Equal Opportunity Commission, 2000). The report found that rural schooling in Australia was inferior on every indicator included in their study. Webster and Fisher (2000) highlight the need for a better understanding of school community types across the urban-rural continuum,

...research in the area of rural and urban differences should take into account more categories of location

which would include remote, semi-rural, outer suburban and inner-city as all these locations have characteristics which are specific to the very location (p. 358). Attracting teachers to rural areas is an ongoing concern for many governments. In

Australia there has been a push to have rural education incorporated into teaching degrees to better prepare graduate teachers for rural placement. Beutel, Adie and Hudson (2011) found that preservice teachers' perceptions of teaching in rural communities can be better informed through the introduction of a structured rural teaching experience. A similar education reform is promoted by White (2008), who sees the value of fostering well-trained teachers who are personally and professionally equipped to address the educational needs of rural communities.

Conclusion

This literature review has summarised the significance of education opportunities, experiences and outcomes, especially in the rural-urban context. It is well established in Australia that students who attend schools in rural areas have lower educational outcomes than their urban peers. It is also well established that rural schools in Australia have difficulty attracting and retaining experienced teachers, although the degree to which this is different from urban contexts is not clear. Much less is known about the differences between rural-urban schools, especially regarding school resources and learning environments. The PISA dataset provides rich information about school resources and learning environments across Australia's school communities. As such, it is very useful for mapping how schools in rural and urban contexts differ.

Understanding how school resources and learning environments vary across school communities is an important first step for identifying how schools can be improved. These improvements may help reduce inequalities of outcomes between students in rural and urban contexts. It is just as important, however, to map the nature and extent of rural-urban inequalities of educational opportunities and experiences because they are an end in themselves and not just as a predictor of academic outcomes. All students deserve the right

to enjoy orderly classrooms or positive relationships with their teachers, regardless of where they live. It is also plausible that students in rural contexts experience more positive school experiences compared to their urban peers. Understanding how schools in different ruralurban contexts vary can not only highlight how schools can be improved, but also showcase the strengths of rural schools, which in turn could enhance their ability to attract and retain experienced teachers.

Chapter 3:

Methodology

This thesis investigates how school resources and learning environments vary across rural, regional and metropolitan school communities within Australia, and contrasts findings with two countries of similar historic, economic and cultural diversity: Canada and New Zealand. In this chapter, I describe the general methodological approach, discussing the research design, data source and analytical strategy. More specific methodological detail is provided in each of the papers that comprise the next three chapters.

Research design

I used a descriptive research design to examine quantitatively how schools vary across rural-urban communities, from the perspective of school principals and students. My aim was to produce a detailed and comprehensive mapping of the differences between schools, focusing on the nature and extent of these differences. By using a large and nationally representative dataset, I was able to create a more comprehensive and accurate mapping of rural-urban school differences than has been available to date. My mapping of differences between school communities will be useful for uncovering educational inequalities and for identifying areas for further research.

Data source

I used data from PISA to answer the study's research questions. PISA was launched by the OECD in 1997, with the aim of examining the reading, mathematical and scientific literacy of 15-year-old students within the OECD member and partner countries. PISA "...examines how well students are prepared to meet the challenges of the future, rather than how well they master particular curricula" (OECD, 2012, p. 3). PISA also includes a wide range of variables about students, classrooms and schools that may be related to student performance. The aim of PISA is to provide researchers and policy-makers with an internationally comparative evidence base for guiding educational reform and uncovering different ways of organizing schools and learning.

Through its rich collection of data across many countries, PISA can be used to identify educational structures and practices for enhancing the educational outcomes of students from disadvantaged backgrounds (Thomson, et al., 2010). Measuring student literacies (academic performance) on a regular basis is designed to ascertain the opportunity for students to learn. The OECD Assessment Framework (2010, p. 168) states,

> A central pre-occupation of PISA is that of fairness or justice in access to education and the opportunity to learn...The second concern is that of how to measure equity. Equity can be assessed in terms of the distribution of access to schooling, learning resources and opportunities, and educational outcomes.

PISA questionnaires are designed to gather information on school resources and learning environments and present data to highlight equity in the distribution of learning opportunities and show what is possible in terms of education outcomes in reading, mathematics and science.

PISA is administered every three years to a nationally representative sample of 15year-old students and schools in all OECD member countries and a growing number of voluntary non-member countries. Each cycle measures student literacies in all three subject domains. In addition, each cycle measures one of the domains in-depth, with the focused domain alternating between cycles. I used data from the 2009 cycle since this was the most recent cycle when I commenced my studies. The subject of focus in the 2009 cycle was reading. Reading is useful for measuring educational inequalities because it is a fundamental competency that undergirds all learning.

PISA is a very large dataset. Approximately 470,000 15-year-old students from 65 countries and economies participated in PISA 2009. Between 5,000 and 10,000 15-year-old students, from at least 150 schools, were typically tested in each country. The number of

Australian students included just over 14,000 from 353 schools (Thomson et al, 2010). In Canada, the approximate number of participating 15-year-olds was 23,000 students from 1,000 schools spanning across the ten provinces (Knighton, Brochu & Gluszynski, 2010). New Zealand's sample comprised 4,643 students from 163 schools (Telford & May, 2010).

PISA comprises data from students and school principals. In addition to the cognitive literacy assessment, students complete a 30-minute questionnaire about their individual characteristics, parents' backgrounds and characteristics, home resources, and attitudes to school. The 2009 cycle also asked students questions that are specific to reading engagement and instruction, such as their individual engagement with reading, classroom and school climate, views on their native speaking language lessons, teachers' instructional strategies, access to and use of libraries and strategies in reading and understanding text (OECD, 2012). Principals from schools selected in the survey also answered a 30-minute questionnaire. This questionnaire is designed to gain an understanding of the level of school resources and provides information on the school environment and qualifications of staff (Thomson et al., 2010). The School Questionnaire contains questions on the structure and organisation of the school, the student and teacher body, the school's resources, the school's instruction, curriculum and assessment, the school climate, the school policies and practices and the characteristics of the principal or designate (OECD, 2012).

Variables

The PISA 2009 dataset utilised in this thesis comprises 470,000 students across 65 countries. Most countries typically sample between 5000-10,000 students, but to allow for comparisons between jurisdictions (states and territories), Australia sampled a larger number of students. The large sample size ensures that there is a wide cross-section of students, representative of each of Australia's states, important groups within the population (e.g., Indigenous Australians), and distinct school communities.

PISA 2009 geographically categorises the geographic communities from which schools and students derive as follows:

- i) villages, hamlets or rural areas with fewer than 3,000 people;
- ii) small towns with 3,000 to 15,000 people;
- iii) towns with 15,000 to 100,000 people;
- iv) cities with 100,000 to one million people; and
- v) large cities with over a million people.
 - (OECD, 2011, p. 56)

The five geo-location categories used in OECD's PISA 2009 reports differed from those utilized by the Australian Council for Educational Research (ACER), which administers PISA in Australia. Rather than five categories, ACER subdivided some of the categories, to make a total of eight community categories. I made contact with ACER and requested the geographic location coding for School Community that ACER had applied to the Australian PISA 2009 dataset. Once permission and access were granted to this dataset, I imported school characteristic variables into an ACER version of the PISA 2009 source file. This allowed me to conduct secondary analyses using the same geographic communities used by ACER.

Table 1 above outlines the distinctions in school community population size present in the Australian PISA 2009 data as utilized by ACER and the Programme for International Student Assessment. I used the eight community categories for papers 1 and 2. I used the five community categories for paper 3 so that comparisons could be made with Canada and New Zealand.

ACER Filtered PISA 2009 data		Raw PISA 2009 data		
School Community	Population size	School Community	Population size	
Small rural community	< 1,000	Village	< 2.000	
A small country town	1,000 to about 3,000	Village	< 3,000	
A medium-sized country town	3,000 to about 15,000	Small town	3,000 to about 15,000	
A larger town	15,000 to about 50,000		15,000 to about 100,000	
A very large town	50,000 to about 100,000	Town		
A city	100,000 to about 1,000,000	City	100,000 to about 1,000,000	
Elsewhere in a very large city	> 1,000,000			
Close to the centre of a very large city	> 1,000,000	Large City	> 1,000,000	

Table 1. Overview of ACER and PISA classification of school community

The variables used to provide information on school resources and learning environments across urban-rural school communities stem from PISA 2009 student and principal questionnaire data whilst student outcomes are measured against reading, mathematics and science mean literacy performance in PISA 2009. Other interesting PISA data are incorporated into my study to increase understanding of student, school and family, including SES, Indigeneity and highest parent education level attained.

When investigating school resources, I utilised the principal questionnaire survey. To gain an appreciation for school community resource availability I accessed the information pertaining to the ratio of student numbers to computers. To gain a stronger appreciation for the distribution of school resources across school communities I examined the principal questionnaire survey response to: shortages of teaching resources (science laboratory equipment, instructional material, computers, Internet, computer software, library materials and audio-visual materials). This information was supplemented by principal responses to questions focused on staff, including: lack of teaching personnel (science teachers, mathematics teachers, English teachers, qualified teachers, library staff and other personnel).

When identifying school learning environment, principal and student questionnaire data were utilised. Principals' perceptions of school learning environment are based on their responses to the following questions: To what extent is learning of students hindered by such things as teacher's low expectations; student absenteeism; student teacher relations; student disruptions; student's needs not met; and teacher absenteeism? To better understand school learning environment, students were asked to respond to the following: school has done little to prepare me for adult life; school has been a waste of time; school has helped give me confidence to make decisions; and school has taught me things that could be useful in a job.

To gather information on the school environment I utilised PISA student questionnaire data that measured student and teacher relationships. Students were asked to indicate their level of agreement with items like: I get along well with most of my teachers; most of my teachers are interested in my well-being; most of my teachers really listen to what I have to say; if I need extra help I will receive it from my teachers; and, most of my teachers treat me fairly.

In my research, student attitudes towards the classroom environment was determined through analysis of student questionnaire responses to classroom disciplinary climate. Students were asked to indicate their agreement with these five items: students don't listen to what the teacher says; there is noise and disorder; the teacher has to wait a long time for the class to quiet down; students cannot work well; and students don't start working for a long time after the lesson begins.

The PISA 2009 dataset includes a SES measure for each student, which PISA calls ESCS (economic, social and cultural status). I created a new variable for each participating school to represent an average of the student-level ESCS scores, from that school: for example, if 30 students completed PISA 2009 at school X, the students' ESCS scores were averaged to create a 'mean school SES' variable. Next, the school SES variable was used to divide the participating schools in each country into quintiles arranged from lowest to highest school SES.

Analytical strategy

My aim in all three papers was to compare school characteristics across school communities. To this end, I used descriptive statistics such as means, standard deviation and frequencies. Means and frequencies were compared across the Australian school communities in papers 1 and 2. In paper 3, means were compared across school communities within and between Australia, Canada and New Zealand.

Questionnaire items in PISA typically have four responses. The four responses vary by item. Some items have four responses that range from "strongly agree" to "strongly disagree"; another variation is four responses that range from "every lesson" to " never/hardly ever". A third variation is as follows:

- 1: Not at all;
- 2: Very little;
- 3: To some extent; and
- 4: A lot.

For some items, I calculated means and frequencies for each of the four responses: for example, I calculated the proportion of students that responded, "to some extent" compared to the proportion of students that responded, "a lot". In other instances, I collapsed response categories into two larger categories (e.g., strongly agree and agree, vs strongly disagree and disagree) and then reported frequencies for those larger categories. In other instances, PISA creates a numeric index that is comprised of multiple items: for example, "classroom

disciplinary climate" is an index that is comprised of five separate items. I calculated and compared means for some of these index variables as well, especially in the third study. Specific details about the variables and analytical strategy used in each study is described in the relevant paper.

For the Australian context, I calculated frequencies of the school-level variables for the eight different school communities as utilized by ACER for papers 1 and 2. This comparison provided a comparison of school learning environment and resource characteristics by geographic location (e.g., percentage of schools that experience teacher shortages in remote locations versus urban locations). In addition to the comparisons of school learning environment and resource characteristics these different school contexts.

Clarification of terms

This thesis utilises language contained within the OECD (2009) data analysis manual, focusing on two key subsets of school community: school resources and school learning environments. During my research I came to realise that the classification for seemingly simple terms, such as rural education, school resources and learning environments within educational research varies considerably. I utilised the OECD's Glossary of Statistical Terms (OECD, 2007) to provide a clear understanding of variables utilised in my investigation.

Index of economic, social and cultural status (defined as SES in this research) is created using the following variables: the highest level of education of the student's parents, the PISA index of family wealth, the PISA index of home educational resources, and the PISA index of possessions related to classical culture in the family home.

School location refers to the community in which the school is located, such as a village, hamlet or rural area (fewer than 3,000 people), a small town (3,000 to about 15,000

people), a town (15,000 to about 100,000 people), a city (100,000 to about 1,000,000 people), close to the centre of a city with over 1,000,000 people.

School resources includes instructional materials such as the school library, calculators, computers, the internet, and laboratories. It also includes teaching and other instructional staff.

Learning environment, as classified by PISA, includes: teacher and student behaviours that affect learning, the disciplinary climate, teacher-student relations, how teachers stimulate students' engagement in reading, parents' involvement in and expectation of schooling, and school principals' leadership. (OECD, 2010b, p.56)

In this chapter, through the analysis of methodological approach, research design, data source and analytical strategy I have endeavoured to highlight the thoroughness of my thesis research, the relationship between investigations and the complexity of this task.

Chapter 4:

Paper 1: How do school resources and academic performance differ across Australia's rural, regional and metropolitan communities?

Background

Focused within the Australian context, my first article, titled "How do school resources and academic performance differ across Australia's rural, regional and metropolitan communities?" was published in the *Australian Educational Researcher* (AER) in April 2013. The AER is the flagship journal of the Australian Association for Research in Education. My initial analysis of PISA 2009 student data raised awareness that some of the status categories my research wanted to encapsulate, such as student Indigenous status, were not available for public viewing in the OECD PISA 2009 dataset. Consequently, I requested access to Indigeneity data from ACER. I then integrated the data into the database that I was utilising. The endeavour for my first paper was to gain a better understanding of three groups of Australian students (those from Indigenous, rural, and low socioeconomic backgrounds) performing at significantly lower academic levels in comparison to their peers in PISA 2009 reading, mathematics and science. Analyses of PISA 2009 data provided information on school resources that helped to contextualize academic achievement across Australian school communities.

When investigating school resources, I utilised the PISA 2009 principal questionnaire which seeks perceptions from principals of their school community's learning resources. To gain an appreciation for school community resource availability I accessed the information pertaining to student numbers to computers ratio. To gather a more detailed understanding of the distribution of school resources across school communities I examined the principal questionnaire survey response to shortages of teaching resources. This information incorporated the following areas: science laboratory equipment; instructional material; computers; Internet; computer software; library materials; and audio-visual materials. These data were supplemented by principal responses to questions focused on staff, including: lack of teaching personnel by school community. Principals provided information on: science teachers; mathematics teachers; English teachers; qualified teachers; library staff; and other personnel.

To support my findings, I researched current literature on the significance of learning resources and gained a better understanding of the inter-relatedness of learning resources with other variables and the possible implications for learning outcomes. My research of the PISA 2009 Australian principal questionnaire dataset revealed that principals of schools in small towns indicate that their school communities have fewer resources than schools in very large cities. I discovered that many principals, especially those in less populated school communities, report instruction that occurs within their school is hindered to some extent by 'a lack of resources', in particular, 'shortages of teaching personnel'. By utilising the PISA SES variable, I also learned that there appears to be a positive association with school SES and students' academic performance in reading, mathematics and science.

How do school resources and academic performance differ across Australia's rural, regional and metropolitan communities?

Kevin Sullivan · Laura B. Perry · Andrew McConney

Received: 28 November 2012/Accepted: 11 June 2013/Published online: 21 June 2013 © The Australian Association for Research in Education, Inc. 2013

Abstract This study uses data from the Programme for International Student Assessment (PISA) to gain a better understanding of how academic performance and resources vary across rural-urban school communities in Australia. While it is well known that schools in rural areas have difficulty recruiting and retaining teachers, the degree to which schools in larger sized communities across Australia also face this problem is less understood. Moreover, very little is known about the degree to which shortages of instructional materials and equipment are associated with rural-urban location. The analysis includes 353 schools across eight community types that range in size of <1,000 people in small country towns to more than a million people in large capital cities. School principals reported the degree to which instruction in their school is hindered by a shortage of resources, which include qualified teaching staff and instructional materials and equipment. The findings highlight the extent to which school resources vary across geographic location, as reported by school principals. Principals of schools in the centre of large cities were the least likely to report that shortages of teaching staff or instructional materials hinder learning, while principals in rural and remote communities were the most likely to report that such shortages hinder instruction. These differences closely mirror student PISA academic performance and school socioeconomic composition. PISA data indicates that schools located in small rural communities have the lowest

Curtin University, Bentley, WA, Australia e-mail: kevin.sullivan@curtin.edu.au

L. B. Perry · A. McConney Murdoch University, Murdoch, WA, Australia e-mail: 1.perry@murdoch.edu.au

A. McConney e-mail: a.mcconney@murdoch.edu.au

K. Sullivan (🖂)

socioeconomic profiles, the lowest academic performance and the largest shortages of teaching staff and instructional materials, while schools in central neighborhoods of large cities enjoy the highest socioeconomic profiles, the highest academic performance and the fewest shortages.

Keywords Geographic location · School resources · Rural education · Academic performance · PISA

Introduction

This study adds to our understanding about the shortage of school resources in rural communities by comparing rural, regional and urban school principals' responses on the *Programme for International Student Assessment* (PISA), an international assessment created by the Organization for Economic Cooperation and Development (OECD). PISA is an international assessment of reading, mathematics and science literacy designed for 15-year-old students. The performance of Australian students taking part in PISA 2009 has been characterized by the OECD as high performing and equitable compared to other countries (Thomson et al. 2010). However, a recent report for the Review of Funding for Schooling in Australia by the Nous Group (2011) suggests Australian students' performance in recent international tests:

...masks a wide degree of variability within our education system. That variability relates to educational outcomes, and to equity—that is, the degree to which people from all backgrounds are able to realise their potential in school. (p. 5)

So whilst Australian students on average display positive educational outcomes, PISA data also indicate that three groups of students in Australia consistently have lower academic performance than their peers: students with lower socioeconomic status (SES), students in rural and remote communities, and Indigenous students (De Bortoli and Thomson 2010; Lokan et al. 2001; Thomson and De Bortoli 2008). Our focus in this paper is on analysing PISA questionnaire data concerning students and schools in rural and remote communities, including data provided by school principals.

According to an Australian government document (Baxter et al. 2011), Australia's population in 2009 was just <22 million people living in the following community types:

Over two-thirds (69 %) of Australians live in major cities, one in five (20 %) live in inner regional areas, one in ten (9 %) in outer regional areas and around. one in forty (2.3 %) live in remote or very remote areas (1.5 % remote and 0.8 % very remote). (p. 1)

Educational opportunities and outcomes are limited in many rural and remote communities. In terms of educational opportunities, the Australian Human Rights and Equal Opportunity Commission (2000) states:

State and Territory education departments provide primary schools in rural and remote locations once there is a critical mass of primary aged children. A remote community of fewer than 1,000 people is unlikely to be provided with a secondary school. Some 'primary' schools extend their provision beyond year 6 or 7 to year 8 or 9 and sometimes to year 10. Secondary provision to year 12 is almost non-existent in remote communities. (p. 11)

Australian students who attend schools in rural and remote communities experience lower educational outcomes than their peers in the cities (Human Rights and Equal Opportunity Commission 2000). They are less likely to attend university (James 2001), less likely to finish secondary school (Lamb et al. 2004), and have poorer performance on achievement tests (Williams 2005). In their analysis of PISA 2000 data, Cresswell and Underwood (2004) found that:

...students in remote areas are not achieving at the same level as their city counterparts....It was found that 27 per cent of students from remote areas were achieving at the two lowest levels, compared to 12 per cent of students from major cities. At the other end of the scale, 18 per cent of remote students achieved at the two highest levels, compared to 46 per cent of the city students. (p. 33)

Cresswell and Underwood (2004) found similar patterns in their analysis of PISA 2003 data for Australia.

The reasons why students in rural and remote communities have lower educational outcomes than other students are complex and varied. Family background is a strong predictor of educational outcomes. Numerous studies have shown that students from lower socioeconomic backgrounds, which include many Indigenous and rural and remote students, typically achieve lower educational outcomes than their more privileged peers (Noel and de Broucker 2001; OECD 2010; Sirin 2005; Teese and Polesel 2003). Although international and Australian research has consistently shown that individual level factors such as socioeconomic status and home environment are the largest predictors of educational outcomes (Noel and de Broucker 2001), school resources are also important (Chiu and Khoo 2005; OECD 2005; Vignoles et al. 2000).

Rural and urban funding equity issues have come to the fore in Australia partly due to the Gonski Review (Gonski et al. 2011), a major education funding review commissioned recently by the federal government. The review proposes a more balanced and equitable funding formula to reduce large resource inequalities between schools and to ensure that all schools receive adequate funding to meet the needs of their students. The need for school funding reform in Australia is vital as explained by McMorrow (2011): "Constructing national recurrent target resource standards for schools....would be a major step towards the development of a funding model for schools that has integrity, rationality and sustainability" (p. 15).

It is indeed the case that rural schools often receive higher per-pupil funding than urban schools because they are more expensive to operate due to their small size, and because they often enrol a larger proportion of at-risk students who receive higher funding (e.g., Indigenous students). For example, the federal government's My School website (Australian Curriculum Assessment and 2013) shows that Narrogin Primary School, located in a small rural community of ~4,200 people, has a net recurrent income of \$14,139 per student, while Mandurah Primary School, located in a city of more than 83,000 people, receives \$12,359 per student. The larger per-pupil funding in rural schools is not necessarily sufficient to provide an equitable distribution of school resources, however. It may be the case that rural schools need an even higher per-pupil funding in order to have a comparable level of teaching and learning resources.

In this study, we use questionnaire data from PISA 2009 to gain a better understanding of the extent to which school resources, vary according to where schools are geographically located. The school resource variables included in PISA 2009 relate to shortages of teaching staff, materials and equipment, as reported by school principals. Our primary aim is to examine differences in school resources across rural–urban locations as reported by school principals. Although it is well known that schools in rural and remote communities routinely experience high turnover of teachers and principals (Vinson et al. 2002), much less is known about how shortages of teaching staff, materials and equipment may vary across different types of communities in Australia.

Regardless of whether school resources are significantly related to students' educational outcomes or not, resources amongst Australian schools must be distributed across schools in a manner that ensures equality of access and opportunity for all students, in accordance with the National Declaration On Educational Goals For Young Australians (Ministerial Council on Education, Employment, Training and Youth Affairs 2008). All Australian students (Barr et al. 2008) have the right to:

...equality of opportunity to access and participate in high-quality schooling that is free from discrimination based on gender, language, sexual orientation, pregnancy, culture, ethnicity, religion or disability, and differences arising from students' socioeconomic background or geographic location. (p. 6)

A secondary, related objective of this study is to examine how academic performance (as measured by PISA) varies across a wide range of rural–urban locations in Australia. Again, although it is well established that students in rural communities tend to perform less well than their urban peers, less is known about how the overall academic performance of schools varies by location. Is the relationship between school academic performance and community size, type or urbanicity consistently positive, or not? And, how do these relationships look when school and student socioeconomic composition are added to the mix?

Background

Educational outcomes are influenced and mediated by a complex web of factors derived from multiple sources, including the student, family, peers, community, school and the dominant culture within a society. A particular set of factors, namely those reflective of a school's resources, is the focus of this study. School resources include instructional materials, infrastructure and teaching staff. Previous research has reported that school resources and learning environments are strongly associated with educational outcomes (Chiu and Khoo 2005). Of all school resources, most researchers agree that qualified teachers are the most important for student learning (Akiba et al. 2007; Darling-Hammond and Ball 1997; Darling-Hammond et al. 2005; Hanushek 2007; Hattie 2009).

Researchers for decades have noted that school resources are strongly correlated with both school and student SES, which leaves open the possibility that the importance of school resources is underestimated (Bowles and Levin 1968; Centra and Potter 1980). Chiu and Khoo (2005), among many others, have reported that higher SES schools are, on average, better resourced than lower SES schools in most countries, including Australia. Compared to schools that enroll students with mainly higher SES backgrounds, schools with large concentrations of students from low SES backgrounds have fewer teaching resources (Chiu and Khoo 2005; Tate 1997), have more difficulty recruiting and retaining teachers (Darling-Hammond 2009; Haberman 2006), and have fewer certified teachers (Clotfelter et al. 2007; Hanushek 2007; Ofsted 2000).

Within Australia and across the world the term 'rural' is defined in different ways. Black (2005, cited in Alston 2007), notes that "'Rural' is a highly contested term in Australia because of the diversity of population and geography" (p. 196). Many people question the rurality of Australia's lush coastal regions in comparison to the sparse Australian outback. PISA categorises school communities based on geographic location, taking account of the population size of the community and its distance from the nearest city. The context by which PISA classifies a school community's geographic location is not reflective of the community's proximity to the ocean, the quality of infrastructure or economic development.

In Australia, schools in rural and remote locations face many challenges, especially regarding teaching staff. Vinson et al. (2002) and Welch et al. (2007) have found that rural schools in New South Wales face teacher shortages. Analyses of PISA data indicate that rural schools in Australia have difficulties attracting and retaining experienced teachers (Cresswell and Underwood 2004; Thomson and De Bortoli 2008). None of these studies, however, has shown in detail how resources vary across schools in different locations. This study adds to our understanding about the shortage of school resources in rural communities by comparing principals' responses from rural, regional and urban communities.

While student factors are probably more important than school-level factors in predicting academic performance, the latter are nonetheless important. For example, Rothman and McMillan (2003) report, "Approximately less than one-sixth of the variation in scores on tests of reading comprehension and mathematics [tertiary entrance scores] could be attributed to differences between schools..." (p. 30). Student background characteristics do not explain all of the differences in educational outcomes between students in different geographic locations, however. Young (1998) found that students who attend rural and remote schools in Western Australia, a sparsely populated state, have lower academic performance than their peers in the cities even after controlling for student socioeconomic status (SES). Similarly, Welch et al. (2007) found in New South Wales that students in rural and

remote communities were less likely to complete year 12 than their peers in larger cities, even after controlling for student SES. Welch et al. (2007) also found that school completion rates varied after controlling for concentrations of Indigenous students and school size. These studies suggest that school characteristics (other than school size) may vary by rural–urban location, and that these differences may help explain performance gaps between rural and urban students. This conclusion is also strongly supported by analyses of PISA data that demonstrate that school resources mediate the relationship between school and student socioeconomic status and academic performance (Chiu and Khoo 2005).

Method

This study examines data from PISA 2009. PISA is a large international student performance assessment of 15-year-olds. Since 2000 PISA has conducted assessments every 3 years. Each participating country's sample is drawn to be statistically representative of the total number of students enrolled in different types of schools (e.g., private or public), communities and geographical locations. The latest publicly available PISA assessment was conducted during 2009, with over 65 countries and nearly 470,000 students taking part (data from the last round of PISA, conducted in 2012, has yet to be released). The Australian PISA 2009 sample includes 353 schools and 14,251 students (Thomson et al. 2010). The PISA 2009 dataset includes responses to two main questionnaires: one completed by students and the other by school principals.

PISA is not a perfect tool for evaluating educational systems and student outcomes (Hopmann et al. 2008). Like all cross-sectional datasets, PISA does not allow researchers to show causal relationships among student or school characteristics and student performance. However, its advantage is that the number of participating countries and students is very large, and that it includes an extensive range of student and school variables. Another potential limitation of PISA data is many of the variables relating to school resources and learning environments are reported by the questionnaire respondents (i.e., either students or principals).

The Australian PISA 2009 dataset sourced from the Australian Council for Educational Research (ACER) groups participating schools into eight geographic categories based on the population size of the community; this variable is called 'School Community'. The eight categories range from communities with less than 1,000 inhabitants (the most 'rural' of the eight categories) to communities with more than 1,000,000 inhabitants (the most 'urban' category). ACER has redefined the five categories utilised within the original PISA data into eight geographic categories to better characterise the broad geographic variation of Australian communities. For Australia, the distribution of students and schools in these eight geographic categories is shown in Table 1.

In this investigation we calculated two additional contextual variables for each school community: (1) the average school SES; and (2) the ratio of Indigenous to non Indigenous students. We calculated these variables because, in the Australian case, they tend to be strongly associated with rurality and in Australian school

communities the Indigenous to non Indigenous ratio is a strong indicator of cultural dynamics. Each variable was calculated from individual student records in the PISA 2009 sample. Table 2 provides the ratios of Indigenous to non Indigenous students by community type and school SES. It should also be noted that Australia oversamples Indigenous students in PISA to gain a better understanding of the complexity of issues that affect this group of students.

Table 2 shows that the density of Indigenous students in rural school communities in Australia is greater than in school communities close to the centre of very large cities. There is the option for rural Australian students to transfer to city school communities or attend boarding school. However, as is highlighted by the mean SES variable and explained by the Human Rights and Equal Opportunity Commission (2000) "...for many Indigenous students each of these options violates cultural expectations and needs and is therefore unrealistic" (p. 14).

School community	Population	Number of students	Number of schools	
Small rural community	<1,000	182 (1 %)	6	
A small country town	1,000 to about 3,000	467 (3 %)	15	
A medium-sized country town	3,000 to about 15,000	1,811 (13 %)	45	
A larger town	15,000 to about 50,000	1,571 (11 %)	39	
A very large town	50,000 to about 100,000	1,236 (9 %)	29	
A city	100,000 to about 1 million	4,538 (32 %)	108	
Elsewhere in a very large city	>1 million	2,297 (16 %)	59	
Close to the centre of a very large city	>1 million	2,148 (15 %)	52	
Total		14,250 (100 %)	353	

Table 1 Distribution of students and schools by school community

Table 2	Indigeneity	of	school	communities
---------	-------------	----	--------	-------------

School community	Population	Ratio of indigenous to non indigenous students	Mean school SES	
Small rural community	<1,000	1:10.4	-0.02	
A small country town	1,000 to about 3,000	1:7.6	-0.01	
A medium-sized country town	3,000 to about 15,000	1:7.1	0.11	
A larger town	15,000 to about 50,000	1:6.6	0.15	
A very large town	50,000 to about 100,000	1:6.1	0.22	
A city	100,000 to about 1 million	1:12.6	0.47	
Elsewhere in a very large city	>1 million	1:25.7	0.35	
Close to the centre of a very large city	>1 million	1:31.5	0.52	
Average		1:11.5	0.34	

Our study utilises PISA questionnaire data provided by school principals about students, teachers and resources within individual schools. PISA collects such information because previous studies have reported that school resources are associated with student educational outcomes (Diseth 2007; Hoy et al. 2006; Schleicher 2009; Stewart 2008).

Principal responses to questions of teaching personnel shortages stem from the following questionnaire questions: Question 10, "The goal of the following set of three questions is to gather information about the student-computer ratio in your school" and Question 11, "Is your school's capacity to provide instruction hindered by any of the following issues?" Question 11 contains 13 'issues' that relate to the question stem: six issues concerning shortages of qualified teaching staff and seven issues about shortages of teaching materials and equipment. The response categories to the 13 issues comprise the following: 'not at all' (coded 1), 'very little' (coded 2), 'to some extent' (coded 3), and 'a lot' (coded 4). In keeping with the questionnaire format, we have kept principals' responses about shortages of teaching materials and personnel together in our analysis. We acknowledge, however, that these two domains are likely to have different impacts on student experiences and outcomes.

We calculated descriptive statistics (means, standard deviations, and frequencies) for principal responses to each item, across all eight school communities. Our purpose was to gather information from school principals about the degree to which shortages of teaching staff, materials and equipment vary across the eight rural–urban locations.

Findings

As reported in Table 2, the proportion of Indigenous to non Indigenous students is highest in school communities with 100,000 residents or less. Mean school SES is lowest in small rural communities and highest in school communities close to the centre of a very large city. Patterns in Table 2 indicate that school SES increases with the size of the community, with one exception. The average school SES is reported higher in smaller cities (<1,000,000 residents) than in the 'fringe suburbs' elsewhere in a very large city (more than 1,000,000 residents).

As noted above, we also calculated students' average literacy performance for the three subjects (mathematics, reading and science) assessed in PISA, for each of the eight school communities. These results are presented in Table 3.

Australia's PISA 2009 literacy performance outlined in Table 3 shows that students who attend school in a city centre achieve, on average, considerably higher mean scores than their peers in rural communities. This pattern supports research by Cresswell and Underwood (2004) who reported that Australian students who attended schools in close proximity to major cities and inner regional locations had stronger performance in the PISA 2000 Reading Assessment than students in regional and remote geographic locations. Indeed, student academic performance scores in PISA 2009 mathematics, reading and science appear positively related to community size, wherein increases in the size of the community are generally associated with higher literacy performance average scores. The apparent

School community	Population	Mathematics M (SD)	Reading M (SD)	Science M (SD)
Small rural community	<1,000	469.2 (80.3)	472.0 (91.4)	483.6 (84.8)
A small country town	1,000 to about 3,000	480.6 (87.8)	475.8 (97.0)	501.1 (97.3)
A medium-sized country town	3,000 to about 15,000	491.4 (82.2)	489.8 (93.1)	506.5 (93.2)
A larger town	15,000 to about 50,000	486.3 (84.9)	485.4 (93.8)	501.2 (95.3)
A very large town	50,000 to about 100,000	502.9 (87.7)	503.0 (94.6)	517.0 (96.6)
A city	100,000 to about 1 million	525.8 (88.8)	528.0 (93.6)	540.5 (96.8)
Elsewhere in a very large city	>1 million	514.8 (86.8)	516.2 (92.8)	524.6 (95.0)
Closer to the centre of a very large city	>1 million	541.3 (89.9)	541.7 (94.9)	550.8 (97.8)
Average		514.3 (89.4)	514.9 (96.0)	527.3 (97.6)

Table 3 Mean mathematics, reading and science literacy performance scores by geographic location

relationship between literacy performance and school community size is not completely linear, however. Average scores in all subjects were higher in mediumsized country towns than in larger towns. Another exception is that average scores are higher in smaller cities than "elsewhere in a very large city"; in other words, student literacy performance is higher in large regional cities than in the outer suburbs of the large capital cities. This pattern closely mirrors the pattern between mean school SES and community type reported in Table 2.

Figure 1 highlights the relationships among school community, school SES, indigeneity and student literacy performance in reading, mathematics and science, as assessed in PISA 2009.

Figure 1 illustrates that schools in small rural communities and small country towns enroll students with lower SES backgrounds, whereas schools close to the centre of very large cities tend to enroll students with higher SES backgrounds. Figure 1 reflects the strong association that exists between mean school SES and mathematics, reading and science literacy performance. There are a few exceptions to this pattern and for this reason the performance of students from larger towns (15,000–50,000 residents) is of interest. Likewise, the Indigenous to non Indigenous student ratio represented in Fig. 1 suggests that higher ratios are linked to weaker academic performance in mathematics, reading and science.

Table 4 summarizes principals' responses to questions about student-computer ratios across the eight school communities. Principals were asked to report on the number of computers that are available to 15 year-olds in their school and to identify the number of computers that have Internet access.

Table 4 suggests that, according to school principals, the mean number of computers available to 15-year-old students within year 10, across school communities, closely matches the mean number of students within this range. Similarly these data indicate that almost all school computers have Internet access.



Fig. 1 Mathematics, reading and science academic performance in PISA 2009 by mean school SES and geographic location

Thus, the data show that the student to computer ratio is very similar across the eight school communities, ranging from 0.9 to 1.1. The data provided by principals reports the ratio of computers to students, is largest in both the smallest rural communities and the most urban school communities. This is perhaps the result of the Labour government's education revolution, which included a policy of providing computers to every school (Buchanan 2011; Rudd et al. 2007).

A school's capacity to provide instruction can be hindered in many ways. Tables 5 and 6 report principals' responses to questions about their school's resources, which includes teaching staff, materials and equipment. Principals were asked about the degree to which their school's capacity to provide instruction is hindered by a lack of teaching personnel (Table 5) and teaching resources (Table 6).

THOIC + Rutto	The + Rado of statistic humbers to compares by school community								
School community	Total number of students in modal grade for 15-year-olds (Mean)	Computers for education in modal grade (Mean)	Ratio of computers to students	Computers in modal grade with internet access (Mean)					
Small rural community	34.7	38.7	1.1	38.7					
A small country town	54.7	57.8	1.1	57.8					
A medium- sized country town	119.3	108.4	0.9	107.9					
A larger town	160.8	140.7	0.9	140.7					
A very large town	196.9	172.5	0.9	171.9					
A city	173.4	171.9	1.0	171.3					
Elsewhere in a very large city	165.0	180.7	1.1	179.8					
Close to the centre of a very large city	188.3	202.4	1.1	201.6					
Average	160.4	159.3	1.0	158.3					

Table 4 Ratio of student numbers to computers by school community

According to the responses provided by school principals, shortages in teaching personnel vary moderately across the school communities, with principals in the smallest, most rural communities more likely to report that shortages hinder instruction in their schools compared to principals in more urban areas. The general trend shown in Table 5 is that shortages of teaching personnel become less pronounced, as the size of the school community increases, although there are a few exceptions as is evident in a larger town and a very large town data. The largest differences, according to location, in the degree to which teacher shortages were perceived by principals as hindering instruction were seen in mathematics. On average, school principals in small rural communities reported this a problem to some extent (mean = 2.7), whereas principals in urban schools reported maths teacher shortages a hindrance only to a very little extent (mean = 1.7). Somewhat surprisingly, we found that principals' reported that teaching personnel shortages were also a hindrance in towns ranging in size from 15,000 to 50,000 residents (a larger town). This suggests that shortages of teaching personnel are not limited to the most rural or remote communities. The number of principals who responded that their school is 'to some extent' affected by teacher shortages varies substantially across the school communities. For example, 83 % of principals in small rural communities report that a lack of mathematics teachers hinders instruction to some extent or a lot, compared to only 17 % of principals in communities close to the centre of a very large city. Further, one-half of principals in small rural communities report that a shortage of qualified teachers hinders instruction in their school to some

My school's capacity to provide instruction is hindered by a lack of						
School community	Science teachers	Mathematics teachers	English teachers	Qualified teachers	Library staff	Other personnel
Small rural community	<i>i</i>					
M	2.3	2.7	2.5	2.5	1.5	2.5
% 'to some extent'	50	83.3	33.3	50	16.7	33.3
% 'a lot'	0	0	16.7	16.7	0	16.7
A small country town						
М	2.1	2.5	1.9	2.1	2.2	2.3
% 'to some extent'	20	46.7	20	46.7	26.7	33.3
% 'a lot'	13.3	13.3	6.7	0	20	6.7
A medium-sized count	ry town					
Μ	1.8	1.9	1.6	2.2	1.4	1.9
% 'to some extent'	28.9	22.2	15.6	40	4.4	24.4
% 'a lot	0	8.9	0	2.2	2.2	2.2
A larger town						
М	2.2	2.3	1.9	2.2	1.2	1.9
% 'to some extent'	43.6	56.4	30.8	30.8	2.6	20.5
% 'a lot'	2.6	2.6	0	0	0	7.7
A very large town						
М	2.1	2.1	1.7	2.0	1.5	1.8
% 'to some extent'	34.5	34.5	24.1	27.6	13.8	20.7
% 'a lot'	6.9	6.9	0	3.4	3.4	6.9
A city						
М	1.8	1.9	1.6	1.9	1.4	1.8
% 'to some extent'	19.4	22.2	18.5	32.4	7.4	21.3
% 'a lot'	2.8	6.5	0	0	0	2.8
Elsewhere in a very la	rge city					
М	1.7	1.9	1.5	1.8	1.3	1.5
% 'to some extent'	27.1	27.1	5.1	20.3	5.1	8.5
% 'a lot	1.7	5.1	1.7	5.1	0	3.4
Close to the centre of	a very large	city				
М	1.5	1.7	1.5	1.7	1.3	1.8
% 'to some extent'	11.5	15.4	17.3	19.2	9.6	21.2
% 'a lot'	1.9	1.9	0	1.9	0	7.7
Mean	1.9	2.1	1.8	2.0	1.5	1.9

 Table 5
 Lack of teaching personnel by school community as reported by principals

Principal questionnaire response categories for question 11 coded as: 1: not at all, 2: very little, 3: to some extent and 4: a lot

Table 6	Shortages	of	teaching	resources
			the second se	

Calcal	0	To star s stá sa s 1		Testerment	Constant	T ilemente	A condition
community	Science laboratory equipment	material	Computers	Internet	computer software	materials	Audio– visual materials
Small rural communit	у						
М	1.8	2.2	2.3	1.7	1.7	1.7	2.3
% 'to some extent'	33.3	50	16.7	16.7	0	16.7	50
% 'a lot'	0	0	16.7	0	0	0	0
A small country town							
Μ	2.1	2.1	2.0	1.7	2.3	2.3	2.4
% 'to some extent'	33.3	40	20	6.7	40	33.3	40
% 'a lot'	6.7	0	6.7	0	6.7	6.7	6.7
A medium-sized coun	itry town						
Μ	2.0	1.7	2.2	2.0	2.0	1.9	2.0
% 'to some extent'	35.6	17.8	37.8	22.2	28.9	17.8	24.4
% 'a lot'	2.2	0	6.7	4.4	4.4	4.4	2.2
A larger town							
М	1.7	1.7	2.1	2.0	1.9	1.7	2.0
% 'to some extent'	10.3	15.4	25.6	10.3	15.4	12.8	25.6
% 'a lot'	2.6	2.6	5.1	12.8	2.6	0	0
A very large town							
Μ	1.9	1.5	2.1	2.0	2.0	1.8	1.8
% 'to some extent'	31	17.2	34.5	37.9	24.1	24.1	24.1
% 'a lot'	3.4	0	3.4	0	3.4	0	0
A city							
Μ	1.9	1.6	1.9	1.7	1.9	1.7	1.8
% 'to some extent'	20.4	14.8	20.4	14.8	17.6	7.4	13
% 'a lot'	4.6	0.9	3.7	3.7	3.7	2.8	2.8
Elsewhere in a very la	arge city						
Μ	1.8	1.6	1.9	1.7	1.7	1.6	1.8
% 'to some extent'	15.3	10.2	18.6	13.6	15.3	5.1	18.6
% 'a lot'	6.8	5.1	8.5	5.1	3.4	5.1	3.4
Close to the centre of	a very large	city					
М	1.61	1.4	1.7	1.6	1.5	1.3	1.4
% 'to some extent'	13.5	7.7	17.3	11.5	9.6	3.8	3.8
% 'a lot'	3.8	0	3.8	3.8	0	0	1.9
Mean	1.8	1.6	2.0	1.8	1.8	1.7	1.8

Principal questionnaire response categories for question 11 coded as: 1: not at all, 2: very little, 3: to some extent and 4: a lot

extent and another 17 % reported a lot. By comparison, 19 % of principals close the centre of a very large city report that a shortage of qualified teachers hinders instruction to some extent and 2 % reported a lot. Across all school communities,

principals reported a greater lack of mathematics, science and qualified teachers than shortages of English teachers or library staff.

When we placed school principals' responses into two categories, favourable (not at all and very little) and unfavourable (to some extent and a lot), the distribution of responses is noticeable. Sixty-six percent of principals of schools in small rural communities responded unfavourably regarding a shortage of qualified teachers as opposed to only 21 % of school principals close to the centre of a very large city.

Table 6 reports principals' responses about shortages of seven types of teaching resources. As portrayed in Table 6, on average, across all geographic regions of Australia school principals report that teaching resource shortages have very little negative effect on their schools' capacity to provide instruction. However, a more detailed examination of the frequencies of responses for each variable identifies that a small percentage of principals indicate that their school's capacity to provide instruction is indeed hindered, and that these trends are patterned by school location. For example, Table 6 reports considerable difference in the distribution of principals' response to, "My school's capacity to provide instruction is hindered by a lack or shortage of: computers". The largest proportion of principals who reported that a shortage of computers affects instruction to some extent or a lot, is found in small country towns (45 % of principals), while the smallest proportion is found in schools near a very large city centre (21 %). This may suggest that computers are especially useful for supporting learning in rural communities, where access to other materials such as instructional and audio-visual materials may be limited. Understanding the value each resource provides individual school communities has the potential to make a difference to how schools are resourced. Teese (2006) argues that "They [disadvantaged students] should be funded as vehicles of system renovation, aimed at delivering benefits to the school system as a whole" (p. 9).

Additionally, the range in principals' responses to shortages of instructional materials and audio-visual materials across school communities is substantial. Overall, school principals in the two smallest community groups report much higher shortages than their peers in larger communities. One-half of principals in small rural communities and 40 % of principals in small country towns report that a shortage of instructional materials hinders instruction in their school to some extent. This number drops substantially in larger communities, from 18 % in schools located in medium size country towns to <8 % in schools close to the centre of very large cities.

A comparison of Tables 5 and 6 indicates that school principals in small rural communities are more likely to respond that their school's capacity to provide instruction is hindered more by shortages of teaching personnel than by shortages of teaching resources, as shown by the higher mean values in Table 5. The findings presented in Tables 5 and 6 indicate that principal questionnaire responses to questions about teaching resources do vary by school community. For instance, school principals located in very large cities tend to suggest that their schools have sufficient resources, on average, in comparison to schools in small rural communities, for which principals on average tend to report resource shortages. The pattern is not completely linear, however, with principals of schools in very

large towns reporting larger shortages of teaching resources than principals of schools in larger towns, for example. Nevertheless, there is a very strong pattern in the data that shows that instruction is perceived, by school principals to be hindered substantially more in smaller communities than in the larger, most urban communities. While this finding is not surprising, our analysis is able to show in detail the extent to which the availability of resources is patterned according to school community.

Discussion

Our analysis of PISA 2009 data for Australia details principals' views and responses about their school's resources, according to eight types of geographic community. Our analysis found the following:

- i. Principals' responses indicate that shortages of resources are associated with school community; overall, principals of schools in small towns report that their schools have fewer resources compared to principals of schools in very large cities.
- ii. Many principals, especially those in less populated school communities, report that within their school instruction is hindered to some extent by a lack of resources, in particular shortages of teaching personnel.
- iii. Principals' responses suggest that the relationship between school resources and school community size is generally strong. However, some principals of schools in large towns report fewer resources than the principals of schools in smaller communities, and some principals of schools in non-central communities of very large cities (more than 1 million residents) report similar levels of resources as compared to schools in smaller communities.
- iv. The differences in principals' responses to shortages of teaching personnel are more pronounced across the school communities than are principals' responses to differences in shortages of teaching materials. This is particularly noticeable in the areas of mathematics, science and qualified teachers.
- v. Trends in the availability of resources across school communities are associated with trends in both school SES and average school literacy performance on PISA.

The trends displayed in Fig. 1 highlight that school SES has a strong positive association with students' academic performance in mathematics, reading and science. Our analysis also reports that principals of schools in rural communities believe their school experiences substantial shortages of resources. While this is perhaps not surprising, it should not be taken for granted as normal or natural. We argue that policymakers should expend more effort on understanding the values and needs of school communities and reduce the resource gap between rural and urban schools. We base this argument on the responses of school principals themselves, as reported in this study, as well as by research by Chiu and Khoo (2005), that suggests inequality in the distribution of resources lowers the performance of disadvantaged

students. Policy makers can certainly address some of the resource inequalities found in our analysis, especially those related to instructional materials.

Student access to computers and the Internet, as alluded to previously, has emerged as a significant issue in recent times. In Australia the issue of high speed broadband being rolled out to rural communities became one important focus of the 2007 and 2010 federal election campaigns. As noted by Fehring (2010), Labor "...policy initiatives were designed to achieve equity of access to information and communication technologies for all students, regardless of socioeconomic status or geographic location" (p. 181). Ilomäki and Kankaanranta (2009) have noted, "The same trend regarding heavy ICT investment in education has become evident in many developing countries..." (p. 101).

As reported in Table 6, principals' responses about the distribution of computer and information communication technology varies, whereas, there is very little difference between principals' responses to questions on school community resources. Reassuringly, principals of regional school communities reflect the most appealing ratio of student numbers to computers. This suggests that recent government education policy has made a difference to digital technology resources in Australia's rural school community sector. However, PISA does not collect information about many other important aspects, such as the speed of Internet access, availability of technical support, the impact of such resources on learning or the quality of resources used within each school community. The extent to which computers are used and valued as an instructional teaching tool is also unclear. Minguez and Ballesteros (2008) state, "According to the PISA Report 2005 report ...[it is unknown if] school-based access [to computers] has an effect strong enough to compensate for the effect of lacking a computer at home" (p. 433).

One limitation of this study is the small number of participating principals of schools in some school communities. Caution should therefore be exercised when generalizing; at the same time, however, the strength of using the PISA dataset is that it is a nationally representative sample. Our analysis is also limited by the unavailability of a variable about teacher experience. It is well known that rural schools often have large numbers of recent teacher graduates and less experienced school principals. Further, although these data reflect the views of school principals rather than an objective measure of these aspects of school resourcing, asking principals about the degree to which instruction in their school is hindered by a shortage of experienced teachers would be highly relevant for the Australian context.

Conclusion

Our analysis of PISA 2009 examines in fine-grained detail principals' responses to questions concerning school resources. The findings of our analysis suggest an unequal distribution of resources (teaching materials and personnel) between rural and urban schools. The analysis provided in this paper is unable to explain why student performance is higher in larger communities, nor does it establish how school resources could mediate the relationship between geographic status and education outcomes. However, this study has unearthed patterns as reported by school principals about the distribution of school resources across Australia's eight school communities. Whilst the trends examined in this paper cannot be used to assess the degree to which school resources relate to learning outcomes, previous studies from a range of international contexts support the claim that they are important for learning outcomes. Moreover, shortages of teaching materials and personnel also affect the learning experiences of students. We agree with Gordon

personnel also affect the learning experiences of students. We agree with Gordon and Monastiriotis (2006)that more research should be centred on, socioeconomic status and school constructs such as school resources and learning environments on educational opportunities, experiences and outcomes. The findings of this study can be useful for a wide audience, including education

researchers, practitioners and policymakers. One policy recommendation that could stem from the feedback of principals would be to increase the availability of instructional materials for schools in rural and remote communities. Addressing teaching shortages in rural communities is difficult, but providing sufficient instructional materials should be a routine matter for a wealthy country such as Australia. The findings of our study could also be useful for graduate teachers from capital cities who are preparing for work in rural communities as it will heighten their awareness of the contrasts that exist between school communities across Australia. Finally, it may also help researchers and policy makers understand how schools in particular settings can be better supported.

Our findings show that the distribution of resources across school communities as reported by principals closely mirrors school academic performance and school socioeconomic status. Our findings suggest that rural schools are more affected by shortages of teaching materials and personnel than are schools in larger towns and cities. Rural schools have lower performance scores and higher levels of social disadvantage. To reduce the performance gap between rural and urban schools, we would argue that schools in rural communities should have the opportunity to have resources distributed according to community needs. This could equate to the same or even more resources than their urban counterparts, not less.

Acknowledgments This study was supported by a Grant from the Australian Research Council's Discovery Projects funding scheme (Project number 1097057) awarded to Laura Perry.

Significance of study

The analyses of school resources provided in the first research paper helps to contextualize the academic achievement findings in PISA 2009 within an Australian geographical context: for example, how does the average achievement of rural schools in Australia compare with those near a large city? The research and review of literature that was completed in the first paper of my thesis highlights the complexity of the interrelatedness of Australia's three groups of students performing at significantly lower academic levels than their peers. The paper presents data that provide analysis of reading, mathematics and science mean literacy performance in PISA 2009 by mean school social economic status and school community (location).

Study 1 is focused upon the Australian school context. Prior to undertaking research for the first paper I had envisaged a dramatic difference in principals' responses to questions relating to school resources across Australia's school communities. However, that difference was not as distinct as I had imagined. The data did suggest, however, that school resources within Australia varied according to school community. For instance, principals of schools in small towns, compared to principals of schools in very large cities, reported that their schools have fewer resources. Many principals, especially those in less populated school communities, also reported that instruction that occurs within their school is hindered to some extent by 'a lack of resources', in particular, 'shortages of teaching personnel'.

The findings of my research suggest that Australia's school communities need to be better recognized and understood. For instance, some principals of schools in large towns report fewer resources than the principals of schools in smaller communities and some principals of schools in non-central communities of very large cities (more than 1 million residents) reported similar levels of resources as schools in smaller communities. This could suggest that strategies need to be put in place to ensure school community allocated funding is closely monitored as factors such as geographic location and population may have greater implications in some communities than others.

Overall, the trends that I analysed in the first research paper highlight that school SES appears to have a positive association with students' academic performance in reading, mathematics and science and that principals of schools in rural communities believed their schools suffer from shortages of resources. I hope this paper can assist education policy-makers and researchers to gain a clearer understanding of the needs of school communities. Ultimately, I hope that the research I have conducted enables Australian rural school communities to become better resourced.

Chapter 5:

Paper 2: How do school learning environments differ across Australia's rural, regional and metropolitan communities?

Background

My second paper, "How do school learning environments differ across Australia's rural, regional and metropolitan communities?" was published in the *Australian Educational Researcher* (AER) in July of 2014 and received the accolade of being among five articles nominated by the Australian Association for Research in Education (AARE) as the best papers published in *Australian Educational Researcher* in 2014. This article was also awarded the "Top Article by a Post-Graduate Student in 2014" by the School of Education at Murdoch University.

Utilising the PISA classification of learning environment, which includes teacher expectations, teacher morale and relationships between students and teachers, I set out to discover to what extent these variables are associated with educational outcomes. My literature review, consistent with earlier findings, suggested that the strength of the relationship between learning environments and student achievement is inconclusive.

Unlike previous PISA questionnaire datasets, PISA 2009 did not ask students about their sense of belonging to school and their perceptions of their school's climate. However, students were asked about their attitudes towards school and their perceptions of their learning environments, classroom disciplinary climate and relations with teachers. One area of interest that my second research paper investigated is the construct of 'classroom discipline', a key component of the learning environment. Data from PISA 2009 showed that schools that have a better school disciplinary climate (e.g., fewer interruptions and distractions in the classroom) have higher academic achievement (OECD, 2010b). However, neither the OECD report nor the Australian national report for PISA 2009 examined how school disciplinary climate varies across schools in different locations or with different cohorts of students. It was this stimulus that lead my investigation into PISA 2009 for the second article.
How do school learning environments differ across Australia's rural, regional and metropolitan communities?

Kevin Sullivan · Laura B. Perry · Andrew McConney

Received: 22 November 2013/Accepted: 14 February 2014/Published online: 1 March 2014 © The Australian Association for Research in Education, Inc. 2014

Abstract This study uses data from the Programme for International Student Assessment, a large and nationally representative dataset, to examine how learning environments vary across metropolitan, rural and regional schools in Australia. Research has shown that school climate and learning environments are related to student academic performance, but little is known about the degree to which they differ across school communities in Australia. We examined principals' perceptions of teacher and student behaviour related to school climate and students' perceptions of teacher support, classroom disciplinary climate, and the relevance of education. The findings show that regardless of where they live, most students believe that schooling is worthwhile, and report positive relationships with their teachers. Perceptions of classroom disciplinary climate vary more across school communities, however, with students reporting less positive disciplinary climate in rural communities than in very large cities. Principals' perceptions of teacher and student behavior related to school climate varied; with urban schools having much more positive results than schools in towns and rural communities. Finally, our findings show that students' and principals' perceptions of their school climate and learning environments are more positive in urban communities than in rural communities, but that the least positive environments are generally found in country towns rather than remote communities. Our findings suggest that attention should be paid to improving learning environments not just in the most rural/remote communities, but also in largish regional towns of up to 50,000 residents.

Curtin University, Bentley, WA, Australia e-mail: kevin.sullivan@curtin.edu.au

A. McConney e-mail: a.mcconney@murdoch.edu.au

K. Sullivan (🖂)

K. Sullivan · L. B. Perry · A. McConney Murdoch University, Murdoch, WA, Australia e-mail: l.perry@murdoch.edu.au

Keywords School learning environments · Communities · Geographic location · Rural education · PISA · Australia

Introduction

The term 'school learning environment' is a broad construct that includes a range of student and teacher relationships, behaviours, attitudes and expectations. It can include, among others, relationships between teachers and students, school climate, and classroom disciplinary climate. In the context of this paper school learning environment is not exclusive to but includes classroom and school disciplinary climate, student and staff communication, expectations and absenteeism. Research has shown that positive student perceptions of learning environments can promote positive student outcomes such as achievement and attainment (Allen and Fraser 2007; Patrick et al. 2011). At the same time, however, a positive learning environment is an outcome in itself, not just a means to another end. Whether related to academic outcomes or not, all students deserve to be treated equitably by their teachers and to learn in orderly, well-resourced classrooms that are conducive to learning.

While it is well known that students who attend rural schools in Australia have lower educational outcomes than their peers in urban communities, much less is known about how rural and metropolitan schools differ from each other. Research has shown that compared to students in urban areas, rural students have lower high school retention rates (Godden 2007; Marks and Fleming 1999), are less likely to complete Year 12 (Alston and Kent 2006; Marks et al. 2000), are less likely to attend university (James 2001), and have lower results on standardized tests of academic performance (Cresswell and Underwood 2004). Research has also shown that compared to urban schools, rural schools have difficulty recruiting and retaining teachers (Ministerial Council on Education et al. 2003; Vinson et al. 2002; Yarrow et al. 1999), have fewer instructional resources (Cresswell and Underwood 2004; Sullivan et al. 2013), and offer fewer academic and vocational courses (Human Rights and Equal Opportunity Commission 2000). Very little is known, however, about how learning environments differ between rural and urban schools in Australia.

In this study we examine how school learning environments, as perceived by students and principals, vary across rural and metropolitan communities in Australia. We use a large, nationally representative dataset—the *Programme for International Student Assessment*—to conduct secondary analyses of school learning environments. Our approach is primarily descriptive and exploratory, which is appropriate since very little is known about the degree to which learning environments vary across rural, regional and urban contexts. The findings of our study may serve as a foundation for further research about how and why differences in perceptions of school learning environments exist, based on school location. Understanding the degree to which perceptions vary by geographic location and community size can provide insight about ways to improve student educational

outcomes and experiences. It can also shed light on ways to improve the working conditions of teachers and school principals.

Our rationale for this study which sets out to examine principals' and students' perceptions about school learning environment across a range of differently sized and located Australian school communities is threefold. First, from PISA data we have learned that students in rural communities have, on average, lower educational outcomes than students in metropolitan communities (Cresswell and Underwood 2004; De Bortoli and Thomson 2010). While many rural students also have socioeconomically disadvantaged family backgrounds, it is plausible that some of the variation in student outcomes is due to differences in school learning environments (OECD 2004). Second, understanding how perceptions of learning environments vary across rural and metropolitan schools can be useful for stakeholders who are seeking to develop and promote rural schools and rural communities more generally. Third, we are committed to ensuring that all students are able to enjoy positive learning environments, regardless of where they live. If substantial differences are found across metropolitan and rural schools, our findings can highlight to educational policy makers what aspects of school learning environments need to be improved to ensure that a more equitable education learning environment exists across all Australian school communities.

Background

Perceptions of learning environments can enrich our understanding of learning outcome variance (Marton 1981; Fraser and Fisher 1982; Lizzo et al. 2002).

This research paper uses questionnaire data, sourced from PISA 2009, to build on research that shows school learning environments can positively contribute to students' educational outcomes. School learning environments are shaped by teachers, including teacher expectations, teacher morale, and relationships between students and teachers. Research indicates that student learning is enhanced by a positive school learning environment that is cohesive and promotes high academic expectations (Diseth 2007; Hoy et al. 2006; Stewart 2008), orderly classrooms (Ma and Willms 2004; Shin et al. 2009; Willms 2010) and adequate educational resources (Orr 2003; Portes 2005). Research has also shown that in terms of teaching, student learning is promoted by supportive and caring teachers who have high expectations (Auwarter and Aruguete 2008; Hardré and Reeves 2003; National Research Council 2003; Rosenfeld et al. 2000), qualified and experienced teachers (Crofford et al. 2011; Nye et al. 2004) and inspiring teachers (Hardré and Reeves 2003).

While ethnographic studies of Australian schools (see, for example, Angus 1993; Connell et al. 1982; Smyth et al. 2003) have suggested strong interplay between school learning environment, learning resources and policies, very little is known about the degree to which the learning environments in rural schools differ from other schools in Australia. To our knowledge only a few studies have examined how school learning environments vary across geographic contexts and community size. The most comprehensive study of this topic is by Cresswell and Underwood (2004).

Using data from the 2000 cycle of PISA, Cresswell and Underwood (2004) compared principals' responses about a range of school characteristics across rural and metropolitan communities in Australia. They found that neither teacher qualifications nor teacher morale differed substantially across geographic context. They did find, however, that principals' perceptions of teacher factors related to school learning environment were less positive in remote communities than in large cities. Waldrip and Fisher (2007) found very small differences in students' perceptions of teacher-student interpersonal behaviour in remote, regional and urban schools in Western Australia and Queensland. Young (1998) found small but statistically significant differences in student perceptions of the support that they received from their teachers in Western Australia. Students in regional, rural and remote communities reported higher levels of support from their teachers than did students in Perth. None of these three studies examined students' perceptions of classroom disciplinary climate, and none of them reported their findings about school differences in much detail. Understanding how classroom disciplinary climates vary is important since studies have shown it is strongly related with positive student outcomes (see, for example, Frempong et al. 2012; Ma and Willms 2004; Schleicher 2009; Shin et al. 2009).

Studies from the USA (Reeves 2011) and Australia (Cresswell and Underwood 2004; Young 1998) have shown that inequalities in educational outcomes between metropolitan and rural schools are largely due to student characteristics such as socioeconomic status, ethnicity and poverty rather than school characteristics. It is plausible, however, that real differences between schools, based on geography, may be masked because learning environments tend to be related with social background. For example, it is well known that students from privileged social-economic backgrounds typically enjoy more positive learning environments than their less advantaged peers (OECD 2005; Thomson 2002), and in general, students in rural environments in Australia are less economically advantaged than their urban peers (Lamb 1994).

The PISA 2009 dataset includes a large number of variables that relate to learning environment. The OECD's primary report for PISA 2009 shows that schools with a better school disciplinary climate (e.g., fewer interruptions and distractions in the classroom) have higher academic performance (OECD 2010). Neither the OECD's primary report nor the Australian national report for PISA 2009 examines how disciplinary climate or other aspects of school learning environment vary across schools in different locations. Our study builds on the literature by showing in more detail than has been done so far the degree to which learning environments differ across rural and metropolitan schools in Australia.

Method

We use data from the 2009 cycle of the *Programme for International Student Assessment* (PISA) to compare learning environments in rural and metropolitan schools in Australia. This was the latest cycle when we conducted the study; the most recent cycle, PISA 2012, was released after we completed the study. PISA

measures the academic literacy of 15 year-old students in maths, science and reading and is organized by the Organisation for Economic Development and Cooperation (OECD). The OECD has administered PISA every 3 years since 2000. The dataset for PISA 2009 includes nearly 470,000 students from 65 countries. The Australian PISA 2009 sample is nationally representative of the total number of students enrolled across all school types (e.g., private or public), communities and geographic locations. It includes 14,251 students from 353 schools (Thomson et al. 2010).

PISA collects data from students and school principals. In addition to measuring students' academic performance, PISA also collects data about a wide range of possible predictor variables, including time spent on learning in the classroom and at home, the use of technology, reading habits and the skills, values, attitudes, experience, practice, and student access to resources at home and school. Of particular interest to this study is the learning environment of the school.

We sourced the PISA 2009 dataset from the Australian Council for Educational Research (ACER). ACER redefines the five categories that are reflective of population size within the original Australian PISA 2009 dataset into eight geographic categories. ACER classifies participating schools into eight geographic categories to better characterise the broad geographic variation of Australian communities; this variable is called 'School Community'. The eight categories range from communities with less than 1,000 inhabitants (the most 'rural' of the eight categories) to communities with more than 1,000,000 inhabitants (the most 'urban' category). Information about the number of schools and students in the sample is presented in Table 1.

Table 1 shows the average student socioeconomic status for each school community group. We calculated this contextual information since research has shown that learning environments are related to socioeconomic status and ethnicity (Camburn and Han 2011; Cresswell and Underwood 2004; De Bortoli and Thomson 2010; Portes 2005). Student socioeconomic status in PISA is represented by an index comprised of multiple items related to parent education, parent occupation, and economic and cultural resources in the home. In PISA it is named the Index of Economic, Social and Cultural Status (ESCS); throughout the paper we refer to this variable as student socioeconomic status (SES). PISA calculates an ESCS value for each student; we calculated the mean SES for all students in each school community, using weighted data. In the Australian dataset, the index ranges from a minimum of -3.40 to a maximum of 2.98, with a mean of .34 and a standard deviation (SD) of .754.

As can be seen in Table 1, the social context of schools varies across the eight school communities. The average socioeconomic status of students is positively related to school community size, with the smallest school community having the lowest average student SES and inner city communities in very large capital cities having the highest average SES. The relationship is largely linear except for outlying neighbourhoods of very large cities, which have a lower average student SES than regional cities. This relationship is shown more clearly in Table 1.

PISA collects data about learning environments because research indicates that they are associated with student educational outcomes (Diseth 2007; Hoy et al.

School community	Population	No. Schools	No. Students	Mean, index of student socioeconomic status
Small rural community	<1,000	6	182	021
A small country town	1,000 to about 3,000	15	467	.011
A medium-sized country town	3,000 to about 15,000	45	1,811	.114
A larger town	15,000 to about 50,000	39	1,571	.153
A very large town	50,000 to about 100,000	29	1,236	.234
A city	100,000 to about 1 million	108	4,538	.469
Elsewhere in a very large city	>1 million	59	2,297	.358
Close to the centre of a very large city	>1 million	52	2,148	.533
Mean or total		353	14,250	.344

Table 1 Classification of school communities & distribution of students

2006; Schleicher 2009; Stewart 2008). There are a range of learning environment variables included in the student and principal questionnaires. Students are asked questions about their teachers (e.g., "My teachers treat me fairly" and "My teachers are interested in my well-being") and classroom discipline and environment (e.g., "There is a lot of noise and disorder in my classroom"). Principals are asked questions about teachers at their school (e.g., "Teachers at my school have low expectations of their students"). Each item comprises a string of four Likert-scale response categories.

We examined the principal and student PISA questionnaire items about school learning environments. Principal data come from Question 17, which asks principals about the degree to which learning in their school is hindered by 13 sub-items related to teacher and student behaviour. A full list of all 13 sub-items is included in Table 2. Student data are drawn from Questions 33, 34 and 36. Question 33 comprises four sub-items about students' attitudes about the relevance of schooling, Question 34 comprises five sub-items about student-teacher relationships, and Question 36 five sub-items about classroom disciplinary climate. Each provides four Likert-scale responses (e.g., strongly agree, agree, disagree and strongly disagree). A full list of all the sub-items is included in Tables 3, 4 and 5.

For each of these principal and student questions, PISA has calculated a numerical index. The indexes from the student questionnaire are named student attitudes toward schooling, student-teacher relations, and classroom disciplinary climate. Principals' responses from Question 17 are divided into two indexes, one about teacher-related behaviors and the second about student-related behaviors. Using these indexes, we have compared means across the school communities. We also compared the average SES of students in each school community since SES is related to learning environments and geographic location (Camburn and Han 2011; Cresswell and Underwood 2004).

	hereputure u	T ICACITCI ATTA S		TAVIUI ICIAIC		. ובמוווחוק כווע							
To what extent is princ	ipals' perceptic	ons of the learni	ing of stude	nts hindered	by the follov	ving							
School community	Teacher's low expectations	Student absenteeism	Student– teacher relations	Student disruption	Student's needs not met	Teacher absenteeism	Skipping classes	Students lacking respect	Staff resisting change	Student drug use	Teachers too strict	Students being bullied	Students not encouraged
Small rural community													¢.
% 'to some extent'	33	33	17	17	67	33	17	33	33	0	17	17	17
% 'a lot'	0	0	0	0	0	0	0	0	0	0	0	0	17
A small country town													
% 'to some extent'	40	40	40	60	60	20	47	47	47	L	13	47	27
% 'a lot'	0	40	0	0	0	0	0	0	13	0	0	0	0
A medium-sized country town													
% 'to some extent'	38	51	27	42	47	9	22	42	33	11	L	24	24
% 'a lot'	0	16	5	L	7	0	4	2	4	0	0	2	2
A larger town													
% 'to some extent'	41	62	26	56	59	31	41	38	38	10	£	33	26
% 'a lot'	3	23	0	10	0	0	e.	£	N N	0	0	0	ę
A very large town													
% 'to some extent'	34	41	21	24	28	10	24	14	41	L	L	21	21
% 'a lot'	0	10	Э	7	10	0	7	L	0	3	0	Э	Э
A city													
% 'to some extent'	28	34	7	23	35	14	20	18	32	Ч	4	13	18
% 'a lot'	3	10	5	З	2	0	9	1	4	T	0	1	1
Elsewhere in a very large city													
% 'to some extent'	34	25	15	25	46	15	17	22	32	5	0	10	19
% 'a lot'	2	14	2	7	0	0	8	0	2	0	0	З	N.

14 Table 2 Principals' nercentions of feacher and student behavior related to school learning

街 Springer

How do school learning environments differ across Australia?

 $\underline{ \mathfrak{G}}$ Springer

528

To what extent is prin-	cipals' perceptic	ons of the learni	ng of studer	nts hindered	by the follow	ving							
School community	Teacher's Iow expectations	Student absenteeism	Student- teacher relations	Student disruption	Student's needs not met	Teacher absenteeism	Skipping classes	Students lacking respect	Staff resisting change	Student drug use	Teachers too strict	Students being bullied	Students not encouraged
Close to the centre of a very large city													ľ
% 'to some extent'	19	31	4	10	21	10	8	8	36	0	4	14	17
% 'a lot'	0	8	0	0	2	0	6	0	0	2	0	0	0
Principal questionnaire number	: response catego	ory for Question	. 17 coded a	s: (1) not at a	dl, (2) very l	ittle, (3) to son	ne extent, an	d (4) a lot.	Percentages	s have beer	ı rounded to	the nearest	even whole

Table 3	Indexes	of principals'	perception	of	teacher	and	student	behavior	related	to	school	learning
environm	nent											

School community	Index of terms $N = 353$, S	acher behaviour SD = .923	Index of sti $N = 353$, S	udent behaviour D = .986
	Mean	SE	Mean	SE
Small rural community	7488	.07998	.0186	.23016
A small country town	4372	.05691	5982	.18668
A medium-sized country town	2967	.03855	4110	.11770
A larger town	7412	.04398	6695	.10642
A very large town	1329	.07814	2398	.15826
A city	.0245	.04741	.1218	.09689
Elsewhere in a very large city	3139	.03919	.0588	.13087
Close to the centre of a very large city	.0384	.05161	.4205	.13167
Mean	2268	.01931	0621	.05140
Gap between lowest and highest mean score, expressed in SD units	.853		1.149	

We use descriptive statistics (means, standard deviations, and frequencies) to compare learning environment response variables across the eight different geographic locations. Our aim is primarily descriptive and exploratory, with the purpose of mapping in detail how learning environments vary by rurality and across different school communities.

Findings

Before examining how learning environments differ across the eight school communities in Australia, we calculated average academic performance in mathematics, science and reading score on PISA for each community. The academic performance of students who attend school in a city centre, as illustrated in Fig. 1, is considerably higher than their peers in rural communities.

As shown in Fig. 1, academic performance is associated with community size/ location, with increases in the size of the community/proximity to a very large city associated with higher student performance. However, the apparent relationship between reading performance and school community size/location is not completely linear. Average performance, across all subjects, is higher in medium-size country towns than in larger towns. Another exception is a higher average performance in smaller cities than elsewhere in a very large city; in other words, student performance is higher in large regional cities than in the outer suburbs of the large capital cities.

Next, we calculated descriptive statistics for the responses to 13 sub-items from Question 17 of the principals' questionnaire, "To what extent is learning hindered in your school by...". These results are presented in Table 2. The 13 sub-items relate to student and teacher behaviours, such as teacher and student absenteeism, teacher

School community	Population	School has done little to prepare me for adult life	School has been a waste of time	School has helped give me confidence to make decisions	School has taught me things that could be useful in a job
Small rural community					
% 'agree'	<1,000	19	8	62	55
% 'strongly agree'		4	0	25	40
A small country town					
% 'agree'	1,000 to	18	7	61	52
% 'strongly agree'	about 3,000	5	5	19	39
A medium-sized country town					
% 'agree'	3,000 to	18	8	63	55
% 'strongly agree'	about 1,5000	4	3	17	35
A larger town					
% 'agree'	15,000 to	16	7	64	53
% 'strongly agree'	about 50,000	3	2	20	40
A very large town					
% 'agree'	50,000 to	15	7	65	52
% 'strongly agree'	about 100,000	4	2	18	40
A city					
% 'agree'	100,000 to	13	6	63	49
% 'strongly agree'	about 1 million	4	1	22	43
Elsewhere in a very large city					
% 'agree'	>1 million	16	4	62	50
% 'strongly agree'		4	2	22	42
Close to the centre of a very large city					
% 'agree'	>1 million	12	5	62	51
% 'strongly agree'		4	1	22	41

Table 4Student attitudes towards school

Student questionnaire response category for Question 33 coded as: (1) strongly disagree, (2) disagree, (3) agree, and (4) strongly agree

expectations, student drug use, and bullying. As the table shows, response differences exist between Australia's rural, regional and metropolitan school communities for some of these sub-items. A large number of principals responded in a less than positive manner to questions concerning student learning environments. For clarity's sake, we present the frequencies of these negative responses in Table 2 and subsequent Tables. It is these negative responses that are most instructive for

How much do you dis	agree or agree	with each of the f	ollowing statement	s about teachers at	your school
School community	I get along well with most of my teachers	Most of my teachers are interested in my well-being	Most of my teachers really listen to what I have to say	If I need extra help, I will receive it from my teachers	Most of my teachers treat me fairly
Small rural community					
% 'agree'	70	68	70	72	71
% 'strongly agree'	16	11	7	14	13
A small country town					
% 'agree'	58	64	54	61	62
% 'strongly agree'	21	9	10	17	14
A medium-sized country town					
% 'agree'	66	62	56	64	68
% 'strongly agree'	16	10	9	14	14
A larger town					
% 'agree'	63	63	57	68	69
% 'strongly agree'	19	10	9	14	13
A very large town					
% 'agree'	67	67	61	66	69
% 'strongly agree'	19	11	10	17	16
A city					
% 'agree'	66	66	62	67	69
% 'strongly agree'	20	14	12	18	16.5
Elsewhere in a very large city					
% 'agree'	65	64	61	66	68
% 'strongly agree'	20	14	12	20	17
Close to the centre of a very large city					
% 'agree'	66	68	64	68	70
% 'strongly agree'	26	14	12	20	17

 Table 5
 Student perceptions of their relations with teachers

Student questionnaire response category for Question 34 coded as: (1) strongly disagree, (2) disagree, (3) agree, and (4) strongly agree

understanding where and to what degree learning environments are hindering student learning. Moreover, these less favourable responses tend to be related with community size. Based on the PISA data, principals in 'larger towns' (15,000–50,000 residents), 'medium-sized country towns' (3,000–15,000 residents), and 'small country towns' (1,000–3,000 residents) are substantially more likely to report less favourable responses for most items; principals in schools close to the centre of very large cities are the least likely to report less favourable responses.



Fig. 1 Distribution of academic performance in PISA 2009 by school community

Student drug use, staff resisting change, and teachers too strict are items within the questionnaire data that reflect relatively small differences in response across school communities. The remaining items vary substantially across school communities, however. For example, up to 60 % of principals in communities with fewer than 50,000 residents report that student disruptions hinder learning in their school either to some extent or a lot, compared to only 12 % of principals from schools located close to the centres of very large cities. The teacher-related behaviors that varied the most are teachers having low expectations of their students, and teachers not adequately addressing the needs of their students.

We also compared means across the eight types of school community for the two indexes representing principals' perceptions of their school's learning environment.

Deringer



Fig. 2 School principals' perceptions of teacher and student behavior by community type, in comparison to students' socioeconomic status

The results are shown in Table 3 and Fig. 2. As noted earlier, these two indexes are created by PISA and are based on the 13 items about school learning environment shown in Table 2.

As shown in Table 3, principals' perceptions of the teacher and student related behaviour that hinders learning varies substantially across the eight school communities. On both indexes, the lowest values are found not in the smallest, most rural communities but rather in larger towns of 15,000–50,000 residents. Principals' from schools located in central areas of very large cities (more than 1 million people) have the most positive perceptions. Principals' perceptions of both teaching and student behaviour are more negative in outlying areas of very large cities than in cities with fewer than 1 million residents. The difference between the highest and lowest values is very large on both indexes: .853 and 1.149 standard deviations for the teacher and student behavior index respectively. Slavin and Fashola (1998) define differences of .25 standard deviations as educationally significant.

We then plotted the means from Table 3, along with the means for student socioeconomic status, in Fig. 3. Although a line graph has been utilized to illustrate the trends it is important to note that the values that have been recorded are categorical and not continuous data. The figure illustrates graphically the large differences that are reported in Table 3. By plotting student socioeconomic status, we can also see that differences in principals' perceptions across the school communities do not correspond perfectly to the average socioeconomic status of students. Another interesting finding is the large difference between the two scales in small rural communities. Principals' perceptions in this school community indicate that student learning is hindered much more by teacher behavior than by student behavior.



Fig. 3 Students' perceptions of school learning environments and socioeconomic status

Next we examined students' responses about their school learning environment. The data show students' perceptions across the eight school communities about the relevance of education (Table 4), student and teacher relationships (Table 5) and classroom disciplinary climate (Table 6).

Table 4 presents students' attitudes toward school. The table comprises four items about the relevance and value of schooling. The analysis shows no large differences in student attitudes across the eight school communities. Overall, students in all eight school communities report favorable attitudes towards school. This is a very positive finding that contradicts commonly held assumptions that students in rural and remote communities often perceive school to be irrelevant.

Table 5 reports student perceptions of their relationships with their teachers. As shown in the table, most students across all eight school communities report positive relationships with their teachers. Differences in perceived student-teacher relationships between school communities, as shown in Table 5, are very small. The findings suggest that most students across all school communities perceive that their teachers treat them with respect and provide them with adequate support.

Table 6 reports students' perceptions of their classroom disciplinary climate. As can be seen, students' views of classroom disciplinary climate are not ideal in most school communities. For example, 25 % of students in schools close to the centre of a very large city report that students do not listen to their teachers in most or every lesson, as do approximately 40 % students in communities that have between 3,000 and 100,000 residents. When combining the frequencies for 'most lessons' and 'every lesson', students in schools close to the centre of a very large city were the least likely to report negative responses, and students in larger towns

School community	Students don't listen to what the teachers says	There is noise and disorder	The teacher has to wait a long time for the class to quiet down	Students cannot work well	Students don't start working for a long time after the lesson begins
Small rural community					
% 'most lessons'	26	37	30	19	20
% 'every lesson'	6	9	3	1	2
A small country town					
% 'most lessons'	26	32	25	17	20
% 'every lesson'	9	14	8	5	7
A medium-sized country town					
% 'most lessons'	29	34	25	16	19
% 'every lesson'	9	12	9	6	9
A larger town					
% 'most lessons'	28	34	26	18	21
% 'every lesson'	12	16	13	7	11
A very large town					
% 'most lessons'	28	32	24	14	20
% 'every lesson'	11	12	9	5	7
A city					
% 'most lessons'	22	25	19	12	15
% 'every lesson'	7	11	7	5	7
Elsewhere in a very large city					
% 'most lessons'	23	27	19	13	17
% 'every lesson'	9	12	9	5	7
Close to the centre of a very large city					
% 'most lessons'	19	23	16	10	13
% 'every lesson'	6	9	7	3	5

Table 6 Student perceptions of classroom disciplinary climate

Student questionnaire response category for Question 36 coded as: (1) never/hardly ever, (2) some lessons, (3) most lessons, and (4) every lesson

(15,000–50,000 residents) were the most likely to report negative responses, on all five items.

Finally, we compared means for these three groups of student questionnaire items across the school communities. We used the indexes of student attitudes toward schooling, teacher-student relations, and classroom disciplinary climate. As noted earlier, these indexes comprise all of the sub-questions reported in Tables 4, 5 and 6. Table 7 shows that the means on each index vary across the school communities, with differences in classroom disciplinary climate being the strongest. The table also shows that students in large and very large cities report the most positive responses on all three indexes. The most negative responses are in medium and large sized towns, not the smallest communities. While Table 7 and Fig. 3 show that the means on these three indexes vary across the school communities, the difference between the lowest and highest means as expressed in standard deviation units is not educationally significant as defined by Slavin and Fashola (1998).

We plotted the means of these three indexes, along with the average student SES for each school community, to show visually how these indexes vary by school community and average student SES. A line graph illustrates the trends, however, it is important to note that the values recorded are categorical and not continuous data. As seen in Fig. 3, student perceptions of their school learning environments vary by school community, with the least positive experiences occurring in towns ranging from 3,000 to 50,000 people. On all three indexes, students that attend schools in larger towns (15,000–50,000 residents) report more negative responses than their peers in smaller communities, even though their average SES is higher.

School community	Index of towards $N = 13,4$ SD = 1.4	attitudes school 455 016	Index of student r N = 13.9 SD = .9	teacher- elations 966 86	Index of disciplina N = 13.9 SD = 1.0	classroom uy climate 061 010
	Mean	SE	Mean	SE	Mean	SE
Small rural community	.133	.01990	.092	.01761	194	.01504
A small country town	.056	.01389	047	.01241	230	.01164
A medium-sized country town	021	.00608	054	.00599	200	.00602
A larger town	.117	.00662	014	.00635	320	.00679
A very large town	.110	.00716	.090	.00686	196	.00691
A city	.211	.00419	.146	.00395	.009	.00406
Elsewhere in a very large city	.165	.00469	.150	.00456	081	.00458
Close to the centre of a very large city	.212	.00481	.209	.00445	.117	.00457
Mean	.151	.00213	.107	.00203	074	.00207
Gap between lowest and highest mean score, expressed SD units	.188		.157		.201	

Table 7 Student perceptions of their school learning environments

Discussion

Most students in Australia, regardless of where they live, report very positive attitudes toward schooling. Similarly, most students report positive relationships with their teachers, regardless of the size of the community where they live. By contrast, students' perceptions of their classroom disciplinary climate vary more substantially across school communities and are, on average, less positive in rural communities than in very large cities. For example, 40 % of the students in towns with 15,000–50,000 residents reported that students in their classroom do not listen to their teacher, compared to 25 % of students who attend schools close to the centre of capital cities. When we examined how classroom disciplinary climate across all five questionnaire items varied, however, we found that differences between school communities are not educationally significant. In general, students' responses suggest that classroom disciplinary climate is not ideal in any of the school communities. This finding deserves further examination by educators and policy makers since an orderly classroom with minimal disruptions is one of the most significant predictors of student learning (Schleicher, 2009; Shin et al. 2009).

While students' perceptions do not vary substantially across the school communities, principals' perceptions do. Principals' perceptions of both teacher and student related behavior is considerably more negative in rural communities and country towns than in cities. The difference between the least and most positive school communities is large and educationally significant. The principals' responses suggest that particular attention should be paid to improving rural and regional teachers' expectations of their students, and their ability to manage student behavior and tailor instruction to meet their students' needs.

Students and principals from inner-city schools in very large cities report the most positive learning environments of any of the eight school communities. These students often come from advantaged backgrounds, as shown in Fig. 2. Most schools in inner city areas are either high fee/high status independent schools or public schools in communities with expensive housing. Decades of research has shown that students from privileged social backgrounds typically experience more positive learning environments than socially disadvantaged students (Akiba et al. 2007; Camburn and Han 2011). While our findings are therefore unsurprising, they are nevertheless a matter of concern.

Our findings also show that attention should be paid to improving learning environments not just in the most rural/remote communities, but also in largish towns of up to 50,000 residents. While the relationship between students' academic performance on PISA and community size and location is fairly consistent (as is shown by the range of academic performance in Fig. 1), students' and principals' perceptions of their school learning environments are not. On all of the indexes that we compared, students' and principals' perceptions are the most negative in larger regional towns than in smaller rural communities. This finding cannot be explained solely by student characteristics such as socioeconomic status, since student SES is, on average, higher in these larger towns than in smaller communities. Once the nature and extent to which school learning environments vary across rural and

metropolitan schools is understood, future research will be able to examine in more depth the factors that shape school learning environments in these communities.

This study includes some limitations. First, the number of schools in the smallest school community is only six. It is possible that our findings for this school community would be different if more schools were included. Second, our findings are based on students' and principals' self-reported perceptions. Further research that uses observational techniques could assess the accuracy of these perceptions.

Conclusion

This study has used a nationally representative dataset to show the degree to which students' and principals' perceptions of their school's learning environment vary across rural and urban communities in Australia. Our findings show that students' attitudes toward school and their perceptions of their relationships with their teachers are largely positive and do not vary much across the eight school communities. Students' perceptions of their classroom disciplinary climate were more negative, with some larger differences across the school communities. Taken as a whole, however, these differences are not educationally significant. By contrast, principals' perceptions of teacher and student behaviour varied substantially across the eight school communities.

Overall, students' and principals' perceptions are less positive in rural communities and towns than in cities. Somewhat surprisingly, however, the relationship between community size/location and perceptions of school learning environments is not linear, with the most negative responses occurring in larger towns rather than the smallest rural communities.

While our analysis is not intended to examine definitively how school learning environments mediate the relationship between geographic location and academic performance, our detailed findings may provide a foundation for future research. Understanding the extent to which school environments vary across rural–urban locations can be helpful for policymakers, school leaders and teachers who seek to improve the learning experiences and outcomes of their students. Our findings suggest that particular attention should be paid to improving teachers' expectations of students and teachers' ability to manage their classroom and meet the needs of their students.

Significance of study

As an experienced teacher, I have come to the realisation that personal qualities and the social fabric, such as teacher and student relationships, have a greater capacity to impact on student learning outcomes than physical resources. This is not to suggest that environment is more important than resources but highlights the value I place on human relationships in the profession of teaching. Consequently, I was excited to explore an essential component of teaching and learning, like perspectives of school learning environment. However, prior to delving into the second research paper, I was more realistic about the outcomes that I hoped to discover than I had been at the start of my research journey. Knowing that my earlier research had found principals in different communities' report 'shortages of resources' hinder learning to different degrees, I expected a similar trend to continue regarding questionnaire response to learning environments questions. Consequently, the extent to which patterns mirrored findings from paper 1 across Australia's school communities was of keen interest to me.

The findings showed that many Australian students, across all school communities, believe that schooling is worthwhile and experience positive relationships with teachers. Although I value the quality of education within Australia, this was not the outcome I was expecting. While this positive finding was surprising, it became apparent that, like paper 1, I needed to explore the PISA 2009 questionnaire data at a deeper level. This meant examining the frequencies of responses of the two most negative or positive response categories separately rather than collapsing them.

Analysis of the disciplinary climate and learning environment variables varied more considerably when I reported the frequencies of the negative responses (disagree and strongly disagree) separately. Students and principals in rural communities reported less positive responses than counterparts from large cities. Analysing the data in this way highlighted that student and principal perceptions of disciplinary climate and learning environments do vary across Australia's school communities. Comparing data across school communities identified that large regional towns (up to 50,000 residents) and rural school communities trailed behind the more positive responses of city school communities.

The findings suggest that attention should be paid to improving teachers' expectations of students and teachers' ability to manage their classroom and meet the needs of their students. This information could lead to rigorous discussion of the benefits of adapting teaching strategies according to geographic location of the school community. The findings from paper 2 also show that attention should be paid to improving learning environments not just in the most rural communities, but also in largish towns of up to 50,000 residents. This information is of value to school leaders and teachers, especially those teachers starting their career in a rural or non-metropolitan location. Ultimately the knowledge provided from research paper 2 could be helpful for policy-makers and those who seek to improve the learning experiences and outcomes of rural students.

Chapter 6:

Paper 3: A comparison of rural educational disadvantage in Australia, Canada, and New Zealand using OECD's PISA

Background

The third article is a complex paper, involving data from three countries. Consequently, the analyses reported in the third paper took longer to conduct and write up than the first two publications. Countries that could provide meaningful comparison were selected. The cross-national analyses compared schools from Australia, Canada and New Zealand. The reasoning behind the selection of Canada and New Zealand for comparison is that they are culturally, demographically, and economically similar to Australia. This allows comparison of like education systems possessing rural and remote school communities, broadly similar student outcomes in PISA 2009, and varying social-economic status suggestive of links to geographic location. For instance, Canada's results on PISA show that it has a very equitable and high performing education system (OECD, 2010a). When socioeconomic status is taken into consideration, New Zealand's students are distributed across school communities slightly less equitably than Australia and Canada. However, literacy performance in PISA 2009 would indicate otherwise as the gap between literacy levels is far more distinct across Australia's school communities.

I conducted a secondary analysis of PISA 2009 in Australia, Canada and New Zealand, comparing how school resources and learning environments varied between urban-rural school communities within each country, as well as between the three countries. In all three countries, students in rural school communities have lower reading performance and report less positive learning environments than their peers in larger towns and urban areas. Rural-urban inequalities of learning environments and academic performance are greater in Australia than in Canada or New Zealand. Moreover, across most school communities, students from Canada and New Zealand have higher reading performance than their Australian counterparts. Rural students in Australia have lower performance than their rural counterparts in Canada and New Zealand, even though on average, the socioeconomic status of rural Australian students is similar to that in Canada and greater than in New Zealand. The findings suggest that Canadian and New Zealand school communities are better able to support a greater proportion of their students, not just the most economically and socially privileged.

This research was published by *Sage Open Journal of Educational Research*, in October 2018. The publication of my third, and final paper, raised the profile of my research as Sage is a highly-regarded publisher and the journal aims to report research findings of international significance. *Sage Open* publishes peer-reviewed, original research across social and behavioural sciences and the humanities in an open access format.

Original Research

A Comparison of Rural Educational Disadvantage in Australia, Canada, and New Zealand Using OECD's PISA

SAGE Open October-December 2018: 1–12 © The Author(s) 2018 DOI: 10.1177/2158244018805791 journals.sagepub.com/home/sgo



Kevin Sullivan^{1,2}, Andrew McConney², and Laura B. Perry²

Abstract

This study compares rural educational disadvantage across Australia, Canada, and New Zealand using data from the Organisation for Economic Co-operation and Development's Programme for International Student Assessment (PISA). Across the three countries, student reading literacy and school learning environments are less positive in rural communities than in urban. Furthermore, rural disadvantage in educational outcomes (reading) and opportunities is greater in Australia than Canada or New Zealand. This could be seen as surprising as student socioeconomic status (SES), typically a strong predictor of educational outcomes, is similar for rural communities in Australia and Canada, but lower in New Zealand. Rural school principals in Australia are most likely among the three countries to report that shortages of teaching personnel hinder learning. This could suggest that policies and structures can play a role in ameliorating or exacerbating rural educational disadvantage. We conclude with questions and recommendations for future research.

Keywords

rural educational disadvantage, PISA, school learning environments, school resources

Introduction

International large-scale assessments and comparative studies are useful for examining cross-national variations in educational outcomes and experiences. More generally, they are useful for broadening perspectives and generating theory about the contexts and conditions that mediate educational phenomena. As such, they are useful for illuminating new lines of thinking that can inform both policy and practice. As described by a founding member of the International Association for the Evaluation of Educational Achievement (IEA), established in 1958 by the United Nations Educational, Scientific and Cultural Organization (UNESCO), "If custom and law define what is educationally allowable within a nation, the educational systems beyond one's national boundaries suggest what is educationally possible" (Foshay, 1962, p. 5). In this spirit, the current study examines education systems in three countries to gain a better understanding of long-standing disparities in outcomes between rural and urban school communities.

Rural educational disadvantage exists in many countries. It includes inequalities in educational outcomes, such as test scores and high school graduation rates, as well as inequalities related to educational opportunities and experiences in schools. Data from the Programme for International Student Assessment (PISA) have shown that among economically developed countries, the academic performance of 15-yearold students in cities with 100,000 or more residents is, on average, about one-half year of schooling greater than their peers in smaller towns and communities (Organisation for Economic Co-operation and Development [OECD], 2013). Analysis of individual countries has shown, however, that urban/rural achievement differences do not exist in almost half of the developed countries that participate in PISA. This suggests the possibility that national education policies and school practices may play a role in ameliorating or exacerbating rural disadvantage.

Nearly 2 decades ago in Australia, the Human Rights and Equal Opportunity Commission (2000) found that rural schooling was inferior on every indicator included in its study. Indeed, among developed countries, Australia has one of the largest urban/rural achievement gaps in PISA (OECD, 2013), and students in rural Australia have long been identified as educationally disadvantaged (De Bortoli & Thomson, 2010; Lokan, Greenwood, & Cresswell, 2001). Students in

¹Hale School, Perth, Western Australia, Australia ²Murdoch University, Western Australia, Australia

Corresponding Author:

Laura B. Perry, Murdoch University, 90 South St., Murdoch, Western Australia 6150, Australia. Email: I.perry@murdoch.edu.au

Creative Commons CC BY: This article is distributed under the terms of the Creative Commons Attribution 4.0 License (http://www.creativecommons.org/licenses/by/4.0/) which permits any use, reproduction and distribution of the work without further permission provided the original work is attributed as specified on the SAGE and Open Access pages (https://us.sagepub.com/en-us/nam/open-access-at-sage).

rural communities in Australia are less likely than their urban peers to complete secondary education (Alston & Kent, 2006) or attend university (James, 2001).

Although some studies from the United States (Reeves, 2011) and Australia (Cresswell & Underwood, 2004; Young, 1998) suggest that rural disadvantage in educational achievement is strongly associated with student socioeconomic status (SES), cross-national research from the OECD (2013) found that rural/urban achievement differences are not fully explained by student SES, and that school characteristics can account for some of the rural/urban differences in academic achievement. Mapping how schools differ across rural/urban settings can, therefore, provide insight about ways to ameliorate rural disadvantage and reduce inequalities in educational opportunities, experiences, and outcomes. Academic achievement is important, but so are experiences in school. Regardless of the degree to which school and classroom experiences are associated with student learning, all students should have the opportunity to enjoy safe classroom environments and positive relations with their teachers.

This article builds on previous research by K. Sullivan, Perry, and McConney (2013, 2014) that describes how schools differ across rural/urban communities in Australia. These studies found that students in urban Australian communities tend to have substantially higher achievement on PISA than their peers in rural settings, and that the achievement advantage is associated with inequitable distributions of school resources and teachers, and, to a lesser extent, unevenness in school learning environments.

In the current study, we compare rural disadvantage in educational outcomes and resources across three countries, namely, Australia, Canada, and New Zealand. We purposively chose these three countries because they share broadly similar systems of comprehensive secondary schooling, and similar sociocultural histories of English colonization and postcolonial development. Furthermore, the three share similar levels of economic development (all are considered highly developed), and all are among the top 15 countries on the United Nations Development Programme (UNDP) 2016 Human Development Report's (released in March 2017) list of countries with very high human development (Australia is ranked second, Canada is tied 10th, and New Zealand is 13th). All three are members of the Five Eyes (FVEY), a multilateral intelligence alliance between Australia, Canada, New Zealand, the United Kingdom, and the United States that grew out of common security interests and values.

Our aim is to meaningfully contribute to the development of knowledge and understanding of the nature and extent of rural educational disadvantage and possible solutions for remedying it. Although cross-national comparisons of rural educational disadvantage are somewhat rare, they nevertheless offer a potential to deepen our understanding of conditions and contexts that underlie it. Our research questions include:

- 1. To what extent does student academic performance differ across rural/urban locations within and between the three countries?
- 2. To what extent do school resources and learning environments differ across rural/urban locations within and between the three countries?
- 3. Are rural inequalities of educational outcomes, experiences, and opportunities greater in one country than another, and if so, what are some possible reasons why?

Context

Rural educational disadvantage exists in many economically developed countries, including Australia, Canada, and New Zealand. High school completion rates, for example, are lower in rural areas than urban areas in Australia (Alston & Kent, 2006; Marks, Fleming, Long, & McMillan, 2000) and Canada (Bowlby, 2005). Academic performance as measured by PISA is lower in rural and regional areas than in larger cities in all three countries (OECD, 2013). The reasons underlying these inequalities of outcomes are likely multiple and varied across contexts.

As previously noted, SES is one of the strongest and most stable predictors of academic performance (OECD, 2010; Reardon, 2011; Sirin, 2005). Student SES is positively associated with educational resources, experiences, and opportunities, both at home and at school, and these in turn are related to academic achievement (Bourdieu & Passeron, 1990). For example, students with higher SES backgrounds typically have more books, computers, and other educational materials in their homes than do peers with lower SES (Orr, 2003). They are also more likely to have parents who read to/ with them regularly (Coley, 2002) and to enjoy a stimulating learning environment at home (Nash & Harker, 2006; Yeung, Linver, & Brooks-Gunn, 2002). Educational advantage related to SES also manifests in schools. Students with higher SES backgrounds are more likely than their peers with lower SES to attend well-resourced schools (Chiu & Khoo, 2005; OECD, 2005; Vignoles, Levacic, Walker, Machin, & Reynolds, 2000) and more likely to have experienced and qualified teachers (Akiba, LeTendre, & Scribner, 2007; Hanushek & Rivkin, 2006) with high expectations (Rumberger & Palardy, 2005). Higher SES students also experience more time spent on instruction (Camburn & Han, 2011), and are more likely to have access to a rigorous academic curriculum (Lamb, Hogan, & Johnson, 2001; Oakes, 2000).

SES is also the strongest predictor of rural educational disadvantage in many countries (Williams, 2005). On average, rural students have lower SES than their urban peers, and this is associated with the latter typically having higher academic outcomes. It is not clear, however, whether SES is associated with poorer outcomes because of rural/urban differences in student and family characteristics, rural/urban

School		А	US	C	AN	N	ZL
community	Population	School	Student	School	Student	School	Student
Village	<3,000	21 (6%)	648 (5%)	170 (17%)	3,668 (16%)	(7%)	191 (4%)
Small town	3,000 to about 15,000	45 (13%)	1,811 (13%)	210 (22%)	5,467 (24%)	28 (17%)	721 (16%)
Town	15,000 to about 100,000	68 (19%)	2,807 (20%)	219 (23%)	5,114 (22%)	42 (26%)	1,078 (23%)
City	100,000 to about 1,000,000	108 (31%)	4,539 (32%)	254 (26%)	6,125 (27%)	42 (26%)	1,231 (27%)
Large City	>1,000,000	111 (31%)	4,445 (31%)	119 (12%)	2,755 (12%)	40 (25%)	1,422 (31%)
Total		353	14,250	972	23,129	163	4,643

Table I. Number and Proportion of Participating Schools and Students.

Note. Percentages may exceed 100 due to rounding. AUS = Australia; CAN = Canada; NZL = New Zealand.

differences in schools, or a combination of these factors. It is difficult to disentangle these variables. It seems most likely that SES status is associated with academic achievement via both family/home and school factors.

Method

PISA includes several school variables that are demonstrably related to students' educational outcomes. School resources, for example, including educational resources and instructional materials as well as qualified and experienced teachers, are associated with student learning and achievement (Chiu & Khoo, 2005; Haycock, 2001; Wenglinsky, 2002). Positive, supportive, and caring relationships between teachers and students have been shown to improve students' outcomes because they increase motivation and engagement with learning (Aultman, Williams-Johnson, & Schutz, 2009; McHugh, Horner, Colditz, & Wallace, 2013; National Research Council, 2003; Newberry, 2010; Spilt, Hughes, Wu, & Kwok, 2012). Disciplinary climate is also a strong predictor of student outcomes (Schleicher, 2009). Classrooms that have fewer distractions promote more opportunities for teaching and learning (Frempong, Ma, & Mensah, 2012; OECD, 2005; Shin, Lee, & Kim, 2009). Moreover, highquality instructional practices can promote positive student outcomes (Akiba et al., 2007; Hanushek, 2007; Hogrebe & Tate, 2010; Winheller, Hattie, & Brown, 2013).

As noted earlier, Australia, Canada, and New Zealand were purposively chosen as the comparator countries for this analysis. The three countries share many historical and sociocultural experiences and values. All three enjoy highly developed economies tied to primary resources (agriculture and natural resources). Nevertheless, we also acknowledge that the similarities among the three countries are far from perfect. In physical geography, the distances between rural, remote, and urban settings are clearly much larger in Australia and Canada than in New Zealand. In addition, and importantly, the proportion of indigenous First Nations peoples in the population are notably different-14% for New Zealand compared with 4% for Canada and 2.5% for Australia. Also, New Zealand's GDP is substantially smaller than Australia's or Canada's but New Zealand nevertheless boasts the highest public spending on education as a proportion of its GDP (7.4% compared with 5.6% for Australia and 5.4% for Canada).

To answer our research questions, we conducted secondary analysis of the OECD's PISA 2009 data for Australia, Canada, and New Zealand. Starting in 2000, the OECD has administered PISA every 3 years. Each cycle assesses 15-year-old students' literacies in reading, mathematics, and science. We intentionally chose the 2009 dataset for two reasons. First, we had used the 2009 data in our previous research, and this allowed us to build on our earlier analyses of rural/urban educational disparities for Australia (K. Sullivan et al., 2013, 2014). Second, PISA 2009 is the most recent publicly available dataset for which Reading was the discipline examined in depth (one of the three subjects is the focus of each round); our use of reading as students' foundational literacy allowed us a high degree of confidence in the cross-national comparability of PISA data. We used descriptive statistics to uncover patterns in the distribution of educational outcomes and experiences across the rural-urban continuum, within and between the three countries. This approach is common when comparing national educational systems and outcomes (see, for example, Carnoy, Khavenson, & Ivanova's, 2015, study of Russia, Estonia, and Latvia using PISA).

PISA 2009 comprised 470,000 students across 65 countries. The numbers of students and schools participating in PISA 2009 from Australia, Canada, and New Zealand is shown in Table 1. The sample for each country is nationally representative in terms of student and school characteristics (Knighton, Brochu, & Gluszynski, 2010; Telford & May, 2010; Thomson, De Bortoli, Nicholas, Hillman, & Buckley, 2010). Most countries sample between 2,000 and 5,000 students, but some including Australia and Canada sample larger numbers of students so that state or provincial comparisons can also be made. The Canadian sample is particularly large because organizers wanted to "... produce reliable estimates representative of each province and for both French and English language school systems . . ." (Knighton et al., 2010, p. 10).

Each school selected to participate in PISA is identified according to the size of the community in which it is located. This variable *School Community* comprises the following five categories:

- Village, hamlet, or rural area: A population less than 3,000 people;
- ii. Small Town: A population between 3,000 to about 15,000 people;
- iii. Town: A population between 15,000 to about 100,000 people;
- iv. City: A population between 100,000 to about 1,000,000 people;
- v. Large City: A city with more than 1,000,000 people.

Table 1 also shows the number and proportion of students and schools in each school community, for each of the three countries.

As seen in Table 1, the proportion of students in rural areas is greater in Canada than either Australia or New Zealand. In general, students are more evenly distributed across the five types of school community in Canada than in the other two countries, and the distribution of students across the five school communities is roughly comparable in Australia and New Zealand.

The PISA 2009 dataset includes student achievement data across three subject domains, plus a large array of contextual information about schools, students, and their home environments. Data were also collected from school principals (or their deputies) about a range of school characteristics, and from students about their perceptions of their school as well as background information about their parents and home environment. In this investigation, we compared educational outcomes among the five school communities within each country as well as across the three countries. Student reading performance in PISA 2009 was analyzed to identify patterns of rural disadvantage or inequity in educational outcomes. We also used school and student background questionnaire data to gain a deeper understanding of rural disadvantage. For example, we analyzed the following variables: principals' perceptions of shortages of teaching personnel and quality of educational resources, students' perceptions of classroom disciplinary climate, students' perceptions of their teachers' instructional strategies, and students' perceptions about their relationships with teachers.

We also examined how student SES varies across rural/ urban locations. We calculated the SES of each school community by averaging the SES status of all participating students in each school community type, for each of the three countries. PISA's measure of SES is known as "economic, social and cultural status" (ESCS) and is an index that comprises three variables: parental educational attainment; parental occupational status; and family material, educational, and cultural resources. The family resources variable is particularly comprehensive and includes information from students about the number of books, musical instruments, computers, and artwork in the home, the number of bedrooms and other material items, as well as the frequency of cultural activities such as attending theatrical performances or visiting museums. The proportion of students in each school community that has a parent with a bachelor's degree or higher was also incorporated into this study. We provided these data because SES, which includes parental education, is a strong predictor of educational opportunities and outcomes.

We compared rural-urban achievement gaps within and across the three countries to assess rural disadvantage reflected in students' reading literacy. Reading and other literacy performance scores are scaled to a mean of 500 and a standard deviation of 100 score points. We used the International Data Explorer (IDE), available from the U.S. National Center for Education Statistics (NCES), to calculate mean reading performance for each school community category across Australia, Canada, and New Zealand.

PISA uses a two-stage sampling process in which schools are first sampled and then students sampled within participating schools. This approach has the potential to increase the standard errors of population estimates. In this study, therefore, and consistent with PISA's recommendation, all statistics have been generated using the NCES's IDE, via a Balanced Repeated Replication (BRR) procedure to generate unbiased standard errors that take account of clustering in the samples (OECD, 2009). Furthermore, the NCES's IDE uses t tests to calculate whether differences between means are statistically significant, and we report the results of these tests in our analyses. While t tests are generally not used to compare means from more than two groups because they increase the probability of Type I errors, they are used by the NCES and the OECD for large-scale assessment data including PISA, consistent with these organizations' standards and practice (see the NCES IDE help manual at https://nces. ed.gov/surveys/international/ide/HelpFiles/ PISAIDEHelpGuide508 063014.pdf for more information).

In summary, we calculated means for six composite variables (indexes) in PISA related to school resources and learning environments, for the five school community categories in each country. Four of these index variables are derived from student responses, and the other two are from school principal responses. The four student index variables are "classroom disciplinary climate," "studentteacher relations," "teacher use of structuring and scaffolding strategies," and "teacher stimulation of reading engagement." The two indexes derived from principal questionnaire responses are "quality of educational resources" and "teacher shortage." PISA uses item response theory (IRT) to scale index variables to have a mean of "0" (zero) and a standard deviation of "1" (one; OECD, 2009). We then compared the means across school community types within and between the three countries. We calculated the size of the gap between the highest and lowest scoring community type to measure the extent of rural disadvantage (if any) within each country, in reading literacy. Based on Slavin and Fashola (1998), we considered an effect size of 0.25 standard deviations or greater as educationally significant.

Table 2. Mean Student Readi	g Performance, Socioeconomic Status,	and Parental Education, b	y School Community	/ Туре
-----------------------------	--------------------------------------	---------------------------	--------------------	--------

		SES (ESCS)		% of stud parent w degre	lents who vith a bach ee or high	have a nelor's erª		Reading lit	eracy perfo	mance
Sabaal	Australia SD = .76	Canada SD = .82	New Zealand SD = .78	Australia	Canada	New Zealand	Australia	Canada	New Zealand	Significance of mean difference
Community	M (SE)	M (SE)	M (SE)				M (SE)	M (SE)	M (SE)	
Village	09 (0.028)	.21 (0.013)	20 (0.054)	4%	۱6%	4%	475 (6.2)	501 (3.6)	495 (5.3)	Can > Aus ($p = .000$) NZ > Aus ($p = .015$)
Small Town	.08 (0.017)	.31 (0.011)	05 (0.028)	13%	23%	16%	490 (4.7)	510 (3.0)	509 (6.9)	Can > Aus (p = .000) NZ > Aus (p = .021)
Town	.16 (0.014)	.51 (0.011)	01 (0.023)	19%	22%	22%	493 (5.0)	527 (3.2)	511 (6.8)	Can > Aus ($p = .000$) Can > NZ ($p = .035$) NZ > Aus ($p = .042$)
City	.46 (0.011)	.61 (0.011)	.29 (0.022)	32%	27%	27%	528 (4.9)	527 (3.1)	549 (6.4)	NZ > Aus (p = .008) NZ > Can (p = .002)
Large City	.43 (0.012)	.69 (0.017)	.13 (0.022)	31%	12%	30%	528 (5.0)	539 (6.1)	515 (6.7)	Can > NZ (p = .009)
Country Mean	.32 (0.006)	.46 (0.005)	.10 (0.012)				514 (2.5)	527 (1.6)	519 (2.3)	
Gap	.55 (.72 SD)	.48 (.59 SD)	.49 (.63 SD)				53 points	38 points	54 points	

Note. SES = socioeconomic status; ESCS = economic, social, and cultural status.

^aPercentages may not add up to 100 due to rounding.

Findings

We examined how school communities vary by student SES and parental education. Following this, we calculated the average reading literacy performance for students in each school community type. These findings are shown in Table 2 below and also graphically in Figure 1. We combined these context and performance findings into one table to highlight the degree to which they are related. Table 2 also shows whether between-country differences in reading performance for each school community type are statistically significant as measured by independent *t* tests conducted using the IDE.

As can be seen in Table 2 and graphically in Figure 1, both context variables (mean student SES and proportion of students with a university-educated parent) are generally positively related to school community size in all three countries, notwithstanding the dip in the proportion of parents with bachelor's degrees in "Large City" school communities in Canada and the decline in SES for "Large Cities" in New Zealand. The values for these variables are typically higher in urban settings (cities) and lower in rural areas, and the differences within countries are substantial, especially in Australia and New Zealand. For example, only 4% of Australian and New Zealand students in villages have a university-educated parent, compared with 30% to 31% in large cities in these two countries. Comparison of the two context variables across the three countries shows that mean student SES at the country level is lowest in New Zealand and highest in Canada; this pattern is consistent for all five school community types across the three countries.

Table 2 also shows mean reading literacy performance for each school community. In each of the three countries, reading performance is positively related with community size/type, with villages showing the lowest reading performance, on average, and cities (New Zealand) and large cities (Australia and Canada) showing the highest. In New Zealand, mean reading literacy is the same in large cities as it is in smaller urban communities with 3,000 to 100,000 residents. The gap between the highest and lowest scoring school community is substantially larger in Australia and New Zealand (53 and 54 points, respectively) than in Canada (38 points). A score of 50 points is roughly equal to 1 year of schooling (Thomson et al., 2010). Student performance in the two smallest (most rural) communities (villages and small towns) is substantially lower in Australia than in Canada or New Zealand. In other words, rural disadvantage is sharper in Australia than in Canada or New Zealand. This can be seen more clearly in Table 3, which shows the score-point gap in reading performance between each country's mean and the means for each of its five school community types.

Cross-nationally, students in the most rural school communities performed substantially worse in Australia than in Canada or New Zealand. New Zealand's reading literacy performance is greater than Australia's in all school communities except large cities. Figure 1 shows very clearly the strong positive relationship between mean SES and reading performance across the school communities within each country. It it also the case, however, that mean reading performance is 20 points higher in rural settings (i.e., villages) in New Zealand compared with Australia despite the fact that average SES and parental educational attainment is similar (in fact, lower for New Zealand). A difference of 20 score points is substantial, representing roughly one-half year of



Figure 1. Mean SES and reading performance, disaggregated by school community type in Australia, Canada, and New Zealand. Note. SES = socioeconomic status; PISA = Programme for International Student Assessment; ESCS = economic, social, and cultural status.

 Table 3.
 Score Point Differences Between Country Mean and

 School Community Mean in Reading Literacy Using PISA (2009).

School community	Australia	Canada	New Zealand		
Village	-39	-26	-24		
Small Town	-24	-17	-10		
Town	-21	0	-8		
City	+ 4	0	+30		
Large City	+ 4	+12	-4		

Note. PISA = Programme for International Student Assessment.

school (Thomson et al., 2010). As can be seen in Figure 1, New Zealand school communities in general show substantially higher reading performance than might be predicted by their mean student SES. The context variables, however, do not explain *why* rural (and regional) disadvantage is greater in Australia than in New Zealand or Canada. While the observed difference between New Zealand and Australia in rural settings could be seen as a perplexing finding, it may also suggest that educational performance, and underlying opportunity, can be mediated by education policies and practices.

To shed further light on these cross-national differences in reading performance, disaggregated by school community types (urban vs. rural), we compared differences in school learning resources and learning environments across school community types, as perceived by students and principals. Table 4 shows school principals' perceptions of the degree to which shortages of teaching personnel and educational resources hinder learning in their school. We report the means for these two indexes for each school community type, in each country. We also calculated the difference between the highest and lowest scoring school community for each country, expressed in standard deviation units to show the magnitude of the difference (effect size).

As can been seen from Table 4, school community size is associated with teacher shortages in all three countries, with school principals in smaller (more rural) communities generally reporting more shortages than principals in larger communities (with a few exceptions, such as large cities in New Zealand and towns in Canada). Shortages of teaching personnel vary to a large degree across the three countries. Principals in Australia are the most likely to report that shortages of teaching personnel hinder learning in their school, and principals in Canada are the least likely. School principals in Australian villages report by a large margin the highest shortages (M = .67); this mean is substantially larger than in New Zealand villages (M = .32), which in turn is much larger than in Canadian villages (M = -.04). Principals in villages and small towns in Canada report shortages that are less than the OECD average, whereas shortages are on average considerably more than the OECD average in Australia and New Zealand. Across all school communities except large cities, principals report higher shortages of teaching personnel in Australia than in New Zealand, which may help

		Teacher shortage		Quality of school educational resources				
-	Australia $SD = .97$	Canada SD = .84	New Zealand $SD = .85$	Australia $SD = 1.05$	Canada $SD = .95$	New Zealand $SD = .95$		
School community	M (SE)	M (SE)	M (SE)	M (SE)	M (SE)	M (SE)		
Village	.67 (0.05)	04 (0.02)	.32 (0.08)	05 (0.06)	. (0.03)	08 (0.10)		
Small Town	.27 (0.06)	26 (0.03)	.14 (0.10)	.16 (0.05)	.50 (0.03)	.15 (0.10)		
Town	.50 (0.05)	.17 (0.03)	04 (0.09)	.27 (0.06)	.29 (0.03)	.24 (0.10)		
City	.05 (0.05)	43 (0.03)	21 (0.08)	.49 (0.05)	.52 (0.04)	.51 (0.10)		
Large City	01 (0.03)	37 (0.04)	.27 (0.11)	.43 (0.04)	.43 (0.04)	.15 (0.11)		
Mean Gap	.20 (0.02) .70 SD	–.23 (0.01) .71 SD	.08 (0.04) .62 SD	.32 (0.02) .51 SD	.35 (0.02) .43 SD	.20 (0.05) .62 SD		

Table 4. School Principals' Perceptions About Their School's Resources and Teacher Shortages.

Note. The "teacher shortage" index is reversed, that is, higher values mean more shortages. The "educational resources" index is inverted for scaling, and positive values indicate better quality educational resources.

explain why reading performance is higher in the latter. As suggested by this finding, teacher shortages are not only a problem for rural schools. In New Zealand, principals in the large city community reported the second highest mean on this index (after villages).

The quality of educational resources is generally positively related to community size in all three countries, with school principals in villages and small towns generally reporting that shortages of educational resources hinder learning to a greater extent than principals in larger (more urban) communities. Principals in cities have the most favorable responses regarding the quality of their educational resources in all three countries. Between-community inequalities on this measure are substantial and the size of the observed differences is educationally significant for all three countries. Across the three, New Zealand school principals report the highest shortages of educational materials, while the averages for Canada and Australia are similar. As shown in Table 4, differences in quality of school educational resources between communities with less than 100,000 residents in Australia and New Zealand are minimal. It is not likely, therefore, that this measure explains why New Zealand students in communities with less than 100,000 residents have higher reading achievement than their peers in Australia.

In contrast, Table 5 compares the school community means for the four student-derived indexes about teachers and classroom disciplinary climate. Classroom disciplinary climate is, on average, slightly lower in all three countries than the OECD average. The country means for Canada and Australia are very similar and slightly more positive than in New Zealand. Across all three nations, the lowest values are reported in small towns (3-15,000 residents) and towns (15-100,000), not villages (less than 3,000). In Australia, the difference in the reported disciplinary climate between regional communities (village, small town, and town) and urban communities (city and large city) is large, favoring larger (more urban) communities. In Canada and to a lesser extent New

Zealand, variations across communities are more evenly distributed. Differences in disciplinary climate across the five school communities are not educationally significant in New Zealand or Canada, as they are in Australia. Among the three countries, Australian students in the three smallest school communities report the least positive disciplinary climates, and Australian students in the two largest school communities report the most positive climates.

Table 5 also shows how students' perceptions of their relations with teachers in their schools vary by school community type and country. Within each country, this relationship is positively associated with community size, with the least positive values found in rural settings (villages or small towns), and the most positive in cities and large cities. The difference between the highest and lowest school community means, as expressed in *SD* units, is greatest in Australia and smallest in Canada, although none of these gaps is large enough to be considered educationally significant. Among the three countries, Canada has the highest (most positive) country mean and Australia the lowest. Across all five school community types, Australian students report substantially less positive relationships with their teachers than do their peers in Canada and New Zealand.

The last two indexes presented in Table 5 capture students' perceptions of their teachers' instructional strategies. For teacher use of structuring and scaffolding strategies, students in all five school communities in all three countries report higher values than the OECD average. The use of structuring and scaffolding strategies is positively associated with community size in Australia and Canada. Differences across the school communities are the largest in Australia and the smallest in Canada, although they are less than .25 *SD* in all three countries and are, therefore, not educationally significant by standard benchmarks. Across the three countries, Australian students in all five school communities are the least likely to report that their teachers use structuring and scaffolding strategies, whereas Canadian teachers are the most likely.

102

	Classroom disciplinary climate Student-teacher relations			Teacher use of structuring and scaffolding strategies		Teachers' stimulation of reading engagement						
School	Australia SD = 1.01	Canada SD = .96	New Zealand SD = .96	Australia SD = .99	Canada SD = 1.03	New Zealand SD = .95	Australia SD = 1.05	Canada SD = 1.06	New Zealand SD = 1.05	Australia SD = 1.00	Canada SD = 1.05	New Zealand SD = 1.02
Community	M (SE)	M (SE)	M (SE)	M (SE)	M (SE)	M (SE)	M (SE)	M (SE)	M (SE)	M (SE)	M (SE)	M (SE)
Village	22 (0.009)	08 (0.006)	02 (0.020)	01 (0.010)	.24 (0.007)	.07 (0.018)	.13 (0.011)	.38 (0.006)	.43 (0.023)	02 (0.010)	.09 (0.006)	.08 (0.020)
Small Town	20 (0.006)	10 (0.004)	19 (0.010)	05 (0.006)	.30 (0.004)	.10 (0.010)	.06 (0.006)	.37 (0.004)	.27 (0.011)	09 (0.006)	.10 (0.004)	.01 (0.011)
Town	26 (0.005)	11 (0.003)	19 (0.009)	.03 (0.005)	.30 (0.003)	.15 (0.008)	.12 (0.005)	.38 (0.004)	.20 (0.009)	.02 (0.005)	.15 (0.004)	.06 (0.009)
City	.01 (0.004)	09 (0.003)	05 (0.008)	.15 (0.004)	.35 (0.003)	.27 (0.008)	.25 (0.004)	.49 (0.003)	.41 (0.009)	.19 (0.004)	.35 (0.003)	.17 (0.008)
Large City	.01 (0.003)	−.01 (0.004)	09 (0.008)	.18 (0.003)	.38 (0.004)	.23 (0.008)	.30 (0.003)	.46 (0.004)	.37 (0.008)	.24 (0.003)	.33 (0.004)	.21 (0.008)
Mean	07 (0.002)	08 (0.002)	12 (0.004)	.11 (0.002)	.32 (0.002)	.19 (0.004)	.22 (0.002)	.43 (0.025)	.33 (0.005)	.14 (0.002)	.23 (0.002)	.13 (0.004)
Gap	.27 SD	.10 SD	.18 SD	.23 SD	.14 SD	.21 SD	.23 SD	.II SD	.21 SD	.33 SD	.25 SD	.20

Table 5. Student Perceptions of Classroom Disciplinary Climate, Teacher-Student Relations, and Teacher Instructional Strategies.

Note. For all indexes, smaller values are less positive.

The final instructional index we examined is students' perceptions of teachers' stimulation of students' engagement in reading. Students in all three countries report more positive responses than the OECD average in all school community types except villages and small towns in Australia. In all three countries, this index is positively related to community size, with the most favorable responses occurring in the largest (most urban) school communities, and the least favorable responses occurring in the smallest (most rural) school communities. Differences among school communities are the largest in Australia and the smallest in New Zealand and are educationally significant in Australia and Canada. Among the three countries, Canadian students report the most favorable reading engagement in all five school community types. Student responses in the three smallest, most rural school communities are the least favorable in Australia.

Discussion and Conclusion

Australian, Canadian, and New Zealander school systems share several organizational features and sociocultural traditions. Examining educational inequity associated with ruralurban differences for these three culturally alike systems provides the prospect of better understanding variables that could potentially reduce the disadvantages in educational opportunities and outcomes for rural communities. As researchers based in Australia, we are, of course, concerned with identifying trends in student and school variables to inform efforts that reduce rural educational disadvantage in Australia. The insights uncovered via large-scale international data, however, may also be useful for other countries. The findings presented here may be beneficial for informing theory about the contexts and conditions that potentially mediate rural educational disadvantage in other national contexts.

Our analysis of PISA data shows that rural educational disadvantage is more pronounced in Australia than in Canada or New Zealand. This can be seen in two ways, both within and across the three countries. First, the magnitude of the gap in educational outcomes (specifically, reading literacy performance) between rural and urban students is larger on average in Australia than in Canada or New Zealand. Second, students in rural communities in Australia enjoy less positive educational experiences than their peers in Canada or New Zealand. These findings are intriguing and somewhat surprising given that average SES (e.g., parent educational attainment) in rural school communities is similar in Australia and New Zealand.

Explanations by researchers such as Williams (2005) suggest that students living in villages and small towns, who typically have lower reading performance compared with their urban peers, often have lower SES as well. Furthermore, the perception that SES and academic achievement are closely linked resonates with the broader education community throughout Australia and reflects similar findings in international education research (Reardon, 2011). The current analysis, however, shows that rural students in New Zealand, with similar SES on average to their peers in Australia, have better reading literacy performance at 15 years old, as measured by PISA.

The reason behind New Zealand's rural success compared with similar school communities in Australia is not clear. This analysis may, however, provide some insight. On the four learning environment measures reported by students, Australian students consistently report the least favourable responses among the three countries. Principals in rural

Sullivan et al.

communities in Australia were also the most likely to report that shortages of teaching personnel hinder learning in their schools, a finding supported by Harris, Jensz, and Baldwin's (2005) research. Some researchers, such as Downes and Roberts (2018), question why teacher shortage remains an ongoing issue given our well-developed knowledge in this area. One could argue that there have been many studies about rural school staffing, but few linked to policy or tracked across time. Perhaps one answer lies in geography. For example, one could reasonably hypothesize that significantly smaller geographic distances between rural and urban communities in New Zealand, contrasted against enormous and challenging distances in Australia, reduce the level of shortages of teaching personnel for rural communities in New Zealand, and, therefore, weaken the relationship between rurality and educational disadvantage. Given greater distances between urban and rural or regional centers in Australia, it is not difficult to envisage greater staffing difficulties. The impact on attracting and retaining staff and implementing staff development would be great. For example, with regard to staff professional learning, there would likely be reduced opportunities for skill development in rural communities, and that could have direct impact on career development and skills, and, by extension, teacher recruitment and retention. Perhaps, therefore, future research could examine the apparent differences in these relationships. It would seem quite plausible that rurality (the distinction between larger and smaller school community types) impacts more negatively when geographic distances between school communities are large, with many rural settings classified as "remote," as in the case of Australia, than when distances separating communities are smaller and less challenging, such as in New Zealand.

Whether or not a hypothesis about the differential effects of distance on rural disadvantage via (for example) the differential recruitment, retention, or professional learning of teachers ultimately is supported, this study nevertheless has limitations. Perhaps most important, school-level variables used in the analysis are based on principals' and students' reported perceptions and experiences rather than independent observations. In addition, PISA samples 15-year-old students, typically in their last year of compulsory schooling, and the associations observed among rurality, SES, and educational outcomes, which typically manifest as rural educational disadvantage, may not hold at other levels of schooling (e.g., primary).

As an exploratory study that documents differences between school communities and education systems, this study lays a foundation for several future lines of enquiry. Future research could explore how schools in rural communities and small towns differ among Australia, Canada, and New Zealand, for example. Observational studies of classrooms and ethnographic studies of schools could help explain why educational outcomes are weaker in rural Australia than in comparable communities in New Zealand. Another line of research could examine how teachers are allocated to rural schools in the three countries. Future studies that examine the nature and degree of support and professional development provided to rural school teachers in the three countries could illuminate how best to support teachers in Australia, as well as contribute to a larger theoretical framework about the role of schools and school systems in mediating rural educational disadvantage.

We believe that it is vital for education systems to understand the significance of "gaps" in educational outcomes among school communities within a country. Since PISA 2009 and despite efforts to close urban-rural disparities in Australia's literacy performance, unequal outcomes persist between Australia's urban (metropolitan) and rural (provincial and remote) communities, and these differences remain large (Thomson, De Bortoli, & Underwood, 2016). For example, this study indicates that fewer Australian rural students report that their teachers foster supportive relationships with their students, promote a productive classroom climate, or use teaching strategies that are generally thought effective for teaching reading, in comparison to rural school teachers in New Zealand or Canada. Wallace (2008, p. 6) states, "The disparity between learners' interpretations of life and knowledge, and educational institutions' interpretations of learners' lives and knowledge in regional Australia, can impact on learners' engagement with education and training." It is evident that teachers need to be culturally assimilated into a community and cannot relate to, or teach, what they do not know or understand. This message is reinforced by several researchers committed to exploring rural education issues in the context of the values and motivations of the people who live in rural and remote communities, and who argue that social justice should frame our education programs (Cuervo, 2016; Pini & Bhopal, 2017; Roberts & Cuervo, 2015; White, 2015).

Considering the notably higher level of government expenditure on public education, as a percent of GDP, provided by New Zealand, it would be interesting to examine how the distribution of education funding varies between Australia, Canada, and New Zealand. Researchers such as Kline, White, and Lock (2013) have suggested that teachers in Australian rural school communities could be better supported and trained, and Schleicher (2009) has used PISA data to foreground aspects of educational policy associated with high-quality learning outcomes and equitable distribution of learning opportunities. Perhaps a finer grained examination of the distribution of educational funding for Australia could apply an urban-rural lens, as well as lessons from Canadian and New Zealander systems, to more effectively address the persistent challenge of rural disadvantage. We hope that, like our predecessors, this study contributes in a positive way to improving rural educational disparities in Australia. We hope community awareness of rural education inequity is thereby increased and policies promoted that influence educational systems to make changes that support equal opportunities and equitable outcomes for rural and remote students across Australia.

Declaration of Conflicting Interests

The author(s) declared no potential conflicts of interest with respect to the research, authorship, and/or publication of this article.

Funding

The author(s) received no financial support for the research, authorship, and/or publication of this article.

ORCID iD

Laura B. Perry D https://orcid.org/0000-0003-4252-2379

Appendix: Questionnaire items for selected indexes

Classroom disciplinary climate

Students are asked how often the following five items occur in their English/reading class:

- Students don't listen to what the teachers say
- There is noise and disorder
- The teacher has to wait a long time for the students to quiet down
- Students cannot work well
- Students don't start working until after a long time after the class begins

Students are given four response categories, with 1 = "never or hardly ever", 2 = "in some lessons", 3 = "in most lessons" and 4 = "in all lessons". The index is inverted whereby smaller values represent less positive disciplinary climates.

Student-teacher relations

Students are asked how much they disagree or agree with the following statements about teachers at their school. The index is derived from five items in the student questionnaire:

- I get along well with most of my teachers
- Most of my teachers are interested in my well-being
- Most of my teachers really listen to what I have to say
- If I need extra help, I will receive it from my teachers
- Most of my teachers treat me fairly

Students are given four response categories, with 1 = "strongly disagree", 2 =

"disagree", 3 = "agree" and 4 = "strongly agree". Higher values on the index represent more positive relations.

Teacher use of structuring and scaffolding strategies

Students are asked how often the following occur in their reading/English class:

- The teacher explains beforehand what is expected of the students
- The teacher checks that students are concentrating while working on the <reading assignment>
- The teacher discusses students' work, after they have finished the <reading assignment>
- The teacher tells students in advance how their work is going to be judged
- The teacher asks whether every student has understood how to complete the <reading assignment>
- The teacher marks students' work
- The teacher gives students the chance to ask questions about the <reading assignment>
- The teacher poses questions that motivate students to participate actively
- The teacher tells students how well they did on the <reading assignment> immediately after

Students are given four response categories, with 1 = "never or hardly ever", "in some lessons", "in most lessons" and 4 = "in all lessons".

Teacher stimulation of reading engagement

Students are asked how often the following occur in their reading/English class:

- The teacher asks students to explain the meaning of a text
- The teacher asks questions that challenge students to get a better understanding of a text
- The teacher gives students enough time to think about their answers
- The teacher recommends a book or author to read
- The teacher encourages students to express their opinion about a text
- The teacher helps students relate the stories they read to their lives
- The teacher shows students how the information in texts builds on what they already know

Students are given four response categories, with 1 = "never or hardly ever", "in some lessons", "in most lessons" and 4 = "in all lessons".

Shortages of teaching personnel

Principals are asked to evaluate the degree to which, 'student learning in their school is hindered by shortages of qualified teachers and support personnel'. The teacher shortage index is comprised of six items about support personal and teachers in different learning areas (e.g., mathematics and science). Higher values on the teacher shortage index represent greater shortages.

Quality of educational resources

Principals are asked to evaluate the degree to which, 'student learning in their school is hindered by shortages of educational materials'. The educational materials index comprises seven items about instructional materials, library books, laboratory equipment, audiovisual resources, computers, computer software and access to the Internet. The educational materials index is reversed; wherein higher values represent better resourcing.
Significance of Study

Compared to teacher quality and other aspects of school resources, less is known about how learning environments may vary between schools in rural and urban communities. A review of the literature, undertaken prior to completing the third research paper, suggests that researchers have examined the relationships between learning environments and educational outcomes and motivation for rural students (see for example, Hardre & Reeve, 2003 and Young, 1998), but have paid less attention to how learning environments vary by geographic location. An exception in Australia is a study by Waldrip and Fisher (2007), who show that students in metropolitan schools are more likely to report negative relationships with their teachers compared to students in rural and remote areas. I believe a better understanding is required of the disparities that exist between schools from different geographic locations. By examining PISA 2009 data and contrasting international school communities, this research paper highlights how school communities from countries such as Canada and New Zealand, share characteristics that can offer insights to improving equity within schools in Australia.

Paper 3 provides a comparative analysis of the Australian education system using PISA 2009 reading literacy outcomes and questionnaire responses associated with school resources and learning environments. The research of 'like' school communities within different, but culturally and economically similar international school systems provide useful comparative data. The outcome of broadening my research to be inclusive of comparable international education systems is that it highlights that trends and patterns associated with rural education in Australia are somewhat unique. The comparative analysis provided in paper 3 also provides clues as to why some countries have smaller achievement gaps between rural and metropolitan students.

My research highlights that students in rural school communities across an international study display weaker educational outcomes and generally enjoy less favourable learning

environments than their city peers. Weaker educational reading outcomes in PISA 2009 may be reflective of rural educational disadvantage. This inequality is pronounced in Australia. My investigation, presented in paper 3, indicates that education equity cannot be attributed entirely to school socioeconomic status but is likely to be influenced by variables such as, 'shortages of teaching personnel'. These variations may provide clues as to why Australia's achievement gap in reading in PISA 2009 is disproportionate to New Zealand and Canadian 'like' school communities. International school community comparison used in this way suggests that the lower than average literacy levels in PISA 2009 reading assessment, associated with rurality in Australia, may be connected in some way to the structure of school resources and learning environments.

In the ensuing chapter, I will elaborate on the key findings of my study and examine the significance, limitations, recommendations and future possibilities associated with my research.

Chapter 7:

Conclusion

Key findings

Throughout the course of my research I have explored PISA 2009 data and worked closely with my supervisors, stayed abreast of current literature, conducted research and analysed data in the endeavour to uncover evidence of associations among school resources, learning environments and student achievement, as measured in PISA 2009. I used descriptive statistics to compare school characteristics and student outcomes across different school contexts, geographies and countries. Analysing this data has enabled me to uncover possible indicators of important differences across urban-rural school communities. The research has also provided a means to develop the skills required to show meaningful patterns and relationships using large-scale, international datasets. However, as is well understood, the cross-sectional nature of data included in my study preclude establishing causal relationships among school resources, learning environments and student learning variables.

This thesis has led to findings about the ways in which perceptions of school resources and learning environments vary across school communities. Questions as to 'why?' rural school communities in Australia are disadvantaged may remain unresolved but the nature and extent of disparity and inequity in Australian rural education has been achieved. Furthermore, in analyzing learning resource and learning environment constructs greater knowledge and understanding of the characteristics of Australian students and their learning experiences has been achieved. In the following paragraphs, I summarize the main findings of this research.

Information that has been brought to light through my research will hopefully contribute to understanding variations in school resources and learning environments across Australian urban-rural school communities, and the association of this variability with variability in students' achievement. Amongst other findings, my research identified learning environments in large Australian regional towns (up to 50,000 residence) are perceived by students as experiencing less positive learning environments, and recruiting and retaining teachers in large regional Australian town school communities is considered by principals to be less positive than in rural regions. The trends from PISA 2009 questionnaire data also signpost a rural-urban geographic bias that indicates greater shortages associated with rurality, such as shortage of instructional material and equipment on learning.

In undertaking an investigation into PISA 2009 data, I gained valuable knowledge about rurality inequalities associated with education opportunities, in particular: learning materials, teachers and facilities across Australia's urban-rural school communities. I have also discovered that low SES rural school communities in New Zealand outperform their counterparts in Australia and instructional practices and relationships between teachers and students play an important role in determining the quality of student experiences and education outcomes.

Research into Australian school community learning resources suggests the presence of an unequal distribution of resources (teaching materials and personnel) between rural and urban Australian schools, favouring schools in close proximity to urban centres. My research suggests that the distribution of resources across school communities as reported by principals closely mirrors school academic performance and school socioeconomic status. For instance, rural schools, on average, have weaker academic literacy across all subjects in PISA 2009 and as indicated by school questionnaire data are more affected by shortages of teaching materials and personnel than are schools in larger towns and cities. However, my research also suggests some Australian 'large town' school communities are just as disadvantaged on some indicators as compared to the most rural/remote school communities in Australia. These findings highlight the need for researchers to gain a better understanding of the characteristics of Australia's school communities and the association of these characteristics with learning outcomes.

Whilst the association of school resources on student outcomes is still an open question, understanding how these vary across school communities provides context for future research, policy and practice. My findings support previous research that asserts school resources impact positively on student learning experiences (Alton-Lee, 2003; Darling-Hammond, 2000; Greenwald, Hedges, Laine, 1996; Hill et al., 2005). For instance, school principals in small rural communities are more likely to respond that their school's capacity to provide instruction, 'is hindered more by shortages of teaching personnel' than by, 'shortages of teaching resources'. My findings indicate that instruction is perceived by school principals to be hindered substantially more in smaller communities than in the larger, more urban communities. Also, my analysis of PISA 2009 showed that 66% percent of school principals in small rural communities responded unfavourably to questions relating to 'shortage of qualified teachers' as opposed to only 21% of school principals close to the centre of a very large city. These findings provide useful information about the distribution of resources across Australia's school communities and may raise awareness about the necessity of addressing this problem.

My research highlights that students' and principals' perceptions of their school's climate and learning environment varied across rural and urban school communities. Students' perceptions of their 'classroom disciplinary climate' were, on average, less positive in rural communities than in very large cities: for example, 40% of the students in towns with 15,000-50,000 residents reported that students in their classroom do not listen to their teacher, compared to 25% of students who attend schools close to the centre of capital cities. Principals' perceptions of both teacher and student behaviour is considerably more negative in rural communities and country towns than in cities. Students and principals from inner-city schools in very large cities reported the most positive learning environments of any of Australia's

school communities. These students, on average, come from families with higher socioeconomic backgrounds.

Information that has become known through my research includes: learning environments in large Australian regional towns (up to 50,000 residents) are perceived by students as less positive; and, recruiting and retaining teachers in large regional Australian town school communities is considered by principals to be less positive than the smallest rural regions. The trends from PISA 2009 questionnaire data also signpost a rural-urban geographic bias that indicates an increase in the shortage of instructional material and equipment associated with rurality.

Educational opportunities, experience, and outcomes for high school students are much more unevenly distributed across geographic rural-urban communities in Australia than in Canada or New Zealand. Cross-national analysis of school communities highlights that school SES is lower on average in Australian and New Zealand rural school communities than in Canada. The interrelationship between school resources and learning environments and academic achievement is difficult to distinguish. However, patterns emerged in my research that indicates that SES does not play a lone hand in determining academic performance. For instance, New Zealand students in rural school communities performed at a much higher literacy level in PISA 2009 reading than their Australian counterparts, despite lower mean school SES and similar levels of parental education. 114

Significance

A consistent finding from my research is that student and principal perceptions are less positive in rural communities and towns than in cities. These data contradict the study of Waldrip and Fisher (2007). Nevertheless, these findings show the degree to which perceptions vary across school communities, along a larger variety of dimensions than has been previously shown. Policy to support this perceived deficit in rural school communities could include the implementation of better rural teaching training for university graduates.

My findings show that the relationship between community size and perceptions of learning environments is not linear, with the most negative responses occurring in larger towns rather than the smallest rural communities. This information may be of enormous benefit to education policy-makers, school leaders and teachers who seek to improve the learning experiences and outcomes of their students. The significance of this finding is heightened when consideration is given to existing research that suggests teachers' expectations of students and teachers' ability to manage their classroom can impact on the literacy level of their students (O'Brennan, Bradshaw & Furlong, 2014; Williams, 2012).

Shortages of teaching personnel vary substantially across school communities within Australia, Canada, and New Zealand. Australian principals are the most likely to report that 'shortages of teaching personnel' hinder learning in their school. The findings contained in my thesis will hopefully lead researchers to investigate why the inter-connectedness of shortages of teaching personnel and learning outcomes is stronger in some school communities and in some national contexts than others. Reform in this area could occur through government support of university initiatives to attract students specializing in difficult to staff learning areas and closer relationships between all levels of education in Australia.

Other important findings from this thesis indicate that rural educational disadvantage is more pronounced in Australia than in New Zealand or Canada. This is illustrated through comparison of rural and urban student outcomes and principal and student questionnaire response to learning resources and learning environments, such as 'shortage of teaching personnel' and 'most of my teachers treat me fairly'. These findings, which were explored in Paper 3, pose further analysis as the average socioeconomic status and parent educational attainment in these rural school communities is similar in Australia and New Zealand. Potentially the data may provide clues to better understanding the interrelatedness of schoolrelated factors.

The higher literacy achievement of New Zealand rural school communities in PISA 2009 reading, in comparison to higher socio-economic peers in similar school communities in Australian is especially important for Australian education policy-makers, rural principals and their school community's parents, teachers and students. Innovative government policy is required that seeks to provide better rural education funding, teacher training and resource allocation to rural school communities shown to have implemented curriculum reform that targets outcome improvements, fosters close relationships between school and community, promotes positive attitudes towards learning and immerses local culture into their school community.

Through the research and analysis in this thesis, it is hoped that a clearer knowledge of academic potential and achievement will be realised in low socio-economic rural school communities across the globe. I strongly believe the message of this thesis, for Australian policy-makers and Australia's rural school communities, is to set high goals and endeavour to promote the seemingly impossible barrier to achieve equivalent academic literacy levels; not only to counterparts in rural New Zealand school communities, but more importantly to peers in Australian urban school communities.

Limitations

This study is based on a secondary analysis design and a descriptive statistics method of analysis. Descriptive analysis used in this research includes PISA 2009 literacy outcomes and principal and student questionnaire response. The data analysis utilised in my thesis provides a better understanding of individual variables, such as 'shortage of school resources', on student literacy performance in PISA 2009, and relationships, associations and distributions of data across my area of inquiry. However, performing regression analysis on specific variables with any certainty is problematic as there is an inability to disentangle variables, such as student socioeconomic status from other variables. Consequentially, I believe that it is more plausible to consider the combined effect of individual variables rather than try and attempt to identify the impact of an individual variable on a school community. Also, the end-point of this research is not just to identify specific questionnaire variables that may predict performance outcomes, but rather to provide (or, support) a better understanding of school communities themselves. This is because regardless of whether positive teacher-student relationships are related to achievement, they are still a positive 'outcome' in their own right.

A possible limitation of this study is that the school resources and learning environment variables are based on the perceptions of students and principals rather than observational data: for instance, it is possible that when measuring school resources and learning environments perceptions rather than reality differ. Also, the analysis of questionnaire mean response limits the potential of this study to measure outliers. However, given PISA's large sample sizes, I believe that the questionnaire research is of value. Another limitation of PISA is that the degree of remoteness and isolation of rural communities is not specified. It is not clear from the dataset whether schools in villages are more remote in terms of distance from students' homes or to larger population centres in Australia than in Canada or New Zealand. Evidence of how school

community size is related to remoteness, for example, may help explain the cross-national differences that were uncovered in my study.

Future research

My research has reinforced the need to learn more about how and why Australia's rural, remote and metropolitan school communities vary. It is clear there is still much to be learned in this important field of research. It will only be through learning more about school community characteristics that researchers will be able to better understand the link between rural-urban variables and be able to identify the role that school community characteristics play in determining academic opportunities and education outcomes. The findings highlight the need to consider learning resources and learning environments as an end in themselves and not just as a predictor of academic outcomes and to introduce education policies and structures to reduce educational disadvantage.

It is clear from this research that comparisons of PISA questionnaire responses across school communities at national and international levels can provide clues as to the different make-up of the school community, but it also underlines that more elements need to be analysed to discover the reasons behind educational inequity and disadvantage. Simple modifications, like expanding the number of rural school communities included within the dataset and re-classifying these school communities as to geographic location, would increase the accuracy of data, and provide a clearer understanding of rural inequities in education. For instance, rural school communities in Australia's far north could be compared with the southeastern region of Australia. Comparisons like these will increase our understanding of the dynamics of rural school communities.

This thesis identifies that a strong positive relationship between mean student SES and academic performance exists across Australia's urban-rural school communities. Further analysis is required to better inform educational policies and practices that seek to improve outcomes for rural school communities in Australia. A comparable qualitative study of rural education in Australia, Canada and New Zealand is an example of research that would strengthen my personal understanding of this quantitative study. In doing a qualitative analysis of rural school communities across Australia, Canada and New Zealand I would endeavour to gain a more thorough understanding of the interrelatedness of variables contained in my thesis. Perhaps then I could assert with a degree of confidence why Canada's rural teacher shortage is less pronounced than in Australia and New Zealand.

Recommendations

Living in a rural community, whilst offering amazing positives and benefits to individuals, is in many ways a pre-existing condition to being vulnerable to natural elements such as weather and associated natural disasters like floods, droughts, fire, famine and to down turns in rural industry associated with economic and political events that affect demands for resources. These factors have ramifications on the social fabric that influences rural school environment and the economic costs associated with school resourcing.

It is difficult to isolate and measure the impact of socioeconomic status, but it can be argued that factors such as low socioeconomic status are magnified in regional and remote communities. At the same time, the findings from the comparative study suggest that socioeconomic status cannot play a lone hand in student academic outcomes. It also suggests that the relationship between socioeconomic status and educational opportunities, experiences and outcomes vary across contexts. Understanding the conditions that mediate the relationship in different contexts would be a worthy future study.

The findings uncovered in this thesis will hopefully encourage universities to be more conscious of educating their students of specific rural education needs and take responsibility to ensure graduate teachers who are sent to remote schools are better prepared for teaching in rural settings. My optimism is that rural school communities become more aware of the need to immerse new staff in local culture and inform them and place them in a position that will empower them to better understand how their school resources and the relationships that they establish within the school community may impact on student achievement. The findings will also enable rural schools to recognise the need to better support new teachers to their school communities. This could be provided through better funded regional specific professional development designed to increase awareness of the need for teachers to establish supportive relationships with their students, promote a productive classroom environment and use effective teaching strategies. Policy-makers may also use information such as recognising the proportion of students in rural areas is greater in Canada than either Australia or New Zealand, to make informed decisions on capital expenditure.

From a personal level the most significant finding from my study is that rural students in Australia, Canada and New Zealand generally enjoy less favourable learning environments than their urban peers, with disadvantage being most pronounced in Australia. One policy recommendation that could stem from the research that I have undertaken is Australian rural schools would benefit from an increase in availability of instructional materials. Addressing teaching shortages in rural communities is difficult as it represents Australia's shrinking rural and expanding urban population. However, providing sufficient instructional materials for Australian rural school communities should be a routine matter for a wealthy country such as Australia.

The underlying message of this research is to raise awareness of the need for a more thorough analysis of the impact of learning environments and learning resources on student achievement within a school community. I would argue that schools in rural communities should have the opportunity to have resources distributed depending on specific needs. The research conducted in this thesis would also benefit from a greater understanding of school communities, for instance providing factual data, on the number of libraries and community centres within a designated radius could help to contextualise information. Exploring how schools in rural communities and small towns differ in Australia and between Australia, Canada and New Zealand would provide tremendous value to interpreting questionnaire analysis findings. Improvement in understanding could also come via observational studies of classrooms and descriptive ethnographic studies of schools. Such studies would help researchers to better understand why rural educational outcomes are lower in Australia than in New Zealand.

References

- Akiba, M., LeTendre, G. K., & Scribner, J. P. (2007). Teacher quality, opportunity gap, and national achievement in 46 countries. *Educational Researcher*, *36*(7), 369-387.
- Allen, D., & Fraser, B. J. (2007). Parent and student perceptions of classroom learning environment and its association with student outcomes. *Learning Environments Research*, *10*(1), 67-82.
- Alston, M. (2007). Rural restructuring and health service delivery in Australia: Policy failure and the role of social work? *Globalisation, Health and Social Care in the Community,* 15(3), 195–202.
- Alston, M., & Kent, J. (2006). *The impact of drought on secondary education access in Australia's rural and remote areas.* Canberra: Department of Education, Science and Training.
- Alton-Lee, A. (2003). *Quality teaching for diverse students in schooling: Best evidence synthesis June 2003.* Wellington: Ministry of Education.
- Amini, C., & Nivorozhkin, E. (2015). The urban-rural divide in educational outcomes: Evidence from Russia. International Journal of Educational Development, 44, 118-133
- Andrews, R. J. (1973). The Karmel Report and special education in Australia. *The Slow Learning Child*, 20(3), 154-159.
- Angus, L. B. (1993). Education, inequality and social identity. London: Routledge.
- Araujo, L., Saltelli, A., & Schnepf, S.V. (2017) Do PISA data justify PISA-based education policy? *International Journal of Comparative Education and Development*, 19(1), 20-34.
- Aultman, L. P., Williams-Johnson, M. R., & Schutz, P. A. (2009). Boundary dilemmas in teacher–student relationships: Struggling with "the line". *Teaching and Teacher Education*, 25(5), 636-646.
- Australian Bureau of Statistics (1998). *Living Arrangements: Rural families*. Retrieved from http://www.abs.gov.au/AUSSTATS/abs@.nsf/2f762f95845417aeca25706c00834efa /1e39520de2c18444ca2570ec001909f9!OpenDocument
- Australian Curriculum Assessment and Reporting Authority. (2013). My school. Retrieved April 22, 2013 from myschool.edu.au
- Australia's PISA results show educational disadvantage gap remains. (2016, December 7). Retrieved from http://www.teachforaustralia.org/2016/12/07/australias-pisa-resultsshow-educational-disadvantage-gap-remains/
- Auwarter, A. E., & Aruguete, M. S. (2008). Effects of student gender and socioeconomic status on teacher perceptions. *The Journal of Educational Research*, 101(4), 242-246.

- Barr, A., Gillard, J., Firth, V., Scrymgour, M., Welford, R., Lomax-Smith, J., Bartlett, D., Pike,
 B., & Constable, E. (2008). *Melbourne declaration on educational goals for young Australians*. Ministerial Council on Education, Employment, Training and Youth
 Affairs. Victoria: Ministerial Council on Education, Employment, Training and Youth
 Affairs.
- Baxter, J., Gray, M. & Hayes, A. (2011). *Families in regional, rural and remote Australia*. Retrieved from http://www.aifs.gov.au/institute/pubs/factssheets/2011/fs201103.html
- Beutel, D., Adie, L., & Hudson, S. (2011). Promoting rural and remote teacher education in Australia through the over the Hill project. *International Journal of Learning*, *18*(2), 377-388.
- Bourdieu, P., & Passeron, J. C. (1990). *Reproduction in education, society and culture* (Vol. 4). Thousand Oaks, CA: Sage.
- Bowlby, G. (2005). Provincial dropout rates trends and consequences. *Education matters: Insights on education, learning and training in Canada.* Ottawa: Statistics Canada.
- Bowles, S. & Levin, H. (1968). The determinants of scholastic achievement-An appraisal of some recent evidence. *The Journal of Human Resources*, 1(3), 3-24.
- Buchanan, R. (2011). Paradox, promise and public pedagogy: Implications of the federal government's digital education revolution. *Australian Journal of Teacher Education*, 36(2), 67-78. Retrieved from https://ajet.org.au/index.php/AJET/article/viewFile/64/25
- Carnoy, M., Khavenson, T., & Ivanova, A. (2015). Using timss and PISA results to inform educational policy: A study of russia and its neighbours. *Compare: A Journal of Comparative and International Education*, 45(2), 248-271. doi: 10.1080/03057925.2013.855002
- Caro, D. H., McDonald, J. T., & Willms, J. D. (2009). Socio-economic status and academic achievement trajectories from childhood to adolescence. *Canadian Journal of Education/Revue canadienne de l'éducation*, 32(3), 558-590.
- Camburn, E. M., & Han, S. W. (2011). Two decades of generalizable evidence on U.S. Instruction from national surveys. *Teachers College Record*, *113*(3), 561-610.
- Centra, J., & Potter, D. (1980). School and teacher effects: An interrelational model. *Review of Educational Research*, 50(2), 273-291.
- Chiu, M. M., & Khoo, L. (2005). Effects of resources, inequality, and privilege bias on achievement: Country, school, and student level analyses. *American Educational Research Journal*, 42(4), 575-604.
- Chudgar, A., & Luschei, T. F. (2009). National income, income inequality, and the importance of schools: A hierarchical cross-national comparison. *American Educational Research Journal*, *46*(3), 626-658.
- CIA. (2014). CIA world factbook. Retrieved from https://www.cia.gov/library/publications/the-world-factbook/geos/as.html
- Clotfelter, C., Ladd, H. F., Vigdor, J., & Wheeler, J. (2007). High-poverty schools and the distribution of teachers and principals. *North Carolina Law Review*, 85, 1345-1380.
- Coleman, J. S., Campbell, E., Hobson, C., McPartland, J., Mood, A., Weinfeld, F., & York, R. (1966). *The Coleman Report*. Equality of Educational Opportunity. Retrieved from https://eric.ed.gov/

- Coley, R. J. (2002). An uneven start: Indicators of inequality in school readiness. Princeton, NJ: Educational Testing Service.
- Connell, R. W., Ashenden, D. J., Kessler, S., & Dowsett, G. W. (1982). *Making the difference: Schools, families and social division*. Sydney: Allen & Unwin.
- Connell, R. W., Campbell, C., Vickers, M., Welch, A., Foley, D. & Bagnall, N. (2007). *Education, change and society*. Victoria: Oxford University Press.
- Considine, G. & Zappalà, G. (2002) Factors influencing the educational performance of students from disadvantaged backgrounds. In T. Eardley and B. Bradbury (Ed.), Competing visions: Refereed proceedings of the National Social Policy Conference 2001, SPRC Report 1/02 (pp. 91-107). University of New South Wales, Sydney: Social Policy Research Centre
- Cresswell, J., & Underwood, C. (2004). *Location, location, location: Implications of geographic situation on Australian student performance in PISA 2000.* Camberwell, VIC: Australian Council for Educational Research.
- Cresswell, J., Schwantner, U., & Waters. C. (2015). A Review of International Large-Scale Assessments in Education: Assessing Component Skills and Collecting Contextual Data, PISA, The World Bank, Washington, D.C./OECD Publishing, Paris, https://doi-org.libproxy.murdoch.edu.au/10.1787/9789264248373-en.
- Crofford, G.D., Pedersen, J., & Garn, G. (2011). Exploring the relationship between National Board Certification and high school student achievement. *New Horizons for Learning*. Retrieved from http://education.jhu.edu/newhorizons/Journals/Winter2011/Crofford
- Cuervo, H. (2016). *Understanding social justice in rural education*. New York: Palgrave Macmillan.
- Darling-Hammond, L. (2000). *Teacher quality and student achievement: A review of state policy evidence*. Seatlle: University of Washington, Center for Teaching and Policy.
- Darling-Hammond, L., & Ball, D. L. (1997). Teaching for high standards: What policymakers need to know and be able to do. In J. Hunt (Ed.), *Implementing academic standards* (pp. 1-61). North Carolina: National Educational Goals Panel.
- Darling-Hammond, L., Holtzman, D. J., Gatlin, S. J., & Vasquez Heilig, J. (2005). Does teacher preparation matter? Evidence about teacher certification, Teach for America, and teacher effectiveness. *Education Policy Analysis Archives*, *13*(42), 266-297.
- De Bortoli, L., & Thomson, S. (2010). *Contextual factors that influence the achievement of australia's Indigenous students: Results from PISA 2000–2006*. Camberwell, VIC: Australian Council for Educational Research.
- Diseth, Å. (2007). Students' evaluation of teaching approaches to learning, and academic achievement. *Scandinavian Journal of Educational Research*, *51*, 185–204.
- Downes, N & Roberts, P. (2018). Revisiting the schoolhouse: A literature review on staffing rural, remote and isolated schools in Australia 2004-2016. Australian and International Journal of Rural Education, 28 (1). pp.31-54.
- Fehring, H. (2010). Multiple literacies in the ICT Age: Implications for teachers and teacher educators, an Australian perspective. In S. Rodrigues (Ed.), *Multiple literacy and* science education: ICTs in formal and informal learning environments (pp. 180-206). Pennsylvania: IGI Global.

- Foshay, A. W. (1962). Educational achievements of thirteen-year olds in twelve countries: Results of aniInternational research project, 1959-61 (Vol. 4). Hamburg: Unesco.
- Fraser, B. J., & Fisher, D. L. (1982). Predicting students' outcomes from their perceptions of classroom psychosocial environment. *American Educational Research Journal*, 19(4), 498-518.
- Frempong, G., Ma, Z., & Mensah, J. (2012). Access to postsecondary education: Can schools compensate for socioeconomic disadvantage? *Higher Education*, 63, 19-32. doi: DOI 10.1007/s10734-011-9422-2
- Godden, N. (2007). *Regional young people and youth allowance: Access to tertiary education*. Wagga Wagga: Centre for Rural Social Research.
- Gonski, D., Boston, K., Greiner, K. L., C., Scales, B., & Tannock, P. (2011). *Review of funding for schooling—final report*. Canberra: Australian Government, Department of Education, Employment and Workplace Relations.
- Goodwin, A. L., & Kosnik, C. (2013). Quality teacher educators = quality teachers? Conceptualizing essential domains of knowledge for those who teach teachers. *Teacher Development*, *17*(3), 334-346.
- Gordon, I., & Monastiriotis, V. (2007). Education, location, education: A spatial analysis of English secondary school public examination results. *Research Papers in Environmental and Spatial Analysis, 116*, 1-32.
- Greenwald, R., Hedges, L., & Laine, R. (1996). The effect of school resources on student achievement. *Review of Educational Research*, 66(3), 361-396.
- Haberman, J. (2006). The impact of differential expenditures on school performance. *Educational Researcher*, 18, 45-62.
- Halsey, J. (2018). *Independent review into regional rural and remote education—Final report*. Canberra: Department of Education and Training, Commonwealth of Australia.
- Hanushek, E. A. (2007). Some U.S. evidence on how the distribution of educational outcomes can be changed. In L. Wößmann & P. E. Peterson (Eds.), *Schools and the equal opportunity problem* (pp. 159-190). Cambridge, Massachusetts: MIT Press.
- Hanushek, E. A., & Rivkin, S. (2006). School quality and the black white achievement gap. Working Paper no. 12651. Cambridge, MA: National Bureau of Economic Research.
- Hardré, P., & Reeve, J. (2003). A motivational model of rural students' intentions to persist in, versus drop out of, high school. *Journal of Educational Psychology*, 95(2), 347-356.
- Hardré, P., Sullivan, D., & Crowson, H. (2009). Student characteristics and motivation in rural high schools. *Journal of Research in Rural Education*, 24(16), 1-19.
- Harris, K., Jensz, F., & Baldwin, G. (2005). Who's teaching science? Meeting the demand for qualified for science teachers in Australian secondary schools. Melbourne: Centre for the Study of Higher Education, The University of Melbourne.
- Haslett, B. (1976). Attitudes toward teachers as a function of student academic self-concept. *Research in Higher Education.* 4(1), 41-58.
- Hattie, J. (2009). Visible learning: a synthesis of over 800 meta-analyses relating to achievement. New York: Routledge.
- Haycock, K. (2001). Helping all students achieve: Closing the achievement gap. *Educational Leadership*, 58(6), 6-11.

- Hill, H., Rowan, B., & Loewenberg Ball, D. (2005). Effects of teachers' mathematical knowledge for teaching on student achievement. *American Educational Research Journal.* 42(2), 371-406.
- Hogrebe, M. C., & Tate IV, W. F. (2010). School composition and context factors that moderate and predict 10th-grade science proficiency. *Teachers College Record*, 112(4), 1096-1136.
- Hopmann, S. T., Brinek, G., & Retzl, M. (Eds.), (2008). *PISA according to PISA: Does PISA keep what it promises?* Vienna: LIT-Varlag.
- Hoy, W. K. (1990). Organizational climate and culture: A conceptual analysis of the school workplace. *Journal of Educational and Psychological Consultation*, *1*(2), 149-168.
- Hoy, W. K., Tarter, C. J., & Hoy, A. W. (2006). Academic optimism of schools: A force for student achievement. *American Educational Research Journal*, 43(3), 425-447.
- Human Rights and Equal Opportunity Commission. (2000). *National inquiry into rural and remote education*. Sydney: HREOC.
- Ilomäki, L., & Kankaanranta, M. (2009). The information and communication technology (ICT) competence of the young. In L. Tan Wee Hin & R. Subramaniam (Eds.), *Handbook of research on new media literacy at the K-12 level: Issues and challenges* (pp. 101-118). Hershey, PA: Information Science Reference Publishing.
- Jacobs, L. A. (2016). Dealing fairly with winners and losers in school: Reframing how to think about equality of educational opportunity 50 years after the Coleman Report. *Theory and Research in Education, 14*(3), 313-332.
- James, R. (2001). Participation disadvantage in Australian higher education: An analysis of some effects of geographical location and socioeconomic status. *Higher Education*, 42(4), 455-472.
- Kaplan, L. S., & Owings, W. A. (2013). Culture re-boot: Reinvigorating school culture to improve student outcomes. Los Angeles, CA: Corwin Press.
- Kline, J., White, S., & Lock, G. (2013). The rural practicum: Preparing a quality teacher workforce for rural and regional Australia. Journal of Research in Rural Education, 28(3), p1-13.
- Konstantopoulos, S., & Borman, G. (2011). Family background and school effects on student achievement: A multilevel analysis of the Coleman data. *Teachers College Record*, *113*(1), 97-132.
- Knighton, T., Brochu, P., & Gluszynski, T. (2010). *Measuring up: Canadian results of the OECD PISA study: the performance of Canada's youth in reading, mathematics and science 2009: first results for Canadians aged 15.* Ottawa: Statistics Canada.
- Lamb, S. (1994). Dropping out of school in Australia: Recent trends in participation and outcomes. *Youth Society*, 26(2), 194-222.

- Lamb, S., Glover, S., & Walstab, A. (2014). Educational disadvantage and regional and rural schools. Retrieved from https://research.acer.edu.au/research_conference/RC2014/4august/14/
- Lamb, S., Hogan, D., & Johnson, T. (2001). The stratification of learning opportunities and achievement in Tasmanian secondary schools. *The Australian Journal of Education*, 45(2), 153-167.
- Lamb, S., Walstab, A., Teese, R., Vickers, M., & Rumberger, R. (2004). Staying on at school: Improving 599 student retention in Australia. Brisbane: Department of Education and the Arts, The State of 600 Queensland.
- Leithwood, K., Louis, K. S., Wahlstrom, K., Anderson, S., Mascall, B., & Gordon, M. (2010). How successful leadership influences student learning: The second installment of a longer story. In A. Hargreaves, A. Lieberman, M. Fullan & D. Hopkins (Eds), Second international handbook of educational change (pp. 611-629). Netherlands: Springer.
- Lizzio, A., Wilson, K., & Simons, R. (2002). University students' perceptions of the learning environment and academic outcomes: implications for theory and practice. *Studies in Higher Education*, 27, 27-52.
- Lock, G., Reid, J.A., Green, B., Hastings, W., Cooper, M., & White, S. (2009). Researching rural-regional (teacher) education in Australia. *Education in Rural Australia*, 19(2), 31.
- Lokan, J., Greenwood, L., & Cresswell, J. (2001). *The PISA 2000 survey of students' reading, mathematical and scientific literacy skills: How literate are Australia's students?* Camberwell VIC: Australian Council for Educational Research.
- Lonsdale, M. (2003). *Impact of school libraries on student achievement: A review of the research*. Camberwell VIC: Australian Council for Educational Research.
- Lounkaew, K. (2013). Explaining urban–rural differences in educational achievement in Thailand: Evidence from PISA literacy data. *Economics of Education Review*, *37*, 213-225.
- Ma, X., & Willms, J. D. (2004). School disciplinary climate: Characteristics and effects on eighth grade achievement. *Alberta Journal of Educational Research*, *50*(2), 169-188.
- Marton, F. (1981). Phenomenography—describing conceptions of the world around us. *Instructional science*, *10*(2), 177-200.
- Mandela, N. (2003, July 16). Launch of mindset network. Planetarium, university of the Witwatersrand Johannesburg South Africa. [Transcript]. Retrieved from http://db.nelsonmandela.org/speeches/pub_view.asp?pg=item&ItemID=NMS909&txtstr =education%20is%20the%20most%20powerful
- Marks, G. N., Fleming, N., Long, M., & McMillan, J. (2000). Patterns of participation in year 12 and higher education in Australia: Trends and issues [LSAY research report no. 17]. Melbourne: Australian Council for Educational Research.
- Marks, G., & Fleming, N. (1999). Early school leaving in Australia: Findings from the 1995 Year 9 LSAY cohort. *Longitudinal surveys of Australian youth research report*. Retrieved from http://research.acer.edu.au/lsay_research/71
- McHugh, R. M., Horner, C. G., Colditz, J. B., & Wallace, T. L. (2013). Bridges and barriers: Adolescent perceptions of student–teacher relationships. *Urban Education*, 48(1), 9-43.

- McMorrow, J. (2011). Real reform in school funding. Retrieved from http://publiceducationalliance.org.au.
- Minguez, A., & Ballesteros, E. (2008). The digital divide in education in the knowledge society. In G. Putnik & M. Gunha (Eds.), *Encyclopedia of networked and virtual* organizations (pp.430-435). Pennsylvania: Idea Group Publishing.
- Ministerial Council on Education, Employment, Training and Youth Affairs. (2003). *Demand* and supply of primary and secondary school teachers in Australia. Retrieved from http://www.mceetya.edu.au/mceetya/default.asp?id=11940
- Mullis, I. V., Martin, M. O., Foy, P., & Arora, A. (2012). *TIMSS 2011 international results in mathematics*. International Association for the Evaluation of Educational Achievement. Massachusetts: TIMMS & PIRLS International Study Center.
- Nash, R., & Harker, R. K. (2006). Signals of succes: Decoding the sociological meaning of associations between childhood abilities and adult educational achievement. In H. Lauder, P. Brown, J.-A. Dillabough & A. H. Halsey (Eds.), *Education, globalization* and social change (pp. 420-434). Oxford: Oxford University Press.
- National Research Council. (2003). Engaging schools: Fostering high school students' motivation to learn. Washington, DC: National Academies of Science.
- Newberry, M. (2010). Identified phases in the building and maintaining of positive teacher– student relationships. *Teaching and Teacher Education*, 26(8), 1695-1703.
- Nias, J. (1989). Refining the 'cultural perspective'. *Cambridge Journal of Education*, 19(2), 143-146.
- Noel, S., & de Broucker, P. (2001). Intergenerational inequities: A comparative analysis of the influence of parents' educational background on length of schooling and literacy skills. In W. Hutmacher, D. Cochrane & N. Bottani (Eds.), *In pursuit of equity in education: Using international indicators to compare equity policies* (pp. 277-298). Dordrecht: Kluwer Academic.
- Nous Group. (2011). Schooling challenges and opportunities. A report for the review of funding for schooling panel. Retrieved from http://www.deewr.gov.au/Schooling/ReviewofFunding/Documents/Nous-SchoolingChallengesandOpportunities.pdf.
- Nussbaum, J. (1992). Effective teacher behaviors. *Communication Education*, 41(2), 167-180.
- Nye, B., Konstantopoulos, S., & Hedges, L. V. (2004). How large are teacher effects? *Educational Evaluation and Policy Analysis*, 26(3), 237-257.
- Oakes, J. (2000). The distribution of knowledge. In R. Arum & I. R. Beattie (Eds.), *The structure of schooling: Readings in the sociology of education* (pp. 224-234). Mountain View, CA: Mayfield Publishing.
- O'Brennan, L., Bradshaw, C., & Furlong, M. (2014). Influence of classroom and school climate on teacher perceptions of student problem behavior. *School Mental Health*, *6*(2), 125-136.
- Organisation for Economic Co-operation and Development. (2004). Learning for tomorrow's world: First results from PISA 2003. Paris: OECD.

- Organisation for Economic Co-operation and Development. (2005). School factors related to quality and equity: Results from PISA 2000. Paris, France: Author.
- Organisation for Economic Co-operation and Development. (2007). *OECD glossary of statistical terms*. Retrieved from https://stats.oecd.org/glossary/download.asp
- Organisation for Economic Co-operation and Development. (2009). *PISA data analysis manual*. Paris, France: Author.
- Organisation for Economic Co-operation and Development. (2010). PISA 2009 Assessment Framework: Key Competencies in reading, methematics and science. Paris: OECD.
- Organisation for Economic Co-operation and Development. (2010). *PISA 2009 results: Overcoming social background—Equity in learning opportunities and outcomes* (Vol. 2). Paris, France: Author.
- Organisation for Economic Co-operation and Development. (2010b). PISA 2009 results: What makes a school successful? Resources, policies and practices. (Vol. 4). Paris: OECD.
- Organisation for Economic Co-operation and Development. (2011). *Education at a glance* 2011: OECD indicators. Paris: OECD.
- Organisation for Economic Co-operation and Development. (2012). *PISA 2009 technical report*. Paris: OECD.
- Organisation for Economic Co-operation and Development. (2013). *What makes urban schools different?* (PISA in Focus No. 28). Paris, France: Author.
- Ofsted. (2000). Improving city schools. London: Ofsted.
- Opoku-Asare, N.A.A., & Siaw, A.O. (2015). Rural-Urban disparity in students' academic performance in visual arts education: Evidence from six senior high schools in Kumasi Ghana. *Sage Open*, 1-14. doi: 10.1177/2158244015612523
- Orr, A. J. (2003). Black-white differences in achievement: The importance of wealth. *Sociology of Education*, *76*, 281-304.
- Patrick, H., Kaplan, A., & Ryan, A. (2011). Positive classroom motivational environments: Convergence between mastery goal structure and classroom social climate. *Journal of Education Psychology*, 103(2), 367-382.
- Pini, B., & Bhopal. K. (2017). Racialising rural education. *Race Ethnicity and Education*, 20(2), 192-196. Doi:10.1080/13613324.2015.1115620
- Portes, P. R. (2005). *Dismantling educational inequality: A cultural-historical approach to closing the achievement gap.* New York: Peter Lang Publishing.
- Ramos, R., Duque, J. C., & Nieto, S. (2012). *Decomposing the rural-urban differential in student achievement in Colombia using PISA microdata*. Discussion paper. Bonn: Institute for the Study of Labor.
- Randhawa, B. S., & Michayluk, J. O. (1975). Learning environment in rural and urban classrooms. *American Educational Research Journal*, *12*(3), 265-279.
- Reardon, S. F. (2011). The widening achievement gap between the rich and the poor: New evidence and possible explanations. In G. Duncan & R. Murnane (Eds.), *Whither opportunity? Rising inequality, schools, and children's life chances*. New York: Russell Sage Foundation.

- Reeves, E. B. (2011). The effects of opportunity to learn, family socioeconomic status, and friends on the rural math achievement gap in high school. *American Behavioral Scientist*, *56*(7), 887–907.
- Roberts, P., & Cuervo, H. (2015). What next for rural education research? *Australian and International Journal of Rural Education*, 25(3), 1-8.
- Rosenfeld, L., Richman, J., & Bowe, G. (2000). Social support networks and school outcomes: The centrality of the teacher. *Child and Adolescent Social Work Journal*, 17(3), 205–226.
- Rothman, S., & McMillan, J. (2003). Influences on achievement in literacy and numeracy. *Longitudinal Survey of Australian Youth Research Report*, 36. Melbourne: Council for Educational Research.
- Rudd, K., Smith, S., & Conroy, S. (2007). *A digital education revolution: Election 2007 policy document*. Canberra: Australian Labour Party.
- Rumberger, R. W., & Palardy, G. J. (2005). Does segregation still matter? The impact of student composition on academic achievement in high school. *Teachers College Record*, 107(9), 1999-2045.
- Rutkowski, L., Gonzalez, E., Joncas, M., & Davier, M. v. (2010). International large-scale assessment data: Issues in secondary analysis and reporting. *Educational Researcher*, *39*(2), 142-151. doi:10.3102/0013189X10363170
- Rutkowski, L., & Rutkowski, D. (2010). Getting it 'better': The importance of improving background questionnaires in international large-scale assessment. *Journal of Curriculum Studies*, *42*(3), 411-430. doi:10.1080/00220272.2010.487546
- Schleicher, A. (2009). Securing quality and equity in education: Lessons from PISA. *Prospects*, *39*, 251-263.
- Schmidt, W. H., & Burroughs, N. A. (2013). Opening the Black Box: Prospects for using international large-scale assessments to explore classroom effects. *Research in Comparative and International Education*, 8(3), 236-247.
- Sellar, S., Rutkowski, D., & Thompson, G. (2017). The Global Education Race: Taking the Measure of PISA and International Testing. Edmonton, Alberta: Brush Education. Retrieved from https://search-ebscohostcom.libproxy.murdoch.edu.au/login.aspx?direct=true&db=nlebk&AN=1500013&sit e=ehost-live
- Shields, L. (2015). From Rawlsian autonomy to sufficient opportunity in education. *Politics, Philosophy & Economics, 14*(1) 53–66.
- Shields, L., Newman, A., & Satz, D. (2017) Historical Overview. In Stanford Encyclopedia of Philosophy. Retrieved from https://plato.stanford.edu/archives/sum2017/entries/equaled-opportunity/
- Shin, J., Lee, H., & Kim, Y. (2009). Student and school factors affecting mathematics achievement. *School Psychology International*, 30(5), 520-537. doi: 10.1177/0143034309107070
- Sirin, S. R. (2005). Socioeconomic status and academic achievement: A meta-analytic review of research. *Review of Educational Research*, 75(3), 417–453.
- Slavin, R. E., & Fashola, O. S. (1998). Show me the evidence: Proven and promising programs for America's schools. Thousand Oaks, CA: Corwin Press.

- Smyth, J., McInerney, P., & Hattam, R. (2003). Tackling school leaving at its source: A case of reform in the middle years of schooling. *British Journal of Sociology of Education*, 24(2), 177-193.
- Spilt, J. L., Hughes, J. N., Wu, J. Y., & Kwok, O. M. (2012). Dynamics of teacher–student relationships: Stability and change across elementary school and the influence on children's academic success. *Child Development*, *83*(4), 1180-1195.
- Stewart, E. (2008). School structural characteristics, student effort, peer associations, and parental involvement: The influence of school and individual level factors on academic achievement. *Education and Urban Society*, *40*(2), 179-204.
- Stewart, W. (2013). *How PISA came to rule the world: Feature*. London: TES Global Limited.
- Sullivan, K., Perry, L. B., & McConney, A. (2014). How do school learning environments differ across Australia's rural, regional and metropolitan communities? *The Australian Educational Researcher*, 41(5), 521-540. DOI: 10.1007/s13384-014-0144-1.
- Sullivan, K., Perry, L., & McConney, A. (2013). How do school resources and academic performance differ across Australia's rural, regional and metropolitan communities? *Australian Educational Researcher*, 40(3), 353-372.
- Tate, W. F. (1997). Race-ethnicity, SES, gender, and language proficiency trends in mathematics achievement: An update. *Journal for Research in Mathematics Education*, 28(6), 652-679.
- Tayyaba, S. (2012). Rural-urban gaps in academic achievement, schooling conditions, student, and teachers' characteristics in Pakistan. *International Journal of Educational Management*, 26(1), 6-26.
- Teddlie, C., & Reynolds, D. (2000). *The international handbook of school effectiveness research*. London: Psychology Press.
- Teese, R., & Polesel, J. (2003). Undemocratic schooling: Equity and quality in mass secondary education in Australia. Melbourne: Melbourne University Publishing.
- Telford, M., & May, S. (2010). *PISA 2009: Our 21st century learners at age 15*. Wellington, NZ: Ministry of Education.
- Thomson, P. (2002). Schooling the rustbelt kids. Stoke on Trent, UK: Trentham Books.
- Thomson, S., & De Bortoli, L. (2008). *Exploring scientific literacy: How Australia measures up*. Camberwell, Victoria: Australian Council for Educational Research.
- Thomson, S., & De Bortoli, L. (2010). *Contextual factors that influence the achievement of Australia's Indigenous students: Results from PISA 2000–2006*. Camberwell, VIC: Australian Council for Educational Research Press.
- Thomson, S., Cresswell, J., & De Bortoli, L. (2004). *A focus on mathematical literacy among Australian 15-year-old students in PISA 2003*. Melbourne: Australian Council for Educational Research.
- Thomson, S., De Bortoli, L., Nicholas, M., Hillman, K., & Buckley, S. (2010). *Challenges* for Australian education: Results from PISA 2009. Camberwell, VIC: Australian Council for Educational Research Press.

- Thomson, S., De Bortoli, L., & Underwood, C. (2016). *PISA 2015: A first look at Australia's results*. Camberwell, VIC: Australian Council for Educational Research Press.
- UNICEF. (2013). Child well-being in rich countries. Retrieved from http://www.unicefirc.org/Report-Card-11/
- Vignoles, A., Levacic, R., Walker, J., Machin, S., & Reynolds, D. (2000). *The relationship* between resource allocation and pupil attainment: A review. London: DfEE.
- Vinson, A., Esson, K., & Johnston, K. (2002). Inquiry into the provision of public education in NSW: Report of the 'Vinson inquiry'. Sydney: New South Wales Teachers' Federation and the Parents and Citizens Council.
- Waldrip, B. G., & Fisher, D. L. (2007). Student perceptions of teacher-student interpersonal behaviour and cultural factors of learning environment in metropolitan and country schools. *Education in Rural Australia*, 17(2), 37-54.
- Wallace, R. (2008). Reluctant learners: Their identities and educational experiences (NCVER Occasional Paper). Adelaide, Australia: National Centre for Vocational Educational Research.
- Webster, B. J., & Fisher, D. L. (2000). Accounting for variation in science and mathematics achievement: A multilevel analysis of Australian data Third International Mathematics and Science Study (TIMSS). School Effectiveness and School Improvement, 11(3), 339-360.
- Welch, A., Helme, M. & Lamb, P. (2007). Rurality and inequality in education: The Australian experience. In R. Teese, S. Lamb & M. Duru-Bellat (Eds.), *International studies in* educational inequality, theory and policy, Vol. 2: Inequality in education systems (pp. 271-293). Dordrecht: Springer.
- Wenglinsky, H. (2002). The link between teacher classroom practices and student academic performance. *Education policy analysis archives*, 10(12). Retrieved from http://eric.ed.gov/
- White, K. R. (1982). The relation between socioeconomic status and academic achievement. *Psychological Bulletin*, *91*(3), 461. Doi:10.1037/0033-2909.91.3.461
- White, S. (2008). Placing teachers? Sustaining rural schooling through place-consciousness in teacher education. *Journal of Research in Rural Education*, 23(7). Retrieved from http:// jrre.psu.edu/articles/23-7.pdf
- White, S. (2015). Extending the knowledge base for (rural) teacher educators. *Australian and International Journal of Rural Education*, 25(3), 50-61.
- Williams, J. H. (2005). Cross-national variations in rural mathematics achievement. Journal of Research in Rural Education, 20(5). Retrieved from http://jrre.psu.edu/articles/20-5.pdf.
- Williams, A. (2012). The effects of teachers' expectations and perceptions on student achievement in reading for third and fifth grade students. (Doctoral dissertation). Retrieved from http://aquila.usm.edu/cgi/viewcontent.cgi?article=1791&context=dissertations
- Willms, J.D. (2010). School composition and contextual effects on student outcomes. *Teachers College Record*, *112*(4), 1008–1037.
- Willms, J. D., & Ma, X. (2004). School disciplinary climate: Characteristics and effects on eighth grade achievement. *Alberta Journal of Educational Research*, *50*(2), 169.

- Winheller, S., Hattie, J., & Brown, G. L. (2013). Factors influencing early adolescents' mathematics achievement: High-quality teaching rather than relationships. *Learning Environments Research*, 16(1), 49-69. doi: 10.1007/s10984-012-9106-6
- Woessmann, L. (2016). The importance of school systems: Evidence from international differences in student achievement. *Journal of Economic Perspectives*, *30*(3), 3-32.
- Yarrow, A., Ballantyne, R., Hansford, B., Herschell, P., & Millwater, J. (1999). Teaching in rural and remote schools: A literature review. *Teaching and Teacher Education*, 15, 1-13.
- Yeung, W. J., Linver, M. R., & Brooks-Gunn, J. (2002). How money matters for young children's development: Investment and family process. *Child Development*, *73*, 1861-1879.
- Young, D. J. (1998). Rural and urban differences in student achievement in science and mathematics: A multilevel analysis. *School Effectiveness and School Improvement*, *9*(4), 386-418.
- Zhang, Y. (2006). Urban-rural literacy gaps in sub-saharan Africa: The roles of socioeconomic status and school quality. *Comparative Education Review*, 50(4), 581-602.